# CHARTER REVISION COMMISSION Special Meeting June 2, 2022, 7:00 PM HYBRID MEETING BOARD OF EDUCATION CONFERENCE ROOM 501 Kings Highway, East

A recording of this meeting can be found here: <a href="https://www.youtube.com/watch?v=cSBICVBaelg">https://www.youtube.com/watch?v=cSBICVBaelg</a>

#### **DRAFT MINUTES**

MEMBERS PRESENT: Bryan Cafferelli-Chair, Christopher Brogan-Vice Chair, Marlene Battista-Secretary, Pamela

Iacono, John Mitola, John Wynne MEMBERS ABSENT: Jay Gross

OTHERS PRESENT (Via Webex): Attorney Steven Mednick, Town Attorney James Baldwin, Members of the

Public both in person and Webex.

#### I. Call To Order

Chair Bryan Cafferelli called the meeting to order at 7:00pm.

#### II. Roll Call

Recording Secretary Pru O'Brien took the roll.

#### III. Pledge of Allegiance

Chair Bryan Cafferelli led the Pledge of Allegiance

#### IV. Approval of Meeting Minutes: April 25<sup>th</sup>. May 11<sup>th</sup>. May 12<sup>th</sup>. May 17<sup>th</sup>

Commissioner Wynne made a motion to approve the minutes in Item IV. Commissioner Mitola seconded the motion.

The motion carried unanimously.

#### V. Public Comment on Proposed Changes to The Charter of The Town of Fairfield

- Steven Sheinberg-15 Flax Road: Chair of DTC. Charter Revision should be non-partisan. There should be no Minority Representation in the RTM and it should stay at 40 members.
- Sharon Pistilli-RTM D3- 107 Lota Drive: These views are her own and do not represent the RTM. Keep the RTM at 40 members and Minority Representation would not represent voters.
- Kevin Flynn- 67 Sachem Road: Constable for 7 years. Doesn't know why the CRC is reducing the number of Constables. RTM should not be reduced. Do not change the Charter.
- Dick Dmochowski-Secretary FECB/FERB-Colonial Drive: Sent email to the CRC regarding changes requested through an email send 11/15/21 regarding alternate members and powers in the State Statute.
- Judy Ewing-98 Sasco Hill Terrace: Concerned about discussions with little due diligence involved. Town Administrator should be hired/fired by BOS and not just First Selectperson. Minority Representation is not democratic.
- Violet Lumani-27 Rhode Island Avenue: No party should have party control and Minority Representation should not be allowed. This is a power grab.

- Dru Georgiadis-RTM D9-321 Puritan Road: address issue of bipartisanship. RTM should stay with 40 members. To reduce the RTM would be gerrymandering. Opposed to Minority Representation.
- Laura Karson-RTM D4-187 Buena Vista Road: Minority Representation is undemocratic. Do not reduce the RTM, they serve the people in the districts
- Christine Brown-RTM D9-82 Rowland Road: Opposes Minority Representation, it ignores democracy and majority rule. Disagreements between parties are good.
- Jay Wolk-RTM D5-1157 Stratfield Road: Constable & RTM, doesn't understand why CRC is going to reduce the Constables. They are called on by probate court. Paid by the person serving papers and not by town. Also doesn't understand why the RTM numbers are redued.
- Patrick Berhene-827 Riverside Drive: Minority Representation is a slap in the face to voters. No reason to take Constables from votes to Executive Branch appointment. Leave the budget process the way it is.
- Bill Hopf-285 Hemlock Hills Road N.: Current size of RTM is governed by RTM, doesn't see a reason to codify into Charter. He knows the CRC worked hard and for a long period of time. It would be a shame for this to not be ratified by voters. If you leave in Minority Representation this will all be for naught.
- Ken Lee-71 Rhoda Avenue: Served on RTM. Applauds Civility section. Left RTM because of
  hostility. Reducing RTM reduces accountability that is necessary. Minority Representation is
  offensive, budget process is fine. Change should be incremental.
- Craig Curley-109 Lakewood Drive: Serves on BOF and has followed budget process for 15 years.
   BOF provides oversight with experts in the field who know what they are doing. Combining budget process is not a good idea. Minority Representation is not a good process.
- John Kuhn-RTM D7-125 Mayweed Road: RTM district is all Democrats. If Minority Representation was in place, he would not have run. If you limit people to bring ideas, it would be a disservice.
- Bill Gerber-RTM D2-25 Shady Hill Road: Has heard comments from people regarding change in the Charter. Those comments can be a risk-change without substantiation. Reducing the RTM goes against CT Statute. Minority party candidates can and have won in districts. 27% of RTM members would have been gifted a seat if this was already in place.
- Joe Sieber-RTM D5-37 Senior Place: Doesn't believe in Minority Representation. Ran and lost and ran and won. If there is minority representation there is no incentive to run.
- Reini Knorr-83 Alberta Street: concerned about reducing the RTM and thinks they should be able to decide how many people are on it. She also objects to Minority Representation.
- Cindy Wapnitsky-41 Valley View Place: CRC members were appointed with pre-existing thoughts. Changes being made are so people can keep political power. It is openly biased against the RTM. Minority rule is undemocratic.
- Alex Durrell-RTM D3-64 Woodcrest Road-Whatever recommendations the CRC has made, he supports.
- Dylan O'Connor-31 Lowell Street-registered Independent. Spent time speaking with people from both parties who think provision is undemocratic by gifting a seat and not by winning at the ballot box. Not for the RTM to have Minority Representation.
- Eve Darcy-827 Riverside Drive: Echoing a letter from Kathy Braun-now is not the time to change the role of the DPW Director, leave engineering degree in. Opposed to combining budget process.
- Roger Aitori-1310 Melville: Please do not amend the constables. Don't delve into BOS when First selectperson is one party and two others are the other party, it doesn't work well.

- Arlene Murphy-23 Millard Street: People don't feel heard. Commission worked hard and the anti-democratic proposals will have to be defeated at the ballot box.
- Elizabeth Zezima-RTM D4-160 Fairfield Woods Road: Concerned public has grown to believe
  this process is to shift power. Recommendations have been sent but she doesn't find them
  reflected. Should be a standing committee on the RTM to vet members of Boards &
  Commissions as it tends to be a friends and family program. RTM should be self regulating and
  combining budget process will dilute it.
- Bill Perugini1080 Fairfield Beach Road: He is a Constable and these are trying times. Doesn't see a strong need to reduce the number of Constables.
- James Oliveri- 157 Bennet Street: Population growing, do not reduce RTM, Minority Represtation is gifting.
- Peter Henkle-160 Larkspur Road: Was confused about what he was reading. Do not reduce the RTM. And the RTM needs legal assistance.
- Jennifer Jacobson-40 Victoria Court: Robert's Rules of Order regarding the three sections drafted as changes. Clarification on language in provisions with boards & commissions and how they adopt their rules. The Town Charter does not dictate to the RTM what they appropriate funds for but in the changes there are sections where it does dictate and wondered if that would need to be approved by the BOF first.
- Selectwoman Nancy Lefkowitz-Redding Road: Opposes Minority Representation in the RTM.
   She was elected to the RTM because she worked hard. She has questions regarding codifying proposals of the CRC prior to it being presented to the Board of Selectmen. Principals of sustainability, how were the changes done and how will it effect the residents. What happen if we do nothing. There is a Civility clause but it does not include equity-it is a failure that the charter represents all people.
- Kathleen Griffin-15 Stonybrook Road: Currently on the BAA and was on the RTM. There is
  professional management needed and she appreciates the step in the right direction but it has
  not been defined. Language is highly political. New position should be modeled after other
  towns, currently Wilton is hiring Town Administrator. Majority of residents would like to have
  this in Fairfield.
- Matthew McKinnis-301 Sasco Hill Road: Reducing the RTM is dangerous. Minority Representation is undemocratic. He worked an RTM election and the candidates who lost should not be gifted a seat. Of all people speaking, only one was in support.
- Sherri Steeneck-75 Parkway: 43 years as resident. On RTM and BOS. RTM unruly body but it is a democratic process. Budget process important and by the time it gets to the RTM it has gotten a lot of discussion. Not convinced there is a need for Town Manager but should be a professional hire. If there is a good moderator on RTM, meetings run smoother. The benefits and risks of doing nothing on charter are important.

With no other members of the public requesting to speak, Chair Bryan Cafferelli thanked everyone for attending and speaking. Chair Cafferelli closed public comment.

#### VI. Preliminary CRC discussion of the Public Hearing

Commissioner Mitola wanted to state that the Constable issues and changes were suggested by the Probate Court.

#### VII. Next Steps

Attorney Mednick told the Commission to think about what they had heard tonight and let him know what should be discussed at the special meeting on Monday, June 6. Attorney Mednick will get a

redline of the changes after Monday's meeting to present to the Board of Selectmen. The Board's approval, whether in part or whole will go on the ballot in November

#### VIII. <u>Adjourn</u>

Commissioner lacono made a motion to adjourn. Commissioner Battista seconded the motion. The meeting adjourned at 9:15pm.

Respectfully submitted,

Pru O'Brien Recording Secretary

#### EMAILS SENT TO CRC FROM 06/06/22 @ 3 PM THROUGH 06/10/22 @ 1 PM

From: Ken Lee <krlee52@gmail.com>
Sent: Monday, June 6, 2022 7:17 PM
To: CRC <CRC@fairfieldct.org>

Subject: Full comments from June 2nd Meeting

#### 6/2/2022

As due to understandable time restraints I was not able to make the points I came to the meeting to make. I ask that this be put in the permanent record. I will check later in case you need me to provide ti again. Thank you.

My name is Kenneth Lee and I live at 71 Rhoda Avenue. My family have lived here for almost 60 years. I attended school here and served several terms as RTM rep.

I would like to applaud the Civility section I heard is being added. Had that been in force during my tenure as an elected official I probably would have continued to serve, but unfortunately that was not the case.

I regret having to be here tonight. When I heard there was a Charter Revision Commission I expected it to review our current Charter and see what needed to be revised to account for changes over the past decade. Clearly that was not the goal. While I can say I have not read the entire document there are a couple of proposals that I would like to address.

When I was a member of the RTM we reduced the size of the body from 50 to 40, and that was a good choice made by the body itself, discussed and voted upon based on mutual bipartisan agreement. Unilaterally reducing the size of the RTM – the legislative body whose charter is to enact laws, be the final arbiter of the budget, and most importantly provide oversight of the town management, reduces the accountability that is necessary. Disenfranchising voters by adding unelected candidates to a legislative body – the so called Minority Party Representation - is frankly un American. When I served we were the minority in both the RTM and BOF, but no one suggested we were entitled to more representation despite losing the elections. And the suggestion that an elected official will only serve and support members of his own party is hugely offensive. I can only wonder why independent candidates can't also get a seat without being elected.

The Budget process is long, if done diligently. I personally attended most of the BOF meetings as I found that it gave me good insight into the issues and the rational behind the decisions. That being said, there were always differing opinions in our body and questions from our constituents to be asked and answered. Again, this feels a lot like taking away the oversight and responsibilities from the RTM – and the BOF also – for no real gain. THE BOF should be able to conduct it's thorough examination of the budget without interference. The budget affects every citizen in town, why do we need to give it short shrift. The suggestion that a privileged few from the RTM would be "allowed" to participate is another step that waters down the representation of the public in the budget process as well as once again making it more difficult to exercise the oversight the body was designed for.

Hiring a professional Chief Administrative Officer is a good idea. However, making this a potentially political appointment is a very poor idea. This person should be hired on merit and be a professional hire – not a political appointee - who serves the entire Town, not simply the First Selectperson. This again provides the type of oversight that has been called for by citizens of all parties as well as unaffiliated voters.

Last, I would hope that these proposals would not be presented to the public as a package deal, but the major changes broken out into specific ballot questions. Many of these proposals are not controversial and many are progress I am sure, but making major systemic changes to the governing structure of a town should not be presented to the public in a 100 page document which most will never read or even understand in many cases.

It is also dangerous. If a change tilts too far in one direction, when the inevitable change of party control occurs the strong temptation will be to do this again but to the betterment of a different party. That might not end well. Changes should be incremental and evaluated for success. Changes should never be made to reduce the balance of power between the executive and legislature branches. And above all, changes should be made only for the betterment of the people, not the initiator of the changes.

Lastly I find it very disappointing that someone in office for such a relatively brief period of time would attempt to remake the town government in such a way as to reduce transparency, sharply reduce the checks and balances currently in place – in many instances eliminate them entirely, sharply eliminate accountability and oversight. All of the communication from the Selectperson's office stresses the need for transparency, the entire election campaign was based on that premise. This is the antithesis of that. While I would like to assume these proposals were made in good faith, I believe a long hard look at them shows that they fall far short of the mark.

Respectfully,

Kenneth Lee

**From:** Judith Ewing < jeewing 8 @ outlook.com >

**Sent:** Tuesday, June 7, 2022 8:11 AM **To:** CRC < CRC @ fairfieldct.org>

**Subject:** Board of Education - 2. 4. C. BOE - 1. and 2.

In my remarks which were distributed at the Public Comment hearing on June  $2^{nd}$ , and appear online in the Backup for the June  $6^{th}$  CRC meeting, in my second point about the BOE 4 seat and 5 seat competitions, I suggested adding the words re: the 5 seat contest "no party shall elect more than three seats in the election cycle" to be consistent with the language in the 4 seat contest.

In re-reading the comment by Jenn Jacobsen, who stated that there was a conflict in the first sentence of the 5 seat contest with the last sentence. She suggested the language should read "Those five (5) candidates receiving the highest number of votes shall be elected." I disagree because of the

majority rule (See GS 9-167a. Minority Representation - the maximum from a 9 member board is 6).

If during the 5 seat election four or five candidates from one party can win, then during the four seat election only one (or two) from that party can win.

But the language for the 4 seat election states that "no party shall elect more than three seats in an election cycle."

Please clarify the language in the 5 seat election so that the maximum from any one party cannot create the impression that there can ever be more than a 6 seat maximum for any party on a 9 member BOE. This issue was mentioned on 6/6 but the language to resolve the conflict was not discussed.

I hope this is resolved at the 6/8 meeting.

Judy Ewing 98 Sasco Hill Terrace Fairfield, CT 06824

**From:** William Gerber < gerber\_william@yahoo.com>

**Sent:** Tuesday, June 7, 2022 3:25 PM **To:** CRC <CRC@fairfieldct.org>

Cc: Mitola, John <jrmitola@aol.com>; Mednick, Steve <smednick01@snet.net>; O'Brien, Pru

<PO'Brien@fairfieldct.org>; Carpenter, Jennifer <JCarpenter@fairfieldct.org>

**Subject:** No logical reason for the Town Charter to reduce the maximum RTM membership

Dear CRC members,

There is no logical reason for the Town Charter to reduce the maximum RTM membership. Following is a summary of some of the points that have been made to you over the past several months:

\*You have not made a substantive case for mandating reduction in RTM membership in the Town Charter.

\*The RTM regulates its own numbers. Years ago, the RTM reduced its own membership to 40, from 50, which was significantly under the (current) Town Charter stipulated ceiling of 56. The RTM body understood its needs at that time and adjusted accordingly, all without a sledge hammer used by a Charter Revision Commission.

\*Mandating a maximum of 30 reps will tie the RTM's hands with regard to operational flexibility. The body will likely consider dividing certain committees that are too broad in scope, or establish new committees to meet the changing needs of our town. It seems like overreach for this CRC to impose its will and choke off possibilities that may be contemplated by the people's body in the near future.

\*Four RTM members per each of 10 districts, approximately one rep per 1,550 residents, provides reasonable odds for meaningful points of intersection between reps and constituents- geographic, age, background, experience, interests, and a myriad of others. Our RTM is smaller- in total and/or per resident- than a number of the (other) best towns to live in Connecticut.

\*Reps each contribute in many different ways, providing a diversity of behind the scenes research, constituent feedback and services, internal caucus deliberation, etc. Some members of your CRC mistakenly believe not speaking out in public at RTM meetings is tantamount to not adding value to the RTM. This is your opinion or speculation, but far from the truth.

\*The RTM has served as the training backbone and invaluable proving ground for higher office in Fairfield. RTM alumni currently in higher office include our state senator, two of three state reps, our first selectwoman, our other selectwoman, and three members of the board of finance, among others. Mandating a reduction in RTM size will thin out our town's bench for town and state-wide offices.

Your CRC should, instead, codify in the Town Charter the RTM's access to essential resources, like objective, independent legal advice and assistance that will allow it to function more efficiently.

Thank you for your consideration.

Sincerely,

Bill Gerber

RTM District 2

From: Laura OBrien < obrien-laura@sbcglobal.net>

**Sent:** Tuesday, June 7, 2022 4:11 PM **To:** CRC <CRC@fairfieldct.org>

**Subject:** Comments for the special meeting on 6/8/22

To Members of the CRC,

As the RTM stands for "Representative" Town Meeting, our charter revision should do its best to preserve representation. With the population growing, reducing the size of this body does not address this. Secondly, mandating minority representation goes even further by actually reducing representation.

Please retain the 40 member RTM with members chosen by the voters in their district.

Laura OBrien

178 Glengarry Rd.

**From:** James Oliveri < joliveri@nyctaxguys.com>

Sent: Tuesday, June 7, 2022 9:18 PM To: CRC < CRC @ fairfieldct.org > Subject: Fairfield Town Charter

Good Evening:

My name is James Oliveri. I currently reside at 157 Bennett St, Fairfield CT 06825.

Although what I am about to say may go against all your hard work, I do want to let you know I appreciate all of the hard work you have put into this charter. As someone who deals with tax law and the tax code on a daily basis, I know getting the verbiage right for something like this is challenging and a tough burden to overcome.

As a new resident (just moved in October of 2021), I was excited to start a new life here in Fairfield, CT. I felt the town has a lot to offer, between good schools, many parks, and nightlife.

However, I am very disappointed that the town deems it fit to lower representation, and instill a minority interest in the legislative body of this town. With regards to the minority interest, I find it disappointing that the town doesn't feel a candidate of my district needs to earn my vote. I do understand the logic behind it, regarding bipartisanship. However, I want to see my potential republican candidate knocking on my door, and my neighbors doors, earning my vote. I believe this is very important, both for the constituency of the town as well as the governing of this town. By imposing a minority interest, you are essentially devaluing my vote, and I believe the rest of the voters would agree.

I like to hear both sides of everything that gets put out there. I question what the Fairfield Democratic Committee posts, and also what the Fairfield Republican committee posts. So far I have read responses from both sides about their talking points, and I am not satisfied with the answers given for this "minority interest."

As for lowering the RTM, admittedly, I cannot watch every RTM meeting. I watched two, and they seem to involve everyone. I cannot really verify that not a lot of people in the RTM do not participate. As far as I know, the Fairfield Democratic Committee won't really give an answer. I would propose, if anything, perhaps lowering the RTM, with the removal of the minority interest. In my opinion though, I feel it's better to have more in the RTM. This way there are more assurances the people of Fairfield get heard.

Should this charter go in front of a ballot as it stands, I would not approve it.

Again, I appreciate you hearing me out and taking the time to read this.

Thank you, and have a good night.

James Oliveri, CPA

----Original Message----

From: Carol Pontrelli < carol.pontrelli@henkel.com>

Sent: Tuesday, June 7, 2022 9:31 PM To: CRC < CRC @ fairfieldct.org >

Subject: Town Charter

To the Charter Revision Commission:

(1) do not reduce the size of the RTM.

Maintain the current size and districts and preserve the process of elections by the people. A smaller-sized RTM will not serve the town well. The RTM is our best form of representative government

(2) I oppose any additional rules of "civility"

We have Robert's Rules of Order and we certainly have the ability as adults to additional manage respectful discussion. There is too high a risk of infringement on Free Speech by way of codifying broad and ambiguous boundaries.

(3) overall, this endeavor to reform the Charter is wholly unnecessary and this is a solution in search of non-existent problems.

Carol Pontrelli 287 Partridge Lane Fairfield Sent from my iPhone

----Original Message----

From: Reini Knorr < reiniknorr 203@gmail.com>

Sent: Wednesday, June 8, 2022 12:07 PM

To: CRC < CRC@fairfieldct.org>

Subject: Fairfield Town Charter Revision

Dear member is the Fairfield Town Charter Revision Commission,

Thank you to all of you for giving up a large amount of your time during the charter revision process and for the opportunity to speak during last week's public comment meeting.

Below is a written version of my comments:

I have to admit that I have not been able to read the 258 pages of the provided back-up material for tonight's meeting, and I was only able to attend a few of the previous meetings.

My main concern is around the artificial and codified reduction of the number of the members of the RTM to 30 when that body has not voiced a desire to further shrink itself beyond the current 40 member level. I suggest that this charter revision commission retain the current maximum number and leave flexibility to the RTM to 'right size' itself as needed. On a personal level, I am pretty

certain that each Fairfield resident personally knows at least one their the four RTM members in their district, which reduces the barrier of access for items of concern.

Further, I agree with all those who objected to the minority presentation provision that was suggested. To enable the RTM to efficiently conduct its business, all members need to be provided with training and the body needs to be given the tools to make informed decisions, such as access to non-partisan legal advice when deliberating ordinances.

As a side note, I would like to state that I think I'm not alone when I say that as a resident who considers herself quite engaged in Fairfield, I feel utterly overwhelmed by the sheer number of initiatives currently underway on top of the general exhaustion brought on by the never ending pandemic. Each of these initiatives deserves our utmost attention, and I fear we will end up with changes and revisions that were not adequately thought through.

If I still have a little time left, I would like to share a couple of items I learned in a recent Civility in Practice webinar hosted by The Institute for Civility in Government at the The Governor M. Jodi Rell Center for Public Service and CT Public on May 26th.

Civility is claiming and caring for one's identity, needs and beliefs without degrading someone else's in the process.

Civility is about more than just politeness, although politeness is a necessary first step. It is about disagreeing without disrespect, seeking common ground as a starting point for dialogue about differences, listening past one's preconceptions, and teaching others to do the same. Civility is the hard work of staying present even with those with whom we have deep-rooted and fierce disagreements.

It is political in the sense that it is a necessary prerequisite for civic action. But it is political, too, in the sense that it is about negotiating interpersonal power such that everyone's voice is heard, and nobody is ignored.

And civility begins with us.

Traditional applications of civility that emphasize manners and behavior over meaningful engagement and shared understanding have led us to a fatal misunderstanding of civility and how to resolve our differences. Please review the section on civility in the proposed charter revision through this lense.

Thank you,

Reini

Reini Knorr 83 Alberta Street, Fairfield, CT 06825 203/331-0922 (h) 203/537-8860 (c) **From:** Laura OBrien <obrien-laura@sbcglobal.net>

**Sent:** Wednesday, June 8, 2022 12:34 PM

**To:** CRC < CRC @ fairfieldct.org >

**Subject:** Comments for the special meeting on 6/8/22

To Members of the CRC,

As the RTM stands for "Representative" Town Meeting, our charter revision should do its best to preserve representation. With the population growing, reducing the size of this body does not address this. Secondly, mandating minority representation goes even further by actually reducing representation.

Please retain the 40 member RTM with members chosen by the voters in their district.

Laura OBrien

178 Glengarry Rd.

**From:** kate braun <klbesq@gmail.com> **Sent:** Wednesday, June 8, 2022 1:09 PM

To: CRC < CRC @ fairfieldct.org >

Subject: Braun Letter to CRC for 6-8-22 meeting

To: Charter Revision Commission

re: Today's meeting (6/8/22 at 4:30)

Please see attached letter for today's final meeting at which I understand proposed changes to the RTM's structure will be voted on.

I'm copying the Chair because some emails in the past have been rejected.

Sincerely, Kathryn L. Braun

June 8, 2022

To: Charter Revision Commission

Re: Let the RTM remain the representative legislative body the Charter established it to be

Dear Chairman Cafferelli and members of the Commission:

I am a former 3-term RTM member and current commissioner on Fairfield's TPZ. I urge you to avoid proposing drastic structural changes to the RTM, which is the legislative branch of our Town Government. In light of scandals and corruption probes in recent years, we need more checks and balances, not less.

The RTM has always been empowered to manage its own size up to the long-standing cap of 56 representatives in 10 districts, and in fact has reduced its own size to the current 40. This approximates about 1 representative per 1,000 voters or 1,500 residents. This far exceeds the per capita RTM size of nearby towns of Greenwich, Darien and Westport (RTM representatives in those towns represent on average of only about 300 voters or 400 residents).

To use the Town Charter to slash the RTM cap from 56 to 30 representatives would effectively disempower our legislative branch from managing its own size and restrict its ability to be truly representative.

Moreover, there is no organic push by our Town's residents to have LESS representation or FEWER representatives to address concerns and take actions for our ever-growing town.

The State legislature will likely continue to impose residential development mandates- which makes our town bigger, with more need for more representatives to adequately represent all town residents.

The proposed minority party representation, on top of the slashed size of the RTM would further erode the RTM's role as our legislative branch by replacing winning candidates with those the voters rejected.

This would result in both major parties' hand-picking insider candidates based on party politics, and leaving the 42% of our Town's voters not in either major party on the outside. Notably, State law exempts minority party representation for geographic district-defined town bodies for a reason-representatives represent all residents of their district as many issues are geographic and neighborhood specific.

It is a good thing to have many people able to run for office and represent their districts- Town residents have more of a voice in the running of our Town, and more of a chance to actually run for office without being members of the political elite from either party.

There certainly doesn't appear to be any groundswell of public desire to change our Town Charter so that it concentrates power in a chosen few, removes Fairfield voters' ability to elect the candidates that best represent their neighborhoods, and have fewer representatives to reach out to.

In my other letters I have both supported and expressed concerns over the revisions being proposed. But the structural changes to the RTM are an order of magnitude larger and thus more harmful if passed. These 2 Charter revisions are not needed, not warranted, not justified, not wanted, and would denigrate the "representative" ability of our Representative Town Meeting.

I ask that these 2 items be omitted from your proposed Charter Revision.

I thank you for your time and attention and the work you have put into this Commission's charge.

Respectfully, Kathryn L. Braun, Fairfield Resident

From: kgriffi1@optonline.net <kgriffi1@optonline.net>

Sent: Wednesday, June 8, 2022 1:17 PM

To: CRC < CRC @ fairfieldct.org >

**Subject:** Chief of Staff as an Appointed Town Officer?

The 'Chief of Staff' position is still listed in the "Appointed Town Officers" section (7.1A.) of the 06-June-2022 proposed Charter. I'm hoping this is a scrivener's error.

This position is not classified as a Town Officer in the current Town Charter let alone mentioned in it. Unlike all the other Appointed Town Officers, there are no qualifications, duties, etc. listed. It's a highly discretionary political position and does not belong with the other professional Appointed Town Officers. It also does not appear to meet the definition of an "Appointed Town Officer" per 1.4C (1).

Please remove it from the list before the document moves on to the Board of Selectmen.

Sincerely, Kathleen Griffin 15 Stonybrook Road

From: Elizabeth Zezima < lizzezimartm@gmail.com>

Sent: Wednesday, June 8, 2022 2:43 PM

**To:** CRC <CRC@fairfieldct.org> **Subject:** The public is being ignored

To the CRC,

You are in the final stretch of your decision making on revisions to our Charter.

By now you have heard from countless voters and residents and the message has been nearly unanimous.

Among other things, you are being asked not to move forward with Minimum Minority Representation on the RTM. This one in particular strikes at the very essence of our country's proud tradition of democratic process.

I have questioned myself as to whether the reason for this is as obvious as it has seemed from the very beginning. Not only because there has been no data driven or analytical study offered as justification, but because a majority on this Commission have been hand selected to favor this outcome. Forcing a solution upon us for a problem that does not exist.

Which brings me to the smear campaign being waged by the Republican Town Committee and our First Selectwoman, targeting those of us who have spoken out against many of the changes being proposed. It feels like a threat. And decidedly un-American.

For the record I am not using anyone else's talking points. Or following a script. It just so happens many of us - R's, D's and U's - see this for what it is. The fact that these opinions are under attack tells me there is a predetermined outcome that is being upset by unexpected scrutiny and pushback.

If our First Selectwoman was being honest and sincere about transparency and bipartisanship she would stay in her lane. Not out herself as the driving force behind the work of this Commission. Which is to grant her expanded powers, guaranteed political appointments at all levels of government, and minority rule for her party on the Legislative Branch.

While I am now convinced this exercise is for the benefit of the Republican party I notice only one of their members showed up at the public hearing to defend this process. He vowed to vote in favor of whatever this Commission puts forth and is approved by the BOS. Ironically, the very essence of the accusations being made against me and others.

I am embarrassed that my town is being dragged through this process. One that would seem like politics as usual in a country that did not enjoy our freedoms or Constitutional process.

Please end this charade and stop ignoring the will of the voters. To gerrymander our town government to favor a political party that has fallen out of favor is not only unethical, it is the height of political deception.

#### Respectfully,

\_\_

Elizabeth Zezima RTM Democratic Majority Leader District 4

Cell: 203.856.4868 lizzezimartm@gmail.com

**From:** Eve Burhenne < Burhenne E@fivestarproducts.com>

Sent: Wednesday, June 8, 2022 3:08 PM

**To:** CRC < CRC @ fairfieldct.org >

**Subject:** Letter to the CRC in advance of the 6/8/22 meeting

To the Charter Revision Commission:

Thank you again for the service you are performing on behalf of the citizens of the town of Fairfield.

I wanted to express my opposition to some of the proposed changes to the charter including (but not limited to!):

Combining Budget sessions of the BOS, BOF, and RTM. Each of these governmental entities has its own role, purpose, procedure & process. The budget process, among the most critical these independent bodies participate in, would be cobbled by the absence of independent review and consideration. "Streamlining", in certain circumstances, is not a bad thing; in this instance, it would be a denigration of this critical undertaking. The budget process should be driven not by whatever is the most expedient, but by the need for exhaustive review from each of these governmental bodies, using its own independent scrutiny, process & perspective.

Removing the requirement that the Director of DPW be a licensed engineer in CT. I realize there is a department of engineers - as well there should be given the glut of issues the town faces, from updating aging infrastructure, climate resiliency planning, the swell of residential housing – but now is not the time to install someone without those necessary qualifications. The director will have enough on his or her plate without having to continually rely on others for expertise. He or she must be fully competent to understand the needs and details of our public works projects and planning short & long term, assembling financial, technical, & operational resources inter/intra-departmentally (as well as outside vendors) to implement the plan for Fairfield's infrastructure.

Reducing the size of the RTM to 30. The RTM, the legislative branch of our town government, should continue to be empowered to set its own size as it has done in the past. Compared to other towns, i.e., Darien, Greenwich, Westport, Fairfield's RTM is already smaller per capita; reducing the RTM to 30 would mean each rep would act for about 2,000 residents (or 1,500 voters). The RTM should remain as is – certainly not be reclassified as a de-facto board or commission – and the proposed changes would hamstring our legislative branch. I see no evidence (I don't count "oh, it's just a good idea"; "some RTM members really don't contribute much..." as adequate research) to support the reduction. If anything, I would love to see a codicil to the charter granting RTM's access to essential resources, like objective, independent legal advice and assistance that will allow it to function more efficiently.

Thank you.

Eve Darcy Burhenne 827 Riverside Drive Fairfield CT 06824

**From:** Karen Wackerman < karenrtm7@gmail.com>

Sent: Wednesday, June 8, 2022 4:35 PM

**To:** CRC <CRC@fairfieldct.org>

**Subject:** Meeting schedule

Please schedule the final meetings of this commission for the evenings so that people who work can attend them. I know that evening meetings are tough - I deal with them all the time with RTM - but the public needs to be able to watch/listen to what is happening.

Thanks you.

Karen Wackerman RTM District 7 Representative 203-984-1673

----Original Message----

From: Carol Waxman < cawaxman40@gmail.com>

Sent: Thursday, June 9, 2022 9:09 AM To: CRC < CRC @ fairfieldct.org >

Subject: Reduction of Members of Legislative Body - RTM

My dear CRC members and Legal Advisor-

I have "attended" many of the CRC meetings and listened as educated proposals were made to clarify and even in some cases reduce the number of members of elected and appointed boards, commissions and committees. I worry that uneducated voters/residents may be swayed to support what the party dominated Board of Selectmen's vote may support as presented on the ballot in November.

I believe in the importance of the three branches of government, especially the legislative and in this case the Representative Town Meeting, which started as a town meeting in New England usually held in a town church in which all residents of the town voted on the budget and other items pertaining to the running of the town efficiently and economically. Simply put as town populations grew, it was determined that representatives be elected to vote for the residents. Several towns chose the non-partisan RTM route, which I proudly support.

This body may form committees to study budget items which may affect different "districts" more than other districts such as the major highway running thru the town or the "town dump" or the police and fire departments.

It is important the impact of different budget items on different areas or voting districts be strongly represented.

Back to Westport's RTM, the committees established by the RTM observe the functioning of the various departments- culture, education, protection for example and at budget time the committee writes a report of the running of the department and budget item and all the reports are circulated in the RTM so the RTM makes informed votes on the budget of the government. This affords the residents more of a say in how the money is spent and the RTM is the most important part of the running of a town such as Fairfield. It is proudly diverse and the RTM with the largest number of representatives can keep in touch with their electors and follow up on the needs and wants of their particular district. As this is virtually a volunteer body of working residents, who need to have time to maintain their own individual lifestyle, it is important to have more representatives per capita to hear all voices. I would hope that the RTM would not be reduced in number.

I thank all who have served on the Commission and I hope you will bear in mind that a lot of thought and work went into the town's constitution and do not cut short the voice of the residents while attempting to bring the "body of work" up-to-date. It is said that democracy moves slowly, please do not streamline the move too much.

Sincerely yours,

Carol Waxman

Fairfield

From: Peter Hinkle <petehinkle04@gmail.com>

**Sent:** Thursday, June 9, 2022 11:38 AM

**To:** CRC <CRC@fairfieldct.org> **Subject:** Question on Reasoning

Hello,

My name is Peter Hinkle. I spoke at the public comment meeting last week and expressed my own concerns with certain revisions to the charter. I'm reaching out now, not to express more concerns, as I've already done, but to ask for the reason for these changes.

The one point I wish I had made last week was to ask why these changes are being introduced. I honestly am just confused, and feel that I may be more comfortable with the changes if I know why, but to my knowledge, that has yet to be done. I've heard that this would increase transparency, but the claim is not substantive and not true.

Once again, as of now, I don't have a problem with the changes, I have a problem with not understanding them. At this point, I only see them from my own lens, but would love to have the neutral one, as well.

Thank you for your service on this commission, for reading my email, and for considering the concerns of all of those who care about this process.

Regards, Peter Hinkle

----Original Message----

From: Jan Carpenter <janc@144h.com> Sent: Thursday, June 9, 2022 11:53 AM

To: CRC < CRC @ fairfieldct.org >

Subject: 10 Year Charter Review Clause

Members of the CRC,

I want to add my voice to that of Commissioner Brogan in advocating for a 10 year Charter Review clause in the CRC.

I give huge credit to the First Selectwoman for convening your body after approximately 15 years. I believe it is, politically, very unlikely that anyone might want to do so, given the potential CRC suggestions that might emerge. It goes to the very essence of that person's future.

I can't imagine that decision was easy and I can't imagine it would be easy for anyone in the future. Nonetheless, I believe it is very much in our best interest of residents to review our "rules" every 10 years. As attorney Mednick indicated, it doesn't necessarily have to be a big deal if the issues discussed turn out to be minor. But, it forces a review.

My understanding is that State of CT recommends (if not mandates) it and I think our Charter should as well..

Thank you for listening.

Jan Carpenter

Jan Carpenter janc@144h.com 203 293 8889

----Original Message----

From: suzannemiska (null) <suzannemiska@aol.com>

Sent: Friday, June 10, 2022 10:46 AM To: CRC < CRC @ fairfieldct.org >

Subject: Public comment

#### Dear Members of the CRC:

It's time to schedule a Town Hall meeting in person to allow the public to take the microphone and be heard. To have decisions being made in the vacuum of Webex is not the manner in which a potential change to the way our Town is governed should happen. To allow this committee to work in a vacuum where public comment is in emails or zoom like calls, while insulating for the committee, only breeds distrust and a lack of involvement.

This committee needs to stop any further discussion and decisions and allow the public to address their concerns in a Town Hall environment. If we can have one for the budget by BOF and ones by the BOE why are you so unwilling to welcome the public into the discussion? The First Selectwoman should never have allowed this committee to conduct its business in webex, especially given the magnitude of what the committee she assembled could potentially be changing in our government and ultimately the taxpayer representation.

The public deserves the right to be heard loud and clear! Schedule the meeting now! You owe the residents both transparency, accountability and a chance to be heard on these very important potential changes.

Suzanne Miska

Suzanne Miska Sent from my iPhone

From: Richard Dmochowski < r.dmochowski@sbcglobal.net>

**Sent:** Monday, June 6, 2022 5:25 PM **To:** CRC <CRC@fairfieldct.org> **Subject:** Comment on language editing

Dear Chairman Cafferelli, The entire document needs to be "proof-read". Please see attached for an example. Dick Dmochowski

- Members and terms. The Water Pollution Control Authority shall consist of seven (7) members, one (1) of whom shall be a member of the Board of Selectpersons, and six (6) of whom shall be appointed by the Board of Selectpersons. No more than four (4) of the members appointed by the Board of Selectpersons shall be registered with the same political party. The members appointed by the Board of Selectpersons shall have terms of four (4) years, which shall be staggered so that not more than two (2) terms expire in one (1) year. The Fiscal Officer shall be a member ex officio, without vote.
- Organization and personnel. For purposes of this Charter, the Water B. Pollution Control Authority shall be deemed a commission except where any provision of this Charter of general application to commissions conflicts with a provision of the General Statutes concerning water pollution control authorities. The Water Pollution Control Authority may establish rules and adopt bylaws for the transaction of its business. The clerk of the Water Pollution Control Authority shall keep a record of its proceedings and shall be custodian of all books, papers and other documents of the Water Pollution Control Authority. The Water Pollution Control Authority may employ such personnel as may be required for the performance of its duties and may fix their compensation.

THERE SEEM TO BE Powers and duties. The Water Pollution Control Authority shall have the ANUMBEROF power to:

STILTED LANGUAGE PASSAGES THAT WOULD(1) BEVEFIT FROM " WORDSMITHING

Operate, maintain and manage, and through the Department of Public Works, plan, lay out, acquire, construct, reconstruct, equip, repair, maintain, supervise and manage and, through the Department of Public Works, operate a sewerage system;

WHICH IS IN KEEPING WITH MR. MARSILLIO'S

COMMENTS

Acquire, by purchase, condemnation or otherwise, any real property or interest in real property which it shall determine to be necessary for use in connection with such sewerage system;

Apportion and assess the whole or any part of the cost of acquiring, constructing or reconstructing any sewerage system or portion thereof upon the lands and buildings in the Town which, in its judgment, shall be especially benefited by the system (whether they abut on such system or not), and upon the owners of DICK DMOCHOWSK/such lands and buildings and fix the time when such assessments shall be due and payable and provide that they may be paid in such number of substantially equal annual installments, not exceeding thirty (30), as it shall determine;

- Establish and from time to time revise just and equitable charges or rates for connection with and use of the sewerage system;
- Order any owner or occupant of any real estate to which the sewerage system is available to connect the drainage and sewerage thereof with the system and to disconnect, fill up and destroy any cesspool, privy vault, drain or other arrangement on such real estate for the reception of such drainage or sewerage;

prescribe and which may be acceptable to the Board of Library Trustees, provided no gift which imposes upon the Town an obligation to incur any expense in order to keep, use or maintain the gift may be accepted by the Board of Library Trustees unless it is approved by the RTM. The Board of Library Trustees may establish one (1) or more library funds withany of such property and shall have the exclusive control and management of, may hold title to, and may manage and invest and reinvest, the property in accordance with the Laws of the State governing the investment of trust funds.

(2) -Management of funds. Subject to the terms and conditions upon which any of such property or funds shall be held, the Board of Library Trustees shall transferis permitted to use the gross income (Dividends) at least quarterly to the Chief-Fiscal Officer to be expended by the Town for capital improvements to the library or for such special purposes as may be required to comply with the terms and conditions of any gift or by state standards. The principalprinciple within the funds is not to be used for operational budgeting purposes-and, it is governed and managed solely by the Board of Trustees. The use of any principalprinciple within the funds is to be used only for capital investment improvements (outside of annual operating budgets) and thus voted on by the Board of Trustees. Subject to appropriation, the Board of Library Trustees may employ such agents, experts, and other personnel as it may deem advisable in connection with the administration and management of any of such property or funds.

38.16.10.(2)

## **CHARTER**

**OF THE** 

## **TOWN OF FAIRFIELD**

Proposed Revisions to Board of Selectmen 2Approved: 8 June 2022<sup>1</sup>

Transmittal to Town Clerk: \_\_ June 2022

## **Charter Revision Commission 2021-2022**

Bryan Cafferelli Chris

Christopher Brogan
Vice Chair

Marlene Battista
Secretary

W Cross

Hon. Pamela lacono

Hon. John Mitola

John Wynne

James T. Baldwin
Town Attorney

Steven G. Mednick
Counsel

<sup>&</sup>lt;sup>1</sup> Please Note: This document remains a draft and will continue to undergo due diligence reviews throughout the entirety of this process. We will continue to clarify and align provisions for the duration. This draft is based upon the 11 January 2022 Reorganizational Baseline. This document remains a work in progress and will be reviewed and refined as the CRC and Board of Selectpersons progresses through the process. Please excuse any errors or inaccuracies. Internal references will continued to be addressed and aligned at the final stage in this process.

## **DRAFTING NOTE: CHARTER REVISION COMMISSION WORKING**

Based upon 11 January 2022 Reorganizational Baseline Note from Counsel: This is a work in progress and will be teviewed and refined as the CRC progresses through it process. Please excuse any excuse or inaccuracies. Internal references have not be addressed and won't be until a later stage in this process.

checkers and won't be a second and won't be a second and won't be a second and a se

#### **Table of Contents**

	E I - INCORPORATION, GENERAL POWERS, CONSTRUCTION, ORGANIZATION AND	
§1.1.	ARDS	
-	Title.	
§1.2.	Incorporation and powers.	
§1.3.	Rights and obligationsDefinitions.	
§1.4.	Standards of Conduct.	()
§1.5.		
§1.6.	Rules of Order and Civility.	
§1.7.	Open Meetings and Public Records.	
<b>ADTICI</b>	E II - ELECTED OFFICIALS AND ELECTIONS	1
§2.1.	Application of General Statutes.	14
§2.1.	Rules Pertaining to Electors.	14
§2.2.	Date of Elections and Terms of Office for Elected Offices	14
§2.3.	Minority representation on elected Boards and Commissions	1/
§2. <del>7</del> . §2.5.	Date of Elections and Terms of Office for Elected Offices.  Minority representation on elected Boards and Commissions.  Single office requirement for Elected Town Officials.	1\ 1-
§2.5.	Vacancies in elected offices.	
§2.0. §2.7.	Composition and Election of the Board of Selectpersons and First Selectperson.	
§2.7. §2.8.	Procedures for RTM elections.	
	Procedures for KTW elections.	
<b>ADTICI</b>	E III – REPRESENTATIVE TOWN MEETING	21
RR HUL	Logislative Rody	2i
83.1. 83.2	Membership	2/
§3.2. §3.3.	Annual meetings, organization, and elections	Zi
§3.3. §3.4.	Meetings of the RTM	
§3.4. §3.5.	Ordinances, Resolutions, Orders or Motions:	کا
§3.5. §3.6	Petition for Overrule (Referendum)	
§3.7.	Appeals from the Board of Finance.	
§3.7. §3.8.	Powers Denied to the RTM Error! Bookmark no	ot dofinad
§3.0. §3.9.	Vacancies in the RTM	ת מפוווופט יץ
33.3.	vacancies in the KTM	
ARTICL	E IV - BOARD OF SELECTPERSONS AND THE FIRST SELECTPERSON	3
§4.1.	Executive Authority.	3
§4.2.	The Board of Selectpersons.	
§4.3.	Powers and Duties of the First Selectperson.	
§4.4.	Compensation of executive branch members.	
§4.5.	Regulations.	
§4.6.	Absence, Disability, Vacancy in the Office of First Selectperson.	
ARTICL	E V - OTHER ELECTED OFFICERS, BOARDS AND COMMISSIONS	
§5.1.	Town Clerk.	
§5.2.	Justices of the Peace	
§5.3.		
§5.4.		
§5.5.	Board of Finance.	
§5.6.	Board of Assessment Appeals.	
§5.7.	Town Plan and Zoning Commission.	
§5.8.	Zoning Board of Appeals.	47
ADTICI	E VI. ADDOINTED OFFICEDS AND DOADDS AND COMMISSIONS IN CENTRAL	4.
	E VI - APPOINTED OFFICERS AND BOARDS AND COMMISSIONS IN GENERAL Eligibility for appointed Town office	
§6.1.	Authority and Qualification of Appointed Town Officers	
§6.2. §6.3.	Minority representation on Boards and Commissions.	48 50

§6.4.	Terms of office	
§6.5.	Resigning from appointed office.	51
§6.6.	Vacancies in appointed office.	
§6.7.	Removal from appointed office for cause	
-		
§6.8.	Required Cooperation.	
§6.9.	Compliance.	
§6.10.	Creation of Boards and Commissions	52
ARTICLI	E VII - APPOINTED OFFICERS	53
	Appointed Town Officers	53
3	7	
	Officers Appointed by the First Selectperson  Town Attorney.  Assistant Town Attorneys.  Fiscal Officer and Controller.	52
C - C	The Administration of the First Selective Soliton	53
§7.2.	Town Attorney.	54
§7.3.	Assistant Town Attorneys.	54
§7.4.		
§7.5.	Town Treasurer.	55
§7.6.	Director of Public Works	56
§7.7.	Director of Public WorksBuilding Official.	57
§7.8.	Purchasing Agent	58
	Purchasing Agent.  Director of Human and Social Services.	50
§7.9.	Director of Human and Social Services.	
§7.10.		58
§7.11.	Director of Community and Economic Development.	59
§7.12.	Director of Human Resources	60
	Officers Appointed by the Board of Selectpersons	60
87 17	Constables	61
31.11.	Constables	
	–§7.13. Internal Auditor or Auditors.  Assessor.	00
	-97.13. Internal Auditor or Auditors.	60
§7.14.	Assessor.	61
§7.15.	Tax Collector	61
§7.16.	Tree Warden.	61
§7.17.	Tree WardenConstables.	61
3		
	Officers appointed by other bodies	62
<b>67 40</b>		
§7.18.		
§7.19.	Chief of Police.	63
§7.20.	Fire ChiefPlanning Director	63
§7.21.	Planning Director	64
§7.22.		64
	Conservation Director	
§7.24.		
31.4.	1 O WIT ELOUGICUTE	
4 D T 1 2 1 .	C VIIII ODECIFIC ADDOINTED DO ADDO AND COMMISSIONS AND DED ADDIEST	• ••
AKTICLI	E VIII - SPECIFIC APPOINTED BOARDS AND COMMISSIONS, AND DEPARTMENT	<b>5</b> bb
§8.1.	Boards and Commissions.	66
	( )	
	Bodies appointed by the First Selectperson	67
<b>§8.2.</b>	Conservation Commission.	
	Permanent bodies appointed by the Board of Selectpersons	60
60.0	Police Commission and Department by the Board of Selectpersons.	09
§8.3.	Police Commission and Department.	
§8.4.	Fire Commission and Department.	
§8.5.	Police and Fire Retirement Board.	
§8.6.	Department of Public Works.	71
§8.7.	Board of Health and Public Health Department.	
§8.8.	Parks and Recreation Commission.	
§8.9.	Board of Building Appeals.	
90.5.	DUALU VI DUIIUIIIY APPEAIS	

§8.10.		ence and Erosion Control Board	
§8.11.		/ <sub>•</sub>	
§8.12.			
§8.13.	Human Services Commission and	d Department	76
§8.14.			
	Appointed by Board of Selectpe	ersons and Approved by RTM	78
§8.15. I	Ethics Commission		78
	Self-Perpetuating Board App	proved by Board of Selectpersons	8
§8.16. I	3oard of Library Trustees	2	81
ARTICLE	IX - BUDGET PROCEDURE AND RE	LATED MATTERS	¢82
§9.1.	The Budget Process		84
§9.2.	Date of annual budget meeting		85
§9.3.	Annual Budget Estimates		85
§9.4.	Review and recommendation by I	Board of Selectpersons to the Board of Finance	<b>e.</b> 87
§9.5.		Board of Finance	
§9.6.		RTM: Approved Annual Town Budget	
§9.7.		et.	
§9.8.	Appeals from the Board of Finance	re .	80
§9.9.	Expenditure in excess of appropr	iation forbidden.	80
30.0.			
ARTICLE	X - MISCELLANEOUS		91
810 1	Official Seal		91
810.2	Existing ordinances		91
810.2	Senarate provisions	$\Delta$	01
§10.3.	Submission and effective date		Q1
310.4.	oubiliosion and encouve date.		
	•	1,	
		•	
	Separate provisions.  Submission and effective date.		
	100,		
(	1		
(1)			

## ARTICLE I - INCORPORATION, GENERAL POWERS, CONSTRUCTION, ORGANIZATION AND STANDARDS<sup>2</sup>

#### §1.1. Title<sup>3</sup>.

The Charter of the Town of Fairfield (the "Town") shall be the organic Law of the Town in the administration of its local affairs.

#### §1.2. Incorporation and powers<sup>4</sup>.

All the inhabitants dwelling within the Town, as previously constituted, shall continue to be a body politic and corporate under the name of the Town and shall have all powers and privileges and immunities previously exercised by the Town and not inconsistent with this Charter, the additional powers and privileges conferred in this Charter, and all powers and privileges conferred upon towns under the General Statutes of the State of Connecticut (the "State") as the same may be amended<sup>5</sup>.

#### §1.3. Rights and obligations<sup>6</sup>.

All property, both real and personal, all rights of action and rights of every description and all securities and liens vested or inchoate in the Town as of the date when this Charter shall take effect are continued in the Town, and the Town shall continue to be liable for all debts and obligations of every kind for which the Town shall be liable on the effective date, whether accrued or not. Nothing shall be construed to affect the right of the Town

<sup>&</sup>lt;sup>2</sup> [HISTORY: Adopted by ballot of the Town of Fairfield 11-7-2006, effective 11-27- 2006 ("2006 Charter")<sup>2</sup>. Amendments noted where applicable.] Editor's Note: This enactment supersedes the former Charter adopted by the Representative Town Meeting 11-4-1997, effective 11-24-1997. **Comment from the 2022 Charter Revision Commission:** The annotations set forth herein are not definitive or comprehensive. The effort is a first step in the process of recreating the historical antecedents to the current provisions of the Charter as well as showing the continued impact, if any, of Special Act provision adopted prior to the adoption of the Home Rule Act and Article Tenth of the 1965 Constitution of the State of Connecticut.
<sup>3</sup> NEW (2022).

<sup>&</sup>lt;sup>4</sup> 2022 recodification of current Article I, §1.1. Derived from Chapter I, §1 of "An Act Concerning a Charter for the Town of Fairfield" (1947) ("1947 Act"). Chapter XXIII of the 1947 Act and 1956 Acts included a Town Court; which was amended by §3 of Special Act No. 382 of the Special Act of 1949: ("1949 Act"); further amended by §21 of "An Act Amending the Charter of the Town of Fairfield" (1951) ("1951 Act"); Chapter I, §1 of "An Act Concerning a Charter for the Town of Fairfield" (1956) ("1956 Act"); Chapter I, §1 of the Charter of Town of Fairfield (1975) ("1975 Charter"); and, Article I, §1.1 of the Charter of the Town of Fairfield (1997)("1997 Charter").

<sup>&</sup>lt;sup>5</sup> Derived from Chapter I, §3 of the 1947 Special Act. Further amended by §§1 and 2 of the 1951 Act.

<sup>&</sup>lt;sup>6</sup> 2022 recodification of current Article I, §1.2. Derived from Chapter I, §2 and 3 of the 1947 Act; reenacted by Chapter I, §2 and 3 of the 1956 Act; Chapter I, §2 and 3 of the 1975 Charter; and modified by Article I §1.2 of the 1997 Charter. Chapter I, §4 of the 1947 and 1956 Acts and 1975 Charter included contained a provision vesting the 'legislative power of the Town" in the Representative Town Meeting; said provision as well as the "corporate powers" provision in Chapter I, 3 of the 1947 and 1956 Acts and 1975 Charter were abandoned by the 1997 Charter.

to collect any assessment, charge, debt, or lien for the construction, alteration, or repair of any public improvement.

#### §1.4. Definitions.

- A. Definitions and Titles Generally<sup>7</sup>. The definitions contained in the General Statutes of the State of Connecticut shall govern the interpretation of this Charter; unless otherwise defined herein. Articles and Sections are for the purpose of ready reference and shall not be held to limit, extend or effect the interpretation and meaning of the text.
- **B.** Capitalized terms<sup>8</sup>. The following rule has been used in determining which terms in this Charter are capitalized: All references to particular Town officials, as defined, below, and to particular Town Boards and Commissions are capitalized, while general references are not. For example: The Board of Education shall have the powers and duties conferred on boards of education by the General Statutes.
- **C. Defined terms<sup>9</sup>.** The following terms shall have the meanings set forth in this paragraph unless otherwise specified in this Charter:
  - (1) "Appointed Town Officer<sup>10</sup>" means an employee who heads any Department in the Town, whether established by Charter or the Town Code; has the authority and qualifications set forth in §§§6.1 through 6.2 of this Charter, below; and, is directly accountable to the First Selectperson, unless otherwise set forth in this Charter.
  - (2) "Board" or "Commission<sup>11</sup>". For the purposes of this Charter and except as otherwise provided by Law, the terms "Board" and "Commission" shall include mean all boards, agencies, commissions, authorities or like entities or bodies of the Town, whether elected or appointed. This term does not include advisory committees, tasks forces or other like entities created by appointing authorities for limited duration, purposes or scope.
  - (3) "Board of Selectpersons<sup>12</sup>"means the executive authority of the Town as defined in Article IV of this Charter.
    - (4) "Charter<sup>13</sup>" shall meanmeans the Charter of the Town of Fairfield.
    - (5) "Contracts" shall meanmeans all contractual relations of the Town

<sup>&</sup>lt;sup>7</sup> NEW (2022)

<sup>&</sup>lt;sup>8</sup> 2022 recodification of current Article I, §1.3.A. Derived from Article I, §1.3.A of the 1997 Charter.

<sup>&</sup>lt;sup>9</sup> 2022 recodification of current Article I, §1.3.B. Derived from Article I, §1.3.B of the 1997 Charter

<sup>&</sup>lt;sup>10</sup> See, Article Articles VI and VII, §7.2, below.

<sup>&</sup>lt;sup>11</sup> NEW (2022).

<sup>12</sup> NEW (2022)

<sup>13</sup> NEW (2022)

(including the Board of Education), including, without limitation, purchase contracts, lease contracts, and service contracts, including but not limited to, agreements, memoranda of understanding, memoranda or agreement, letters of understanding, side letters and other agreements <sup>14</sup>.

- **(6)** "Day(s)<sup>15</sup>" means calendar days; unless, otherwise specifically set forth in this Charter. Moreover, where a Day set forth in this Charter falls on a weekend, holiday or day when the Town is closed for business, the deadline shall be extended through the close of the next Town business day; unless otherwise required by law
- (7) "Department<sup>16</sup>" means any major functional or administrative division of the Town, including any offices, agencies, bureaus or other descriptions serving such purpose as may be set forth in the budget of the Town. When used within the section establishing or describing the duties of the particular department or its related Board or Commission, the term "Department" shall apply exclusively to the functional division referred to in that section.
- (8) "Elected Town Officials" means an individual who holds an elected municipal office as set forth in §2.3.A, below, and as further set forth in this Charter.
  - (9) "Elector<sup>17</sup>" shall have the meaning contained in the General Statutes.
- (10) "First Selectperson" shall meanmeans the chief executive officer of the municipality, as required by the General Statutes 18 and as set forth in this Charter.

<sup>&</sup>lt;sup>14</sup> 2022 modification and recodification of current Article VI, §6.1.C(1) (2006)(third sentence).

<sup>&</sup>lt;sup>15</sup> NEW (2022).

<sup>&</sup>lt;sup>16</sup> NEW (2022).

<sup>&</sup>lt;sup>17</sup>-C.G.S. §9-1. Definitions. (e) "Elector" means any person possessing the qualifications prescribed by the Constitution and duly admitted to, and entitled to exercise, the privileges of an elector in a town". Comment of the 2022 Charter Revision Commission. C.G.S. §9-1. Definitions. (e) "Elector" means any person possessing the qualifications prescribed by the Constitution and duly admitted to, and entitled to exercise, the privileges of an elector in a town". C.G.S. §9-12. Who may be admitted. (a) Each citizen of the United States who has attained the age of eighteen years, and who is a bona fide resident of the town to which the citizen applies for admission as an elector shall, on approval by the registrars of voters or town clerk of the town of residence of such citizen, as prescribed by law, be an elector, except as provided in subsection (b) of this section. For purposes of this section a person shall be deemed to have attained the age of eighteen years on the day of the person's eighteenth birthday and a person shall be deemed to be a bona fide resident of the town to which the citizen applies for admission as an elector if such person's dwelling unit is located within the geographic boundaries of such town. No mentally incompetent person shall be admitted as an elector. (b) Any citizen who will have attained the age of eighteen years on or before the day of a regular election may apply for admission as an elector. If such citizen is found to be qualified the citizen shall become an elector on the day of the citizen's eighteenth birthday. The registrars shall add the name of any person applying under this subsection, if found qualified, to the registry list and, if applicable, to the enrollment list, together with the effective date of his registration. The registrars may place the name of each such person at the end of the registry and enrollment lists for the voting district.

<sup>&</sup>lt;sup>18</sup> Required by C.G.S. § 7-193(a)(2)(C).

- (11) "General Statutes" or C.G.S.<sup>19</sup>" means the official General Statutes of Connecticut under arrangement of the 1958 Revision as amended and updated from time to time. Where chapter references are made, they are to chapter designations as of January 1, 2006. In the event a chapter designation is changed by the publishers of the General Statutes, the new chapter references shall be substituted for the chapter references contained in this Charter.
- (12) "Law" includes, but means, although is not limited to, decisions of courts and administrative bodies (or any agreements sanctions and bodies), federal or state legislative enactments, Ordinances and Regulations, including all applicable rules contained therein.
- (13) "Majority Vote of the RTM<sup>20</sup>" means more than half of the votes have been cast by the members at a meeting of the RTM at which a quorum is present.
- (14) "Meeting<sup>21</sup>" shall have the meaning set forth in C.G.S. §1-200(2), including meetings by means of electronic equipment, as may be amended from time to time.
- "Meeting (or Hearing) Notice<sup>22</sup>" means a notice posted as required by the General Statutes, including posting of regular meetings with the Office of the Secretary of the State and the Town Clerk, as well the requirement pertaining to special and emergency meetings as well as public hearings, in the same manner as set forth in §1.4.C(17) of the Charter, below, and as may be established by Ordinance or written rules of the RTM or any Board or Commission insofar as such notice requirements are in accordance with the General Statutes. In addition to the above-referenced postings, the Town may post notices on its web-site or notify Elected Town Officials, members of the RTM or members of Boards and Commissions through other electronic media and, if otherwise required by Law, said notice shall be published in a daily newspaper of general circulation distributed in the Town. A Meeting Notice (including public hearings) shall state the time and place thereof and shall be published at a minimum in compliance with the General Statutes or by a more stringent requirement as may be set forth in this Charter. Meeting or Hearing Notice may also be governed by regulatory practices and procedures set forth in the General Statutes<sup>23</sup>.
- (16) "Ordinances" or "Ordinances of the Town" or "Town Code" shall meanmeans the legislative powers of the Town to (1) establish rules or regulations

<sup>&</sup>lt;sup>19</sup> 2022 modification and recodification of current Article I, §1.3.B(4). Derived from Article I, §1.3.B of the 1997 Charter

<sup>&</sup>lt;sup>20</sup> NEW (2022).

<sup>&</sup>lt;sup>21</sup> NEW (2022)

<sup>&</sup>lt;sup>22</sup> NEW (2022).

<sup>&</sup>lt;sup>23</sup> The 2022 Charter revision deletes current Article IV, §4.3.D (2006). Modification of Chapter III, §2 (third sentence) and §4 of the 1947 and 1956 Acts and 1975 Charter; and, Article IV, §4.3.D of the 1997 Charter.

of general municipal application, the violation of which may result in the imposition of a fine or other penalty; (2) create a permanent local law of general applicability; or (3) accomplish other objectives permitted by the General Statutes as may be enacted in accordance with the provisions of this Charter.

- "Public Notice" or "Publication<sup>24</sup>" means a notice for matters other than public meetings or hearings, including the public inspection or availability of any documents or data, as may be required by this Charter. It specifically includes matters where "publication" was required under the prior Charter or the General Statutes, in the event the requirement of the Charter is more stringent than the requirements of the General Statutes. Said Public Notice shall be specifically set forth in this Charter or may be governed by the requirements of the General Statutes. Public Notice shall be posted (1) in the Office of the Town Clerk and other public space or location in the Town Hall designated by the Town Clerk in order to assure sufficient disclosure to and access by the public or other public places as may be determined by the Town Clerk; (2) on the Town web-site or through other electronic media by the Town Clerk; and, (3) if otherwise required by Law, by publication in a daily newspaper of general circulation distributed in the Town. Unless otherwise required by Law, newspaper publication is in the discretion of the Town Clerk or the body providing the Public Notice.
- "Regulation<sup>25</sup>" means a statement of general applicability approved by (18)a Department or Board or Commission, without regard to its designation, that implements, interprets, or prescribes law or policy, or describes the organization, procedure, or practice requirements of any such Department, Board or Commission or the Laws under which they operate.
- "Representative Town Meeting" or "RTM26" means the legislative body of the Town, as required by the General Statutes and as specifically set forth in Article III of this Charter.
- "Special Acts" or "Special Laws" shall meanmeans the acts of the General Assembly pertinent to the Town.
  - (21) "State" or "Connecticut" shall meanmeans the State of Connecticut.
- (22)"State Constitution" shall meanmeans the Constitution of the State of Connecticut.
  - "Town" means the Town of Fairfield. (23)

<sup>&</sup>lt;sup>24</sup> NEW (2022)

<sup>&</sup>lt;sup>25</sup> NEW (2022). Derived from C.G.S. §4-166(16).

<sup>&</sup>lt;sup>26</sup> NEW (2022) Required by C.G.S. §7-193(a)(1)(C).

- **(24)** "Town Office" means any position in Town government which is described by this Charter or the Town Code except membership on the Representative Town Meeting<sup>27</sup>.
- (25) "Town Officer" means an individual elected or appointed to a Town office Office, including any Appointed Town Officer whether established by this Charter or the Town Code, other than as a member of a Board, or Commission<sup>28</sup>.
- (26) "Town Official" means any Town Officer and any member of an Board or Commission or the individual members thereof, including any Appointed Town Officer whether established by this Charter or the Town Code<sup>29</sup>.
- (27) Where reference is made to the word "shall" the legislative intention is to make the function a mandatory or imperative obligation for the official or entity charged with an obligation under this Charter or under the Town Code. It is recommended that to avoid any doubt the word "must" should be used in order to impose clarity on the concept of obligation<sup>30</sup>.

30 NEW (2022)

<sup>&</sup>lt;sup>27</sup> 2022 recodification of current Article I, §1.3.B(1). Derived from Article I, §1.3.B of the 1997 Charter

<sup>&</sup>lt;sup>28</sup> 2022 recodification of current Article I, §1.3.B(2). Derived from Article I, §1.3.B of the 1997 Charter

<sup>&</sup>lt;sup>29</sup> 2022 recodification of current Article I, §1.3.B(3). Derived from Article I, §1.3.B of the 1997 Charter

#### §1.5. Standards of Conduct<sup>31</sup>.

- **A. Declaration of Policy**<sup>32</sup>. Elected and appointed Town officers Officials, RTM members, members of Boards and Commissions, and all employees of the Town shall demonstrate by their example the highest standards of ethical conduct, to the end that the public may justifiably have trust and confidence in the integrity of government. As agents of public purpose, they shall hold their offices or positions for the benefit of the public, shall recognize that the public interest is their primary concern, and shall faithfully discharge the duties of their offices regardless of personal considerations.
- B. Conflict of Interest<sup>33</sup>, No Town Officer or employee or Official, RTM member or any member of any Board or Commission employee shall:
  - (1) Solicit or accept any gift, directly or indirectly, whether in the form of money, loan, gratuity, favor, service, thing or promise, or in any other form, under circumstances in which it can reasonably be inferred that the gift is intended to influence the Town officerOfficial, employee, or member in the performance of official duties; (Nothing in this paragraph shall preclude the solicitation or acceptance of lawful contributions for election campaigns.).
  - (2) Disclose confidential information gained by reason of the office or position or use such information for the personal gain or benefit of anyone;
  - (3) Knowingly have or acquire any financial interest or any personal beneficial interest, direct or indirect, in any contract or purchase order for any supplies, materials, equipment or contractual services furnished to or used by the Town in connection with any project, matter or thing which comes within the Town officer's Official's, employee's, or RTM member's jurisdiction or the jurisdiction of the Board, Commission or any other body of which the person is a member (unless such interest is acquired through being the lowest responsible bidder after public advertisement); or
  - **(4)** Engage in any business transaction or activity or have a financial interest, direct or indirect, which is incompatible with the proper discharge of the official duties or which may tend to impair the independence of judgment in the performance of the Town officer's Official's, employee's, or RTM member's official duties 34.

<sup>31 2022</sup> recodification of current Article XI entitled "Standards of Conduct.

<sup>&</sup>lt;sup>32</sup> 2022 recodification of current Article XI, §11.1 (2006). Derived from Article XI, §11.1 of the 1997 Charter,

<sup>&</sup>lt;sup>33</sup> 2022 recodification of current Article XI, §11.2 (2006). Derived from Chapter XXV of the 1975 Charter; and, Article XI, §11.2 of the 1997 Charter,

<sup>&</sup>lt;sup>34</sup> Subsections C. and D. may be derived from Chapter II, §6 of the 1947 and 1956 Acts.

- **C. Disclosure of Interest**<sup>35</sup>**.** Any Town Officer or employee Official, RTM member, or member of any Board or Commissionemployee who possesses or who acquires such private interest as might reasonably tend to create a conflict with the public interest shall make disclosure thereof to such Board, Commission or body and such person shall be disqualified from action on any matter involving the private interest.
- **D.** Fair and Equal Treatment<sup>36</sup>. No Town Officer or employeeOfficial, RTM member, or member of any Board or Commissionemployee shall use an official position to secure or grant special consideration, treatment, advantage, privilege, or exemption to himself or herself or to any person beyond that which is available to every other person. Moreover, no town Officer or employee, RTM member, or member of any Board or Commission shall knowingly impugn the reputation of a member of the public at a public setting or event [37][S1], [S2], [39]. This provision is not intended to prevent an RTM member from properly representing the people of the member's district.
- E. Penalties and Disciplinary Action for Violations<sup>39</sup>. The failure to comply with, or any violation of, the standards of conduct established by this Charter shall be grounds for the removal from office, discharge from employment of the offending Town Officer, employeeOfficial, RTM member, or member of any Board or Commissionemployee, and the Board of Selectpersons in its discretion may void any contract entered into or adopted in violation of this Charter. The Board of Selectpersons or the Ethics Commission may recommend disciplinary measures for RTM members who fail to comply with, or who violate, these standards, but the RTM retains the final authority to discipline its members.

#### §1.6. Rules of Order and Civility<sup>40</sup>.

Elected Town Officials, Appointed Town Officers and Town employees shall treat members of the public with respect and expect the same in return in official interactions as well as in other in-person or virtual/electronic interactions[53]. The Town is committed to maintaining

<sup>&</sup>lt;sup>35</sup> 2022 recodification of current Article XI, §11.3 (2006). Derived from Article XI, §11.3 of the 1997 Charter,

<sup>&</sup>lt;sup>36</sup> 2022 recodification of current Article XI, §11.4 (2006). Derived from Article XI, §11.4 of the 1997 Charter.

<sup>&</sup>lt;sup>37</sup> NEW (2022). Comment of the 2022 Charter Revision Commission: In the model ethics code of the State of Florida this provision usually includes the following presumption language: "If an official or employee believes his or her accusation to be true, and then learns that it was false, even in part, he or she should apologize in the same forum the accusations were made. A failure to so apologize within a reasonable period of time after learning of the falseness of the accusations will create the presumption that the conduct was fully intentional."

<sup>&</sup>lt;sup>38</sup> NEW (2022). **Comment of the 2022 Charter Revision Commission:** In the model ethics code of the State of Florida this provision usually includes the following presumption language: "If an official or employee believes his or her accusation to be true, and then learns that it was false, even in part, he or she should apologize in the same forum the accusations were made. A failure to so apologize within a reasonable period of time after learning of the falseness of the accusations will create the presumption that the conduct was fully intentional."

 <sup>39 2022</sup> recodification of current Article XI, §11.5 (2006). Derived from Article XI, §11.5 of the 1997 Charter.
 40 NEW (2022).

orderly administrative processes and in keeping Town administrative offices free from disruption.

- **A.** The Workplace and Town Operations. In the workplace and other official interactions this Charter promotes mutual respect, civility and orderly conduct among Town employees, Elected Town Officials, Members of the RTM, Appointed Town Officers and the public. This section is not intended to deprive any person of his or her right to freedom of expression, but only to maintain, to the extent possible and reasonable, a safe, productive, and harassment-free workplace for Town staff and a safe and non-threatening environment for the public. The Town encourages all parties to engage in professional, respectful, and courteous communication and discourages hostile, intimidating, or otherwise disruptive actions.
- B. Public Meetings: Meeting Decorum and Procedures. The Town is committed to the democratic process, the rule of law, individual rights of expression, robust debate, and tolerance for disparate views—and the building of better community relationships through increased empathy, greater awareness and decreased reactivity The Town's elected and appointed Boards and Commissions, the RTM and other public bodies and various community groups all convene public meetings to address, from time to time, controversial issues that may engender passionate and often conflicting opinions. An atmosphere of incivility and disrespect at these meetings can stifle participation and debate, threaten the quality of decisions and undermine the local democratic process.
- **C.** Rules of Order. In order to effectuate these provisions of the Charter, the Town may adopt Ordinances generally governing the conduct of public meetings in accordance with this Charter.
  - (1) Parliamentary Guidance: The General Rule<sup>41</sup>. Robert's Rules of Order shall, as a general rule, regulate the conduct of all meetings of the RTM and all elected and appointed Boards and Commissions of the Town, unless the RTM or particular Board or Commission otherwise specifies.
  - (2) Adoption of Rules. Notwithstanding the foregoing, the RTM and each elected and appointed Board and Commission shall may adopt[ISS][IS6] rules of order in order to conduct public meetings and government business in a civil and orderly environment. The rules shall be adopted by a vote of two-thirds (2/3rd) of the members of the Board or Commission, following review by the Town Attorney to ensure that the rules are based on best practices in parliamentary procedure and consistent with the open meeting requirements of the General Statutes and this Charter.

<sup>&</sup>lt;sup>41</sup> Recodification of current Article VIII, §8.1.B(4)[Elected Boards and Commissions]; and, Article X, §10.1.B(4)[Appointed Boards and Commissions] (2006). Derived from Article VIII, §8.1.B and Article X, §10.1.B of the 1997 Charter.

- **D.** The Role of the Presiding Officer. The Presiding Officer of the RTM and each elected and appointed Board and Commission shall be responsible for maintaining the decorum at public meeting and for the uniform enforcement of rules of conductorder.
- E. Compliance with Rules of Order and Decorum. Likewise, all persons who attend a public meeting shall comply with any lawful order of the Presiding Officer to enforce rules of order and decorum. In all circumstance, members of the public and all public officials shall be expected to follow the rules of the body and shall not engage in disorderly conduct, uncivil language or actions as may be defined by Ordinance, Regulation or rules of order of the body.
- **F. Breach of Rules.** In the event any person breaches the rules of <u>order pertaining to</u> civility in a manner that disturbs, disrupts, or otherwise impedes the orderly conduct of the Meeting, the Presiding Officer shall order that person to cease such conduct. The Presiding Officer has the authority to order a member of the public, public official or member of the body to leave the public meeting in the event <u>of</u> continued violations following an initial order from the presiding officer. If said initial order to cease the offending conduct is not obeyed and said conduct continues in spite of an escalation of additional orders from the Presiding Officer, the party may be removed from the Meeting. Removal of a person at an in-person event may be facilitated by a Sergeant at Arms or law enforcement officer. At a virtual or hybrid meeting the presiding officer may block the person from participation. Members of appointed Boards or Commissions may be subject to removal in accordance with the provisions of §6.7.A of this Charter, below.

# §1.7. Open Meetings and Public Records<sup>42</sup>.

- **A.** Records<sup>43</sup>. Each Elected and Appointed Board and Commission shall keep a complete and accurate record of its official acts, votes, meetings, and proceedings and shall have custody of its correspondence, files and other records and shall designate one of its members or its clerk to keep such record. The minutes and recordings of Boards and Commissions shall be public records, in accordance with the General Statutes, and shall be open for public inspection (A) at the office of the Town Clerk, during regular business hours; and, (B) on the Town website in compliance with the requirements of Law.
- **B.** Open and Public Meetings<sup>44</sup>. All Meetings of Elected and Appointed Boards and Commissions and all committees, task forces or other like entities, shall be open to the public except for executive sessions permitted by the General Statutes, and all appointed Boards and Commissions, and all committees, task forces or other like entities shall comply with state freedom of information laws unless otherwise provided by the General Statutes or Law.

\_

<sup>&</sup>lt;sup>42</sup> NEW (2022).

<sup>&</sup>lt;sup>43</sup> 2022 recodification and modification of current Article VIII, §8.1.B (2) and Article X, §10.1.B(2), derived from Chapter XXIV, §1 and §2 of the 1947 and 1956 Acts; and, Chapter XXVI, §2 of the 1975 Charter.

<sup>&</sup>lt;sup>44</sup> 2022 recodification and modification of current Article VIII, §8.1.B(3) and Article X, §10.1.B(3), derived from Chapter II, §6 of the 1947 and 1956 Acts; and, Chapter II, §5 of the 1975 Charter.



# CHARTER OF THE TOWN OF FAIRFIELD ARTICLE II - ELECTED OFFICIALS AND ELECTIONS

# §2.1. Application of General Statutes<sup>45</sup>.

Unless otherwise, specifically set forth in this Charter, the General Statutes, as amended from time to time, relating to elections, including, without limitation, residency requirements and nomination of candidates, shall be applicable to all elections held in accordance with the provisions of this Charter. The nomination and elections of all Federal, State and Town elected officials shall be conducted as prescribed by the General Statutes and as further set forth in this Charter.

#### §2.2. Rules Pertaining to Electors.

- **A.** Eligibility to Vote<sup>46</sup>. Each Elector of this State who shall reside within the limits of the Town upon the date of any election, and who shall be qualified to vote therein, shall be an Elector of the Town.
- **B.** Eligibility to serve as an Elected Town Official<sup>47</sup>. No person shall be eligible for nomination or election to office as an Elected Town Official who is not an Elector of the Town, in accordance with the General Statutes<sup>48</sup>.
- **C. Prepared Lists of Electors**<sup>49</sup>. The Registrars of Voters shall prepare lists of Electors qualified to vote therefore, in the manner prescribed by the State Constitution, the General Statutes and any Special Acts applicable to the Town.
- D. Effect of ceasing to be an Elector or Resident of a District: Vacancy; Exception.
  - (1) General Rule<sup>50</sup>. If any Elected Town Official ceases to be an Elector of the Town, the office shall become vacant, including the position of district representative on the Representative Town Meeting<sup>51</sup>.

<sup>&</sup>lt;sup>45</sup> NEW (2022)

<sup>&</sup>lt;sup>46</sup> NEW (2022).

<sup>&</sup>lt;sup>47</sup> 2022 recodification of current Article II, §2.1.A (2006). Derived from Chapter II, §3 of the 1947 and 1951 Acts; Chapter II, §7 of the 1956 Act; Chapter II, §3 of the 1975 Charter; and, Article II, §2.1.A of the 1997 and 2006 Charters.

<sup>&</sup>lt;sup>48</sup> **Comment of the 2022 Charter Revision Commission.** At the time of adoption, the applicable statute is C.G.S. §9-186.\_

<sup>&</sup>lt;sup>49</sup> NEW (2022).

<sup>&</sup>lt;sup>50</sup> 2022 modification and recodification of current Article II, §2.1.B (2006). Derived from Chapter II, §3 of the 1947 and 1951 Acts; Chapter II, §7 of the 1956 Act; Chapter II, §3 of the 1975 Charter; and, Article II, §2.1.B of the 1997 and 2006 Charters.

<sup>&</sup>lt;sup>51</sup> 2022 recodification and consolidation of current Article IV, §4.2.D (2006).

(2) Exception: Change of Residence<sup>52</sup>. In the event a member of the Representative Town Meeting remains an Elector of the Town yet moves from the district from which the member was elected to serve, said member may continue to serve until the next election of RTM members.

#### §2.3. Date of Elections and Terms of Office for Elected Offices.

- A. Elected Town Officials<sup>53</sup>. The Elected Town Officials are:
  - (1) The First Selectperson;
  - (2) Two (2) additional members of the Board of Selectpersons;
  - (3) Town Clerk:
  - (4) Nine (9) members of the Board of Finance;
  - (5) Nine (9) members of the Board of Education;
  - (6) Seven (7) members of the Town Plan and Zoning Commission;
- (7) Three (3) alternate members of the Town Plan and Zoning Commission;
  - (8) Five (5) members of the Zoning Board of Appeals;
  - **(9)** Three (3) alternate members of the Zoning Board of Appeals;
  - (10) Five (5) members of the Board of Assessment Appeals; and,

The Registrars of Voters are also considered to be Elected Officials. Justices of the Peace are also recognized by this Charter as Elected Officials<sup>54</sup>.

- **B.** Representative Town Meeting. There shall be thirty (30) members of the Representative Town Meeting divided amongst ten (10) districts by the RTM as set forth in §3.2.A(1) of this Charter, below.
- C. Date of Town Elections and Term of Office<sup>55</sup>. A meeting of the Electors of the Town for the election of Elected Town Officials shall be held on the first (1<sup>st</sup>) Tuesday after the first (1<sup>st</sup>) Monday in November in each odd numbered year, as follows:
  - (1) In November 2023, and in the odd numbered years thereafter, as the term of office shall fall:

<sup>54</sup> Recodification of current Article VII, §7.2.A (2006)(Establishment Clause.

<sup>&</sup>lt;sup>52</sup> 2022 recodification and consolidation of current Article IV, §4.2.D (2006). Derived from Chapter II, §3 of the 1947 and 1956 Acts and 1975 Charter; and, Article IV, §4.2.D of the 1997 Charter. In addition, this provision, consolidates and recodifies current Article II, §2.6.B(2) (2006). Derived from Article II, §2.6.B of the 1997 Charter.

<sup>&</sup>lt;sup>53</sup> NEW (2022).

<sup>&</sup>lt;sup>55</sup> 2022 recodification of current Article II, §2.3.A (2006). Derived from Chapter II, §4 of the 1947 Act, which included a Monday election day. The November election dated was established in §4 of the 1951 Act and reconfirmed in Chapter II, §4 of the 1956 Acts and 1975 Charter and Article II, 2.3 A of the 1997 and 2006 Charters.

- (a) First Selectperson for a term of four (4) years<sup>56</sup>;
- (b) Two (2) additional member of the **Board of Selectpersons** for a term of four (4) years<sup>57</sup>;
- (c) Thirty (30) members of the **Representative Town Meeting**, for a term of two (2) years, as further set forth in §3.2.A of this Charter, below<sup>58</sup>;
- (d) Town Clerk, for a term of four (4) years<sup>59</sup>;
- (e) Three (3) members of the **Board of Finance**, for a term of six (6) years, as further set forth in §5.3.A and §5.5 of this Charter, below 60:
- (f) Five (5) members of the **Board of Education**, for a term of four (4) years, as further set forth in in §5.3.A and §5.4 of this Charter, below 61;
- (g) Three (3) members of the **Town Plan and Zoning Commission** for a term of four (4) years, as further set forth in §5.3.A and §5.7 of this Charter, below 62;
- (h) One (1) member of the **Town Plan and Zoning Commission** for a term of two (2) years, as further set forth in §5.3.A and §5.7 of this Charter, below <sup>63</sup>;
- (i) Three (3) members of the **Zoning Board of Appeals**, for a term of four (4) years, as further set forth in §5.3.A and §5.8 of this Charter, below <sup>64</sup>:
- (j) One (1) member of the **Zoning Board of Appeals**, for a term of two (2) years, as further set forth in §5.3.A and §5.8 of this Charter, below <sup>65</sup>; and,
- (k) Two (2) members of the **Board of Assessment Appeals**, for a term of four (4) years, as further set forth in §5.3.A and §5.6 of this Charter, below <sup>66</sup>.
- (2) In November 2025, and in the odd numbered years thereafter, as the term of office shall fall:

<sup>&</sup>lt;sup>56</sup> 2022 recodification of current §1.4.A. and §2.3.C (2006). Note: The four-year term commenced in 2007.

<sup>&</sup>lt;sup>57</sup> 2022 recodification of current §1.4.A and §2.3.C (2006). Note: The four-year term commenced in 2007.

<sup>&</sup>lt;sup>58</sup> 2022 recodification of current §1.4.A and §2.6.E (2006). Also, recodification of current Article II, §2.6.E (2006). Note: The term provisions were established in Chapter III, §3(c) of the 1947 and 1956 Acts and 1975 Charter. There was also a general provision for terms of office for elective officials in Chapter II, §5 of the 1947 and 1956 Acts and the Chapter.

<sup>&</sup>lt;sup>59</sup> 2022 recodification of current §1.4.A. and §2.3.C (2006). Note: The four-year term commenced in 2007.

<sup>&</sup>lt;sup>60</sup> 2022 recodification and clarification of current §1.4.A and §2.3.B (2006).

<sup>61 2022</sup> recodification and clarification of current §1.4.A and §2.3.C (2006).

<sup>62 2022</sup> recodification of current §1.4.A and §2.3.B (2006).

<sup>63 2022</sup> recodification of current §1.4.A and §2.3.B (2006).

<sup>64 2022</sup> recodification of current §1.4.A and §2.3.C (2006)

<sup>65 2022</sup> recodification of current §1.4.A and §2.3.C (2006).

<sup>66 2022</sup> recodification of current §1.4.A and §2.3.C (2006).

- (a) Three (3) members of the **Board of Finance**, for a term of six (6) years, as further set forth in §5.3.A and §5.5 A of this Charter. below <sup>67</sup>:
- (b) Four (4) members of the Board of Education, for a term of four (4) years, as further set forth in in §5.3.A and §5.4 of this Charter, below 68;
- (c) Three (3) members of the **Town Plan and Zoning Commission** for a term of four (4) years as further set forth in §5.3.A and §5.7 of this Charter, below 69;
- (d) Three (3) alternate members of the **Town Plan and Zoning Commission**, for a term of four (4) years, as further set forth in §5.3.A and §5.7 of this Charter, below 70;
- (e) Two (2) members of the **Zoning Board of Appeals**, for a term of four (4) years, as further set forth in §5.3.A and §5.8 of this Charter, below<sup>71</sup>;
- (f) Three (3) alternate members of the **Zoning Board of Appeals**, for a term of four (4) years, as further set forth in §5.3.A and §5.8 of this Charter, below 72;
- (g) Three (3) members of the **Board of Assessment Appeals**, for a term of four (4) years, as further set forth in §5.3.A and §5.8 of this Charter, below<sup>73</sup>; and,
- (h) All Elected Town Officials and members of the RTM, for a term of two (2) years, as further set forth in §3.2.C(1)(c), (h) and (j) of this Charter, below<sup>74</sup>.
- (3) In November 2027, and in the odd numbered years thereafter, as the term of office shall fall:
  - (a) Three (3) members of the Board of Finance, for a term of six (6) years, as further set forth in §6.3.A of this Charter, below 75; and.
  - (b) All Elected Town Officials and members of the RTM as set forth in §2.3.C(1) of this Charter, above, who serve for a term of two (2) or four (4) years, as the case may be<sup>76</sup>.

<sup>75</sup> 2022 recodification and clarification of current §1.4.A and §2.3.A.

<sup>67 2022</sup> recodification and clarification of current §1.4.A and §2.3.B (2006).

<sup>&</sup>lt;sup>68</sup> 2022 recodification and clarification of current §1.4.A and §2.3.C (2006).

<sup>69 2022</sup> recodification of current §1.4.A and §2.3.B (2006).

<sup>&</sup>lt;sup>70</sup> 2022 recodification of current §1.4.A and §2.3.C (2006).

<sup>&</sup>lt;sup>71</sup> 2022 recodification of current §1.4.A and §2.3.C (2006).

<sup>&</sup>lt;sup>72</sup> 2022 recodification of current §1.4.A and §2.3.C (2006).

<sup>&</sup>lt;sup>73</sup> 2022 recodification of current §1.4.A and §2.3.C (2006).

<sup>74</sup> NEW (2022).

<sup>&</sup>lt;sup>76</sup> NEW (2022).

- **(4)** Terms of Justices of the Peace<sup>77</sup>. The forty-five (45) Justices of the Peace<sup>78</sup>, selected as provided in the General Statutes<sup>79</sup>, shall serve a four (4) year term. They shall have such powers as set forth in the General Statutes<sup>80</sup>.
- (5) State Election Registrar of Voters<sup>81</sup>. The Registrars of Voters shall be elected at state elections every four (4) years<sup>82</sup>, as provided in the General Statutes and shall have such powers duties as set forth in the General Statutes<sup>83</sup>.
- **D.** Commencement Date of Terms of Elected Town Officials. Terms of office of all Elected Town Officials declared elected and qualified hereunder shall commence:
  - (1) Elected Town Officials, including the RTM<sup>84</sup>: On the third Monday in November and shall continue until their successors have been elected or otherwise chosen and qualified.
  - (2) State Office Terms of Justices of the Peace<sup>85</sup>. On the first Monday in January after their nomination.
  - (3) Terms of Registrars of Voters<sup>86</sup>. On the Wednesday after the first Monday in January after their election.

#### §2.4. Minority representation on elected Boards and Commissions.

**A.** Even number requirements<sup>87</sup>. Except as provided below with respect to the Board of Education, when an even number of members of a Board or Commission is to be elected, no political party shall nominate, and no Elector shall vote for, more than one-half (1/2) the number of persons to be elected.

 <sup>&</sup>lt;sup>77</sup> 2022 recodification of current Article II, §2.3.E (2006); see also current Article VII 7.2.A (2006)(Establishment Clause). Derived from Chapter II, §1 of the 1947 Act, included 14 justices of the peace.
 <sup>78</sup> Derived from Chapter II, §1 of the 1947 Act, included 14 justices of the peace.

<sup>&</sup>lt;sup>79</sup> **Comment of the 2022 Charter Revision Commission.** At the time of adoption of the Charter, the applicable statutes are C.G.S. §9-183a, b and c and §9-444.

<sup>80 2022</sup> recodification of current Article VII, 7.2.B (2006)(("Powers and Duties Clause").

<sup>81 2022</sup> modification and recodification of current Article II, §2.3.F (2006)(first clause).

<sup>82</sup> Comment of the 2022 Charter Revision Commission; The baseline commencement date was 2008.

<sup>&</sup>lt;sup>83</sup> Comment of the 2022 Charter Revision Commission: At the time of adoption of the Charter the Registrars of Voters is governed by C.G.S. 9-190 and 9-190a (state election cycle). Within Chapter 146; see, C.G.S. §9-164 et seq.)

<sup>84 2022</sup> recodification of current Article II, §2.3.D (2006).

<sup>&</sup>lt;sup>85</sup> 2022 recodification and modification of current Article II, §2.3.E (2006), in compliance with C.G.S. §9-183b.

<sup>86 2022</sup> recodification of current Article II, §2.3.F (2006)(second clause).

<sup>&</sup>lt;sup>87</sup> 2022 recodification of current Article II, §2.2.A (2006). Derived from Article II, §2.2.A of the 1997 and 2006 Charters.

**B.** Uneven number requirements<sup>88</sup>. Except as provided below with respect to the Board of Education, when an uneven number of members of a Board or Commission is to be elected, no political party shall nominate, and no Elector shall vote for, more than a bare majority of the number of persons to be elected.

#### C. Board of Education<sup>89</sup>.

- (1) As pertains to §2.3.B(1)(f). In an election where five (5) candidates shall be elected to the Board of Education, the Electors shall vote for up to five (5) candidates for the Board of Education. Each major or minor party, as defined by the General Statutes and petitioning candidates (if permitted by the General Statutes) may nominate up to three (3) candidates. Those three (3) candidates receiving the highest number of votes shall be declared elected and the two (2) candidates not of the same political party as the party receiving the highest number of votes shall also be declared elected [57].
- (2) As pertains to §2.3.B(2)(b): In an election where four (4) candidates shall be elected to the Board of Education, the Electors shall vote for up to four (4) candidates for the Board of Education. Each major or minor party, as defined by the General Statutes and petitioning candidates (if permitted by the General Statutes) may nominate up to three (3) candidates. Those three (3) candidates receiving the highest number of votes shall be declared elected. The final member shall be and the one (1) candidate withnot of the next same political party as the party receiving the highest number of votes, although such candidate shall not enable any party to elect more than three (3) seats in the election cycle shall also be declared elected.

# §2.5. Single office requirement for Elected Town Officials 90.

No person shall be eligible to serve as an Elected Town Official, including membership on any elected Board or Commission, who is at the same time an elected RTM member, an elected or Appointed Town Officer, Official or an elected state official, or a member of an elected Board or Commission or a permanent appointed Board or Commission. For purposes of this paragraph, the term "Town office" does not include Justices of the Peace, but does include members of the Board of Education.

# §2.6. Vacancies in elected offices.

A. Generally<sup>91</sup>. Except as provided in §2.6.D, with respect to the RTM and §4.6

<sup>&</sup>lt;sup>88</sup> 2022 recodification of current Article II, §2.2.B (2006). Derived from Article II, §2.2.B of the 1997 and 2006 Charters.

<sup>89 2022</sup> recodification of current Article II, §2.2.C (2006). Derived from Article II, §2.2.C of the 2006 Charter.

<sup>&</sup>lt;sup>90</sup> 2022 recodification of current Article II, §2.1.C (2006). Derived from Article II, §2.1.C of the 1997 and 2006 Charters.

<sup>&</sup>lt;sup>91</sup> 2022 modification and recodification of current Article II, §2.5. Derived from Chapter II, §7 of the 1947 Act;

of this Charter with respect to the First Selectperson and Board of Selectpersons, §2.6.D, below, with respect to the RTM, a vacancy in any elected Town office Office 92, including membership on elected Boards and Commissions, shall be filled by the Board of Selectpersons until the vacancy can be filled by election, as follows:

- (1) The vacancy shall be filled at the next Town election if the vacancy occurs prior to the time in which nominations can be made under Chapter 146 of the General Statutes (C.G.S. § 9-164 et seq.). If the vacancy occurs after such time, it may be filled at the next municipal general election for which nominations can be timely made, or at a special election if convened by the Board of Selectpersons or upon application by Electors as provided in Chapter 146 of the General Statutes (C.G.S. § 9-164 et seq.).
- (2) Vacancies shall be filled for a term ending at the same time the vacating member's term would have expired.
- (3) If the person vacating the office was elected as a member of a political party, the vacancy shall be filled during the period of appointment from the membership of the same political party.
- **B.** Vacancies in the Office of Justice of the Peace<sup>93</sup>. Vacancies in the office of Justice of the Peace shall be filled in the manner prescribed in Chapter 146 of the General Statutes (C.G.S. § 9-164 et seq.).
- **C.** Long-term illness or disability of First Selectperson<sup>94</sup>. In addition to the death or resignation of the First Selectperson, a vacancy shall exist in the office of First Selectperson, in the event the First Selectperson is unable to carry out the duties of office for a period of four (4) consecutive months, as certified at the end of that period by the remaining Selectpersons to the Town Clerk. The vacancy shall exist from the date of such certification.
- **D. Method of filling vacancies on the Board of Selectpersons**<sup>95</sup>**.** At any time, a vacancy occurs on the Board of Selectpersons, including First Selectperson, a replacement, who shall be registered with the same political party as the person vacating the office, shall be designated by the remaining Selectpersons. If the Selectpersons designate one (1) of themselves to fill the vacancy, they shall designate another Elector to fill the vacancy of Selectperson so created. If the vacancy is not filled within thirty (30) Days,

further amended by §4 of the 1951 Act and Chapter II, §7 of the 1956 Act; Chapter II, §6 and Article II, 2.5 of the 1997 and 2006 Charters.

<sup>&</sup>lt;sup>92</sup> 2022 recodification of current Article VII, 7.1.H. (2006). Derived from Article VII, §7.1.H of the 1997 Charter <sup>93</sup> 2022 recodification of current Article VII, §7.2.C (2006). Derived from Article VII, §7.2.C of the 1997 Charter.

<sup>&</sup>lt;sup>94</sup> 2022 recodification of current Article VI, §6.3.A (2006). Derived from Article VI, §6.3.A of the 1997 Charter.

<sup>&</sup>lt;sup>95</sup> 2022 recodification of current Article VI, §6.3.B (2006). Derived from Article VI, §6.3.B of the 1997 Charter. See also, **Cook-Littman v. Board of Selectmen of the Town of Fairfield**, 328 Conn. 758,778 (Conn. 2018) in which the Court held that this provision took precedence over the provisions of C.G.S. §9-222.

the vacancy shall be filled in accordance with the procedure set forth in Chapter 146 of the General Statutes for filling vacancies in the office of Selectperson.

#### E. Resignation and vacancies on the RTM<sup>96</sup>.

- (1) Any member may resign by filing a written notice of resignation with the Town Clerk and such resignation shall take effect upon the date specified in the notice or, if none is specified, upon the date of filing.
- (2) Any vacancy in the office of RTM member from whatever cause arising shall be filled for the unexpired portion of the term at a special meeting of the members of the district in which the vacancy occurs, called for that purpose by the Town Clerk within thirty (30) Days after the vacancy occurs, in accordance with the requirements under this Charter pertaining to Meeting Notice.
- (3) If the person previously occupying the office which is vacant was elected as a nominee of a political party, the vacancy shall be filled by a person registered with the same political party.
- (4) The Town Clerk shall provide Meeting Notice, as defined in this Charter, to be delivered to each member not less than three (3) Days prior to the time set for the meeting.
- **(5)** Such meeting shall elect its own chair and clerk and shall vote by ballot. The election of a member as chair or clerk shall not disqualify the member from voting.
- **(6)** A majority of the remaining members from such district shall constitute a quorum, and a majority vote of those present shall elect.
- (7) Each of the remaining members, including the chair and clerk of the meeting, shall have one (1) vote.
- (8) If for any reason such vacancy is not filled within a thirty-day (30) period, the Town Clerk shall report such vacancy to the Moderator and the matter of filling such Vacancy shall be placed on the call of the next regular RTM meeting.
  - **(9)** A Majority Vote of the RTM shall elect a successor on such Vacancy.
- (10) The successful candidate shall be deemed a duly qualified member for the remainder of the unexpired portion of the term upon filing with the Town Clerk a certificate of his election signed by the chair of such special meeting of the district

<sup>&</sup>lt;sup>96</sup> 2022 recodification of current Article II, §2.6.G (2006). Derived from Chapter III, §7 of the 1947 and 1956 Acts and the 1975 Charter; and, Article II, §2.6.G of the 1997 Charter.

or bythe Moderator of the RTM.

**F.** Resigning from elected office<sup>97</sup>. Any elected Town official, except the Town Clerk, may resign by submitting a written notice of resignation to the Town Clerk. The Town Clerk may resign by submitting a written notice of resignation to the Board of Selectpersons. The resignation shall become effective on the date specified in the notice of resignation or, if no date is specified, on the date the notice of resignation is received by the Town Clerk or the Board of Selectpersons, as the case may be.

# §2.7. Composition and Election of the Board of Selectpersons and First Selectperson<sup>98</sup>.

- **A.** There shall be a Board of Selectpersons which shall consist of the First Selectperson and two (2) other Selectpersons, no more than two (2) of whom shall be registered with the same political party<sup>99</sup>.
- **B.** Each major or minor political party, as defined by the General Statutes, and petitioning candidates (if permitted by the General Statutes) may nominate not more than one (1) candidate for First Selectperson and not more than one (1) candidate for Selectperson. Candidates shall be listed separately on the ballot 100.
- **C.** Each Elector may vote for one (1) candidate for First Selectperson and no more than one (1) candidate for Selectperson<sup>101</sup>.
- **D.** The candidate for First Selectperson receiving the highest number of votes shall be elected First Selectperson. The next two (2) candidates, whether for First Selectperson or Selectperson, receiving the highest number of votes shall be elected to the two (2) remaining Selectperson positions<sup>102</sup>.
- **E.** If the candidate receiving the second or third most votes does not take office, then the next highest ranking candidate shall be deemed elected to the office taking into consideration the provisions of §2.7.A of this Charter, above, pertaining to minority party representation 103.

98 2022 recodification of current Article VI, §6.1.A (2006). Derived from Article VI, §6.1.A of the 1997 Charter.
99 2022 recodification of current Article VI, §6.1.A(1) (2006). Derived from Article VI, §6.1.A of the 1997 Charter

<sup>97 2022</sup> recodification of current Article II, §2.4 (2006).

<sup>&</sup>lt;sup>100</sup> 2022 recodification of current Article VI, §6.1.A(2) (2006). Derived from Article VI, §6.1.A of the 1997 Charter.

<sup>&</sup>lt;sup>101</sup> 2022 recodification of current Article VI, §6.1.A(3) (2006). Derived from Article VI, §6.1.A of the 1997 Charter.

<sup>&</sup>lt;sup>102</sup> 2022 recodification of current Article VI, §6.1.A(4) (2006). Derived from Article VI, §6.1.A of the 1997 Charter.

<sup>103 2022</sup> recodification of current Article VI, §6.1.A(5) (2006). Derived from Article VI, §6.1.A of the 1997

# §2.8. Procedures for RTM elections<sup>104</sup>.

#### A. Nomination<sup>105</sup>.

(1) By a political party. The nomination of candidates as members of the RTM shall be in the same manner as provided for the nomination of <u>Elected Town officersOfficials</u> in Chapter 153 of the General Statutes (C.G.S. § 9-372 et seq.). No political party shall nominate more candidates for each district than the total number of members to which a district is entitled.

#### (2) By petition.

- (a) Nomination of a candidate for the RTM may also be made by petition signed in ink on forms approved and provided by the Town Clerk. The petition shall be signed by not less than one (1%) percent of the Electors in the district in which the candidate resides and filed with the Town Clerk not less than fifty-five (55) Days prior to the election.
- **(b)** No petition shall be valid in respect to any candidate whose written acceptance is not noted on or attached to the petition when filed.
- **(c)** No signature on a petition shall be valid or counted if the signer thereof shall have signed other petitions on file with the Town Clerk for more than the number of candidates which a political party may nominate under this Charter for the district in which such signer resides.
- (d) A petition may contain more than one (1) name but not more names than the number of candidates to which the district is entitled.
- (3) The Town Clerk, within the time prescribed by Chapter 153 of the General Statutes (C.G.S. § 9-372 et seq.), shall certify and transmit the names of candidates duly nominated to the Secretary of the State.

#### B. Election 106.

Charter.

Derived from Chapter II, §1 and §4 of the 1947 and 1956 Acts and 1975 Charter. Note: At the time elective officers were nominated and elected as in the special act or, if silent, on the basis of the General Statutes.
 2022 recodification of current Article II, §2.6.C (2006). Petition provision set forth in Chapter III, §5 of the 1947 and 1956 Acts and the 1975 Charter; and, Article II, §2.6.C of the 1997 Charter.

<sup>&</sup>lt;sup>106</sup> 2022 recodification of current Article II, §2.6.D (2006). Modification of Chapter III, §2 (fourth sentence) of the 1947 and 1956 Acts and Chapter III, §5 of the 1975 Charter: "The provisions of the general statutes relating to voting at elections, so far as the same are not inconsistent with this act, shall apply to all elections in the town under this act". The fifth sentence of §2 of the 1947 and 1956 Acts pertaining to certification of voters prior to an election is not in the current charter. There are also elaborate provisions §3 that do not

- (1) At each biennial election for the election of RTM members voting shall be governed by the provisions of the General Statutes as to voting by ballots and voting machines, except as modified by this Charter.
- (2) Electors shall vote <u>for</u> the total number of RTM members to be elected in the district in which the Elector resides in accordance with the provisions of the Ordinance establishing districts and in compliance with the General Statutes addressing the issue of minority party representation <sup>107</sup>.
- (3) The chief election moderator shall forthwith after a biennial election of RTM members, file in the Town Clerk's office a list of members elected, by districts, together with their respective addresses. The Town Clerk shall, upon receipt of such list, forthwith notify all members, in accordance with the notice requirements of this Charter.

#### C. Procedures for tied elections 108.

- (1) In case of a tie vote affecting the election of RTM members, the other newly elected and reelected members from the district in which the tie vote occurs shall by ballot determine which of the tied candidates shall serve as RTM member or members.
- (2) The chief election moderator shall immediately after an election notify the Town Clerk of any and all tie votes, giving the names and addresses of the candidates affected.
- (3) The Town Clerk shall forthwith call a meeting of the other newly elected and reelected members from the district or districts in which a tie vote occurs by causing a notice, in accordance with the notice requirements of this Charter, specifying the object, time and place thereof to each such member not less than three (3) Days before the time set for the meeting.
- **(4)** At such meeting a majority of the members from the particular district shall constitute a quorum, and they shall elect from among their number a chair and a clerk whose right to vote at such meeting shall not be affected by their election to their respective offices.
- (5) The chair and clerk shall count the ballots and the person or persons receiving the highest number of votes shall be declared elected. The chair and clerk shall forthwith make a certificate of the choice and file the same with the Town Clerk.

appear in the current charter. Derived from Article II, §2.6.D of the 1997 Charter.

<sup>&</sup>lt;sup>107</sup> 2022 modification and recodification of Chapter II, §1 of the 1947 Act.

<sup>&</sup>lt;sup>108</sup> 2022 recodification of current Article II, §2.6.F (2006). Note: Tie vote procedures established in Chapter III, §3(d) of the 1947 and 1956 Acts and the 1975 Charter; and, Article II, §2.6.F of the 1997 Charter.

(6)	The membe	r or members	s so chosen	shall thereupon	be deemed
elected and	qualified as a	n RTM memb	er or membe	ers, subject to the	e right of the
RTM to judg	e the election	and qualificati	on of membe	rs as set forth in	§4.2.B of this
Charter.		-			

(7)	If a tie vo	te occurs	at such	meeting th	ie chair ar	nd clerk :	shall f	orthwith
certify same	to the To	wn Clerk,	and the	e matter of	breaking	such tie	vote :	shall be
placed on the	e call and	voted upo	n by the	RTM at its	organizat	ional me	eting.	



# CHARTER OF THE TOWN OF FAIRFIELD ARTICLE III – REPRESENTATIVE TOWN MEETING

# §3.1. Legislative Body<sup>109</sup>.

The legislative power of the Town, including the power to enact Ordinances, shall be vested in the RTM, subject to referendum as provided by the terms of this Charter. The RTM shall constitute a continuing body. The RTM may delegate the power to implement or carry into effect any of the powers set forth in this Charter to any Town Officer, Board, or CommissionOfficial.

# §3.2. Membership.

- A. Districts and Basis of Representation<sup>110</sup>.
  - (1) Districts.
    - (a) The members of the RTM shall be elected by districts<sup>111</sup>.
  - There shall be ten (10) voting districts of the Town, which shall be reapportioned as set forth in §3.2.B, below, for the election of RTM members and shall be as established by Ordinance adopted by the RTM<sup>112</sup>. District representation shall be subject to the provisions of the General Statutes pertaining to minority party representation 113[59].
    - (c) The RTM shall consist of thirty (30) members<sup>114</sup>.
  - **(d)** A district representative on the Representative Town Meeting, upon election, shall be an Elector of the Town and a resident of the district from which elected, subject to the provisions of §3.2.A(1)(d), below) 115,
  - **(e)** In the event a member of the Representative Town Meeting remains an Elector of the Town yet moves from the district from which the

<sup>&</sup>lt;sup>109</sup> 2022 recodification of current Article IV, §4.1.A (2006). Derived from Chapter III, §1 of the 1947 and 1956 Acts and the 1975 Charter; and, Article IV, §4.1.A of the 1997 Charter.

<sup>&</sup>lt;sup>110</sup> NEW (2022). In lieu of current Article II, §2.6.A(1) (first sentence) of the 1947 Act and the 1956 Acts, which reads as follows: "The voting districts of the Town, including the number of districts, for the election of RTM members shall be as established by ordinance adopted by the RTM." Further in lieu of current Article II, §2.6.A(3) (third sentence) of the 1947 Act and the 1956 Acts, which reads as follows: "The RTM shall consist of not more than 56 members."

<sup>&</sup>lt;sup>111</sup> 2022 recodification of current Article II, §2.6.B(1) (2006). Derived from Article II, §2.6.B of the 1997 Charter.

<sup>&</sup>lt;sup>112</sup> 2022 modification and recodification of Article II, §2 (first sentence) of the 1947 Act and the 1956 Acts.

<sup>&</sup>lt;sup>113</sup> NEW (2022)

<sup>&</sup>lt;sup>114</sup> 2022 modification and recodification of Article II, §2.6.A(3).

<sup>&</sup>lt;sup>115</sup> 2022 modification and recodification of current Article II, §2.6.B(2) (2006). Derived from Article II, §2.6.B of the 1997 Charter.

member was elected to serve, said member may continue to serve until the next election of RTM members<sup>116</sup>.

- **(f)** No Elected Town Official <u>or appointed</u>. Appointed Town Officer <u>of Board or Commission member</u> (excluding, however, members of advisory committees, task forces or like entities, as referred to in §1.4.C(2) of this Charter, <u>above</u>) shall be eligible to serve as a voting member of the RTM, and no voting member of the RTM shall hold any other elected or appointed Town <u>officeOffice</u><sup>117</sup>.
- (2) Reapportionment of Voting Districts<sup>118</sup>. After completion of the census of the United States and after any reapportionment of the State General Assembly Districts affecting the Town, voting districts of the Town shall be established by an Ordinance proposed by a committee of the RTM composed of an equal number of members from each party such that the population deviation from the largest to the smallest voting district shall not exceed ten (10%) percent. The redistricting Ordinance adopted by the RTM shall provide for an equal number of members from each district, and, to the extent practicable within the constraints of this Charter and State General Assembly redistricting, shall provide for districts that are located in only one State General Assembly District.
- **B.** Ex-officio non-voting members of the RTM<sup>119</sup>. The RTM shall consist of the elected RTM members. In addition, the First Selectperson, the Town Clerk, the Town Attorney, any Assistant Town Attorneys (in the event they are Electors), the chair of the Board of Education, and the members of the Board of Finance shall be ex officio members of the RTM without vote.
- **C. Right to vote**<sup>120</sup>. The right to vote at RTM meetings shall be limited to RTM members elected as provided in §2.3.B of this Charter.

<sup>&</sup>lt;sup>116</sup> 2022 recodification and consolidation of current Article IV, §4.2.D (2006). Derived from Chapter II, §3 of the 1947 and 1956 Acts and 1975 Charter; and, Article IV, §4.2.D of the 1997 Charter. In addition, this provision, consolidates and recodifies current Article II, §2.6.B(2) (2006). Derived from Article II, §2.6.B of the 1997 Charter.

<sup>&</sup>lt;sup>117</sup> 2022 recodification and modification of current Article II, §2.6.B(3) (2006). Derived from Article II, §2.6.B of the 1997 Charter. and Article IV, §4.2.C (2006). Derived from Article IV, §4.2.C of the 1997 Charter. Consolidates and replaces Article II, §2.6.B(3) (2006), which is derived from Article II, §2.6.B of the 1997 Charter, which reads as follows: "Each RTM member shall fulfill the eligibility requirements of 4.2.D at the time of election."

<sup>118 2022</sup> recodification of Article II, §2.6.A(2) (second sentence) of the 1947 Act and the 1956 Acts.

<sup>&</sup>lt;sup>119</sup> 2022 recodification of current Article IV, §4.2.A (2006) entitled "Composition." Ex officio status of was established in Chapter III, §4 of the 1947 and 1956 Acts and 1975 Charter; and, Article IV, §4.2.A of the 1997 Charter for the following: the three Selectmen; the Town Clerk; the Town Counsel; Chair of the Board of Education and the entire Board of Finance.

<sup>&</sup>lt;sup>120</sup> 2022 recodification of current Article IV, §4.1.B (2006). Derived from Chapter III, §1 of the 1947 and 1956 Acts and the 1975 Charter; and, Article IV, §4.1.B of the 1997 Charter.

- Judge of qualification<sup>121</sup>. The RTM shall be the judge of the election and qualifications of its members.
- Compensation<sup>122</sup>. The RTM members as such shall receive no E. compensation.
- §3.3. Annual meetings<sup>123</sup>, organization, and elections.
- Α. Date of organization meeting 124. An organization meeting of the RTM members shall be held on the fourth Monday in November in each year.
- Election of Moderator: Duties<sup>125</sup>. Each organization meeting shall elect, by В. a vote of the Majority of the RTM, a Moderator. The Moderator shall preside at all RTM meetings and shall hold office for a term of one (1) year and until a successor is elected and has qualified. The Moderator of the RTM shall:
  - preside Preside over all meetings of the RTM and perform such others duties consistent with the office as may be imposed by the RTM, but such Moderator shall not vote more than once on any question; and,
  - (2) Perform such duties consistent with the office or as may be imposed by a Majority Vote of the RTM.
- C. **Deputy Moderator**<sup>126</sup>. Each organization meeting shall elect, by a Majority Vote of the RTM, from among its voting members a Deputy Moderator who shall hold office for a term of one year and until a successor is elected and has qualified. In the event of inability of the Moderator to act, the Deputy shall have all the powers and duties of the Moderator.
- Clerk<sup>127</sup>. The Town Clerk or, in the event of the Town Clerk's absence, an Assistant Town Clerk, shall act as clerk of all RTM meetings.

<sup>121 2022</sup> recodification of current Article IV, §4.2.B (2006). Derived from Chapter III, §4 of the 1947 and 1956 Acts and 1975 Charter; and, and, Article IV, §4.2.B of the 1997 Charter.

<sup>122 2022</sup> recodification of current Article IV, §4.2.E (2006). Derived from Chapter II, §4 of the 1947 and 1956 Acts and 1975 Charter; and, and, Article IV, §4.2.E of the 1997 Charter.

<sup>123</sup> Note: There was an annual town meeting and budget meeting in Chapter III, §6 of the 1947 Act; further amended by §6 of the 1951 Special Act, as reaffirmed by Chapter II, §6 of the 1956 Act.

<sup>124 2022</sup> recodification of current Article IV, §4.4.A (2006). Derived from Chapter III, §6 of the 1947 and 1956 Acts and 1975 Charter; and, Article IV, §4.4.A of the 1997 Charter.

<sup>125 2022</sup> recodification and modification of current Article IV, §4.4.B (2006). Note: Chapter III, §3 and §4 of the 1947 and 1956 Acts and 1975 Charter refer to "presiding officer" and "moderator". The election of the moderator was introduced to the charter in 1975. Derived from Article IV, §4.4.B of the 1997 Charter.

<sup>&</sup>lt;sup>126</sup> 2022 recodification of current Article IV, §4.4.C (2006). Derived from Article IV, §4.4.C of the 1997 Charter.

<sup>&</sup>lt;sup>127</sup> 2022 recodification of current Article IV, §4.4.D (2006). Derived from Article IV, §4.4.D of the 1997 Charter.

- **E.** Moderator Pro-tempore and Clerk Pro-tempore<sup>128</sup>. In the absence of the Moderator and the Deputy Moderator, a Moderator Pro-Tempore may be elected, by a Majority Vote of the RTM. In the absence of the Town Clerk and an Assistant Town Clerk, a clerk pro-tempore for the meeting may be elected, by a Majority Vote of the RTM, byat the meeting.
- **F.** Rules and Committees of the RTM<sup>129</sup>. The RTM shall have the powerauthority to adopt standing rules for the conduct of RTM meetings and the power to appoint such committees as it shall determine.
  - **G.** Parliamentarian<sup>130</sup>. The Moderator may appoint a parliamentarian.

#### §3.4. Meetings of the RTM.

- **A.** Quorum<sup>131</sup>. A majority of the entire membership of the RTM shall constitute a quorum for doing business, provided that a smaller number may organize temporarily and may adjourn from time to time. No RTM meeting shall adjourn past the date of an election of RTM members.
  - B. Open Meetings<sup>132</sup>. All RTM meetings shall be public.
- **C.** Regular meetings<sup>133</sup>. Regular RTM meetings shall be held at least once a month on a regular meeting day decided upon by a Majority Vote of the RTM. However, if there is no business to be acted upon at a regular RTM meeting, the meeting may be dispensed with upon the direction of the Moderator to the Town Clerk who shall notify the RTM members and the public.
- **D.** Special and Emergency Meetings<sup>134</sup>. Special meetings may be held whenever the First Selectperson, Chair of the Board of Finance, or the Moderator shall deem them necessary. In addition, a special meeting shall be held within ten (10) days after the submission to the Town Clerk of a written petition for a meeting signed by one (1%)

11

<sup>&</sup>lt;sup>128</sup> 2022 recodification of current Article IV, §4.4.E (2006). Derived from Article IV, §4.4.E of the 1997 Charter. <sup>129</sup> 2022 recodification of current Article IV, §4.4.F (2006) (First clause). Derived from Article IV, §4.4.F of the 1997 Charter.

<sup>130</sup> NEW (2022)

<sup>&</sup>lt;sup>131</sup> 2022 recodification of current Article IV, §4.3.A (2006) (First sentence). Modification of Chapter III, §4 of the 1947 and 1956 Act, which established a quorum as one-third of the members and 1975 Charter (which established a majority standard); and, Article IV, §4.3.A of the 1997 Charter.

<sup>&</sup>lt;sup>132</sup> 2022 recodification of current Article IV, §4.3.A (2006) (Second sentence). Modification of Chapter III, §4 of the 1947 and 1956 Act, which established a quorum as one-third of the members and 1975 Charter (which established a majority standard); and, Article IV, §4.3.A of the 1997 Charter.

<sup>&</sup>lt;sup>133</sup> 2022 minor modification and recodification of current Article IV, §4.3.B (2006). Regular meeting provision established in Chapter III, §4 of the 1947 and 1956 Acts. Further amended by §5 of the 1951 Special Act ("no business" provision) and Chapter III, §4 of the 1975 Charter; and, Article IV, §4.3.B of the 1997 Charter.

<sup>&</sup>lt;sup>134</sup> 2022 recodification of current Article IV, §4.3.C (2006). Special meeting provision contained in Chapter III, §4 of the 1947 and 1956 Acts, and 1975 Charter; and, Article IV, §4.3.C of the 1997 Charter.

percent of the electors of the Town or upon written petition signed by ten (10) RTM members.

- **E. Public Hearings. Notice**<sup>135</sup>. At least one (1) public hearing shall be held by the RTM or any committee thereof before any Ordinance shall be passed. Meeting Notice of such public hearing shall be given, in accordance with the requirements of the General Statutes; or this Charter, if the standard is stricter.
- **F. Meeting Notice**<sup>136</sup>. The Town Clerk shall notify all RTM members of the time and place at which each regular or special RTM meeting is to be held, in accordance with the notice requirements of this Charter.

#### §3.5. Ordinances, Resolutions, Orders or Motions:

- A. Public Notice of Final Action<sup>137</sup>. The Town Clerk shall cause any action of the RTM adopting, amending, or repealing an Ordinance to be published in summary form, within one (1) week after the adjournment of the meeting at which such action was taken, in accordance with §1.4.C(17) of this Charter, above. The effective date of the adoption or repeal of the Ordinance shall be fourteen (14) Days after the adjournment of the meeting at which it was passed or such later date as may have been set by the RTM at such meeting, unless a petition for referendum concerning an Ordinance is filed as provided in §3.6 of this Article.
- **B.** Publication and Posting of Adopted Legislation<sup>138</sup>. All legislation, after final passage, shall be given a serial number by the Clerk of the RTM and be recorded by the Town Clerk and shall be properly indexed. Within five (5) Days after final passage, as provided in §3.5.A of this Charter, above, the Town Clerk shall transmit for publication on the Town website and notice that such legislation is on file in the Office of the Town Clerk and is available for examination in accordance with §1.4.C(17) of this Charter, above.

# §3.6 Petition for Overrule (Referendum)<sup>139</sup>.

# A. Petition and time for filing<sup>140.</sup>

(1) Effective date of certain RTM votes. Any vote of the RTM: (1) authorizing the expenditure for any specific purpose of \$500,000 or more<sup>141</sup>; or (2)

<sup>136</sup> 2022 recodification of current Article IV, §4.3.D (2006). Modification of Chapter III, §2 (third sentence) and §4 of the 1947 and 1956 Acts and 1975 Charter; and, Article IV, §4.3.D of the 1997 Charter.

<sup>135</sup> NEW (2022)

<sup>&</sup>lt;sup>137</sup> 2022 recodification of current Article IV, §4.5 (2006). Modification of Chapter III, §8 of the 1947 and 1956 Acts and 1975 Charter and Article IV, §4.5 of the 1997 Charter.

<sup>138</sup> NEW (2022)

<sup>&</sup>lt;sup>139</sup> Current Article XII.

<sup>&</sup>lt;sup>140</sup> 2022 recodification of current Article XIII, §13.1 (2006).

<sup>&</sup>lt;sup>141</sup> 2022 recodification of current Article XIII, §13.1.A(1) (2006). Derived from Chapter III, §9 of the 1947 and

for the issue of any bonds by the Town<sup>142</sup>; or (3) the adoption, amendment, or repeal of an Ordinance<sup>143</sup>; shall not be effective until the date for filing a referendum petition has passed. If within that time a petition for referendum is filed with the Town Clerk, the vote shall not be effective unless and until it has been approved by referendum.

- (2) Petition forms. Upon the request of any Elector, the Town Clerk shall promptly prepare petition forms, which shall be available to any Elector at the office of the Town Clerk, setting forth the questions sought to be presented on a referendum ballot.
- (3) Required number of signatures on petitions. To be effective, a petition for referendum must be signed by not less than five (5%) percent of the Electors of the Town according to the most recent voter enrollment list available at the time of the action or vote on which a referendum is sought, and must contain the names and addresses of the signatories.
- (4) Time and place of filing petitions. A petition requesting that a referendum be held must be filed at the office of the Town Clerk not later than the close of business on the fourteenth (14<sup>th</sup>) day after the adjournment of the meeting at which the vote was taken. If the fourteenth (14<sup>th</sup>) day is a day on which the Town Clerk's office is closed, the petition must be filed bythe close of business on the next day that the Town Clerk's office is open.
- (5) Special requirements for petitions on appropriations and bond issues. All petitions for referendum on any action of the RTM with respect to any bond issue or any appropriation in the amount required for a referendum shall set forth each item as to which a vote is desired, with the amount of the item as approved by the RTM, and the amount to which the petitioners desire it to be decreased or increased. However, no increase shall be proposed in excess of the amount approved for the item in question by the Board of Finance or by the RTM on appeal from the Board of Finance.

#### B. Manner of holding referendum<sup>144</sup>.

(1) Certification of Town Clerk. Upon the filing of a petition fulfilling the requirements of §3.6 of this Charter, the Town Clerk shall certify that fact promptly to the Board of Selectpersons.

<sup>1956</sup> Acts (\$25,000.00) and the 1975 Charter (\$50,000.00); and Article XIII, §13.1.A(i) of the 1997 Charter (\$150,000.00).

<sup>&</sup>lt;sup>142</sup> 2022 recodification of current Article XIII, §13.1.A(2) (2006). Derived from Chapter III, §9 of the 1947 and 1956 Acts and the 1975 Charter; and, Article XIII, §13.1.A(ii) of the 1997 Charter.

<sup>&</sup>lt;sup>143</sup> 2022 recodification of current Article XIII, §13.1.A(3) (2006). Derived from Chapter III, §10 of the 1947 and 1957 Act and 1975 Charter; further amended by §10 of the 1951 Act; and Article XIII, §13.1.A(iii) of the 1997 Charter.

<sup>&</sup>lt;sup>144</sup> 2022 recodification of current Article XIII, §13.2 (2006). Derived from Chapter III, §10 of the 1947 and 1956 Acts and 1975 Charter; further amended by §7 of the 1951 Act and; Article XIII, §13.2 of the 1997 Charter.

- (2) Date of referendum. The Board of Selectpersons shall call a special election for all Electors of the Town to be held not less than twenty-one (21) Days nor more than twenty-eight (28) Days after the date of certification by the Town Clerk for the sole purpose of voting approval or disapproval of the question or questions presented in the referendum petition.
- (3) Voting hours and method. For any referendum, the polls shall be opened at twelve o-clock (12:00 P.M.) noon and shall be closed at eight o'clock (8:00 PM) in the evening, but the hours for voting may be increased at the discretion of the Board of Selectpersons. Voting shall be by voting machine or printed ballot, at the discretion of the Board of Selectpersons.
- (4) Ballots for referendum on Ordinance. The ballot labels or ballots used in referenda concerning Ordinances shall state the matter to be voted on in substantially the following form: "Shall the following action of the Representative Town Meeting held on (date of the meeting) be approved?" followed by a statement of the action questioned in substantially the same language and form set forth in the records of the RTM. The voting machine or printed ballot shall provide means of voting "yes" or "no" on each question sopresented.
- (5) Ballots for referendum on appropriation and bond issue. Ballot labels or ballots used for referenda brought on appropriations and bond issues shall present separately each appropriation so referred in substantially one (1) of the following forms:
  - (a) "Shall a special appropriation, etc. be approved?"; or
  - (b) "Shall a special appropriation, etc. be (increased) (decreased) to the sum of \$\_\_\_\_?"; or
  - (c) "Shall the following items contained in the annual town budget be approved?; or
  - (d) "Shall the following items contained in the annual town budget be(increased) (decreased) to the sum of \$\_\_\_\_\_?"
- (6) Vote necessary to pass referenda<sup>145</sup>. In order to reverse or modify the action taken by the RTM, the vote in favor of reversing or modifying the action must both:
  - (a) Exceed twenty-five (25%) percent of the total number of Electors of the Town eligible tovote as of the close of business on the day before the election; and
    - (b) Constitute a majority of votes cast on the question.

Fairfield Charter – Public Hearing Version (25 May CRC Report to Board of Selectmen (6 June 2022)- 31

<sup>&</sup>lt;sup>145</sup> 2022 recodification of current Article XIII, §13.2.F. Derived from Chapter III, §11 of the 1947 and 1956 Acts; Chapter III, 11 of the 1975 Charter; and Article XIII, §13.2.F of the 1997 Charter.

#### §3.7. Appeals from the Board of Finance<sup>146</sup>.

The RTM shall have the power to hear and determine appeals from decisions of the Board of Finance as provided in §9.8 of this Charter[510].

§3.8. Powers Denied to the RTM<sup>147</sup>.

The RTM shall not diminish by Ordinance, vote or otherwise, the powers and duties vested in the First Selectperson by this Charter.

§3.9§3.8. Vacancies in the RTM<sup>148</sup>.

Any vacancy in the RTM shall be filled as set forth in §2.6.A of this Charter, above.

<sup>&</sup>lt;sup>146</sup> 2022 recodification of current Article IV, §4.6 (2006). Modification of Chapter III, §13 of the 1947 and 1956 Acts and Chapter III, §12 of the 1975; see also, Modification of Chapter XVII, §5 of the 1947 and 1956 Acts and 1975 Charter; and Article IV, §4.6 of the 1997 Charter.

<sup>147</sup> NEW (2022).

<sup>&</sup>lt;sup>148</sup> NEW (2022).

# ARTICLE IV - BOARD OF SELECTPERSONS AND THE FIRST SELECTPERSON

#### §4.1. Executive Authority.

- A. The Executive Branch<sup>149</sup>. The executive branch of the Town government shall consist of the First Selectperson and the Board of Selectpersons and the elected and appointed Boards, and Commissions, Elected Town Officials or Appointed Town Officers, and employees set forth in this Charter.
- **B.** The First Selectperson<sup>150</sup>. The First Selectperson shall be the chief executive officer of the Town as well as the town agent<sup>151</sup> and shall have the powers and duties vested in the office by this Charter and the General Statutes. The First Selectperson shall devote full time to the duties of the office<sup>152</sup>.
- **C.** Election and Qualifications. The First Selectperson and Board of Selectpersons shall be chosen by the Electors of the Town, as set forth in §2.3.C(1)(a) and 2.3.C(3)(b) and (3)(b) and §2.7 of this Charter, above 153.

#### §4.2. The Board of Selectpersons.

**A. Meetings**<sup>154</sup>. The Selectpersons shall hold their first meeting not later than the fourth (4<sup>th</sup>) Monday of November after their election. The First Selectperson shall be chair of the Board of Selectpersons and shall preside over all of its meetings<sup>155</sup>. The Selectpersons shall meet at least twice each month, unless there is no business to be conducted<sup>156</sup>.

B. General powers and duties<sup>157</sup>. The executive authority of the Town shall be

<sup>&</sup>lt;sup>149</sup> 2022 recodification of current Article V, §5.1 (2006). Derived from Article V, §5.1 of the 1997 Charter

<sup>&</sup>lt;sup>150</sup> 2022 recodification of current Article VI, §6.2.A (2006). Derived from Article VI, §6.2A of the 1997 Charter.

<sup>&</sup>lt;sup>151</sup> 2022 recodification of current Article VI, §6.2.A(2) which is derived from Chapter IV, §3(d) of the 1947 and 1956 Acts; Chapter IV, §4 of the 1975 Charter; and, Article VI, §6.2A(1) and (2) of the 1997 Charter.

<sup>&</sup>lt;sup>152</sup> 2022 recodification of current Article VI, §6.2.A(1) which is derived from Chapter IV, §3(d) of the 1947 and 1956 Acts; Chapter IV, §4 of the 1975 Charter; and, Article VI, §6.2A(1) and (2) of the 1997 Charter.

<sup>153</sup> NEW (2022)

<sup>&</sup>lt;sup>154</sup> 2022 recodification of current Article VI, §6.1.B (2006). Derived from Chapter IV, §2 of the 1947 and 1956 Acts and 1975 Charter; and, Article VI, §6.1.B of the 1997 Charter.

Derived from Chapter IV, §1 of the 1947 Act; further amended by §8 of the 1951 Act and affirmed by Chapter IV, §1 of the 1956 Act and the 1975 Charter; and, Article VI, §6.1.B of the 1997 Charter.

<sup>&</sup>lt;sup>156</sup> Derived from Chapter IV, §2 of the 1947 Act and 1956 Act.

<sup>&</sup>lt;sup>157</sup> 2022 recodification of current Article VI, §6.1.C (2006). Derived from Chapter IV, §3(a) and (c) of the 1947 Act and 1956 Act. Note: Subsection 3(b) of both Acts confers the authority of the Board of Sewer Commissioners upon the Board of Selectmen and the Department of Public Works as set forth under Special Act No. 222 (1929); see also, Chapter XXVI of the 1956 Special Act; Chapter IV, §3 and Chapter XXIV of the 1975 Charter; and, Article VI, §6.1.C of the 1997 Charter.

vested in the Board of Selectpersons, except to the extent such authority is expressly granted to the First Selectperson in this Charter. The Board of Selectpersons shall have the powers and duties vested in them by the General Statutes, except those expressly vested in the First Selectperson by this Charter or by Ordinance. In particular:

#### (1) Contracts.

- (\$100,000.00) Dollars, which are exempted from the competitive procurement and solicitation requirements of this Charter and any Ordinances or Regulations adopted hereunder 158 to which the Town (including the Board of Education) shall be a party shall be subject to approval of the majority vote of the Board of Selectpersons 159. The RTM is authorized to modify the threshold based upon the consumer price index or other like measure indexed for inflation or deflation 160. [S11]. This provision shall not apply to emergency transactions, permitted by the General Statutes, this Charter or Ordinance 161.
- **(b)** The Selectpersons shall have the power to delegate their authority to other Town <u>officersOfficers</u>, employees, and bodies where the contract would be for a duration of less than three (3) months and would involve an expenditure by or income to the Town of less than Twenty-five Thousand (\$25,000) Dollars and is in the line item under the jurisdiction of such Town Officer or employee <sup>162</sup>.
- **(c)** The approval requirement by the Board of Selectpersons, set forth in this sub-paragraph, above, shall not be construed to eliminate review by other persons or bodies where required by this Charter, by Ordinance, the General Statutes or, otherwise provided by Law<sup>163</sup>.
- (2) Oversight of Appointees: Subpoena Authority<sup>164</sup>. All Town officers, boards, commissions, Officials and employees of the Town appointed by the Board of Selectpersons shall be responsible to them for the faithful performance of their respective duties and shall render a report to the Selectpersons whenever requested to do so. The Selectpersons shall have the power to investigate any and all Town offices and Departments of the Town and for such purpose shall have the power to issue subpoenas.

<sup>&</sup>lt;sup>158</sup> NEW (2022).

<sup>159 2022</sup> modification and recodification of current Article VI, §6.1.C(1) (2006)(first sentence).

<sup>&</sup>lt;sup>160</sup> NEW (2022).

<sup>&</sup>lt;sup>161</sup> NEW (2022).

<sup>&</sup>lt;sup>162</sup> 2022 modification and recodification of current Article VI, §6.1.C(1) (2006)(second sentence).

<sup>&</sup>lt;sup>163</sup> 2022 recodification of current Article VI, §6.1.C(1) (2006)(fourth sentence).

<sup>&</sup>lt;sup>164</sup> 2022 recodification of current Article VI, §6.1.C(2) (2006)(First sentence). Modification of Chapter IV, §4 of the 1947, 1956 Acts and 1975 Charter; see also, Article VI, §6.1C.ii of the 1997 Charter; and, Article VI, §6.1.C(2) of the 2006 Charter. Please note the Special Act authority to issue subpoenas.

- (3) Budget Review and Recommendation<sup>165</sup>. The Board of Selectpersons shall review the budgets of all Elected Town Officials, Appointed Town Officers, Boards, Commissions, and Departments of the Town and make such recommendations in connection with such budgets to the Board of Finance as they deem necessary.
  - **(4) Appointment powers.** The Board of Selectpersons shall appoint:
  - (a) Required by Charter<sup>166</sup>: The Town officers, Boards and Commission, Officials and employees set forth in this Charter, any others required by the General Statutes or by Ordinance<sup>167</sup> to be appointed by the First Selectperson Board of Selectpersons and any other for which no other appointment provision is made in the General Statutes, this Charter or the Ordinances.
  - **(b)** Required by General Statutes or Ordinance<sup>168</sup>: The members of any other Board or Commission which is required by an Ordinance or the General Statutes.
  - (c) Advisory committees, task forces or other similar entities<sup>169</sup>. Such advisory committees, task forces or other similar entities, as they deem necessary or useful from time to time to study and advise on any Town matters which are the concern of the Board.
- (5) Membership on Boards, Commissions, and committees<sup>170</sup>. Except as otherwise expressly provided in §8.11 of this Charter, each member of the Board of Selectpersons shall be an ex officio member, without vote, on all Town Boards, Commissions, and committees.
  - (6) Reorganization of Departments<sup>171</sup>.
  - (a) Notwithstanding any provisions of this Charter, the Board of Selectpersons may propose to the RTM a Resolution which may alter the

<sup>&</sup>lt;sup>165</sup> 2022 recodification of current Article VI, §6.1.C(3) (2006).

<sup>&</sup>lt;sup>166</sup> 2022 recodification of current Article VI, §6.1.D (2006) (first sentence). Derived from Chapter IV, §6 of the 1975 Charter; and, Article VI, §6.1.D of the 1997 Charter.

<sup>&</sup>lt;sup>167</sup> 2022 recodification and modification of current Article VI, §6.1.D (2006) (second sentence). Derived from Chapter IV, §6 of the 1975 Charter; and, Article VI, §6.1.D of the 1997 Charter.

<sup>&</sup>lt;sup>168</sup> 2022 recodification of current Article VI, §6.1.D (2006) (second sentence). Derived from Chapter IV, §6 of the 1975 Charter; and, Article VI, §6.1.D of the 1997 Charter.

<sup>&</sup>lt;sup>169</sup> 2022 recodification of current Article VI, §6.1.E (2006). Modification of Chapter XXIV, §3 of the 1947 and 1956 Act; Chapter XXVI, 1 of the 1975 Charter; and, Article VI, §6.1.E of the 1997 Charter. .

<sup>&</sup>lt;sup>170</sup> 2022 recodification of current Article VI, §6.1.F (2006). Derived from Chapter IV, §6 of the 1975 Charter; and, Article VI, §6.1.F of the 1997 Charter.

<sup>&</sup>lt;sup>171</sup> 2022 recodification of current Article VI, §6.1.G (2006). Derived from Chapter II, 11 of the 1975 Charter and Article VI, §6.1.G of the 1997 Charter.

method of appointment to or organization of any Town office, Department, Board or Commission of the Town, including combining or separating the duties of such individuals or bodies. The Resolution will effect the change when enacted by the RTM in the form of an Ordinance.

- **(b)** The ability to make such changes by Ordinance shall not apply to the Police Department, the Fire Department, the Board of Library Trustees, any Elected Town Official, Elected Board or Commission, and any other office or body where such change by Ordinance is specifically prohibited by statute.
- **(c)** If the duties of two (2) or more bodies or Town offices are combined or separated by Ordinance, the existence, the method of appointment, the organization, powers, and duties of the prior Town offices or bodies shall cease as defined in this Charter and shall be as defined in the Ordinance.
- (d) The method of election to any Town office, Board, Commission, or agency of the Town may be changed by Ordinance, as set forth in §4.2.B(6)(a) of this Charter, above, if the method of election set forth in this Charter is or becomes improper, invalid, or ineffective because of a change in the law or an ambiguity in, or erroneous provision of, this Charter.

#### §4.3. Powers and Duties of the First Selectperson.

**A. Executive Powers**<sup>172</sup>. The day-to-day executive and management authority of the Town shall be vested in the First Selectperson. The First Selectperson shall have the powers and duties vested by Law.

#### **B. Duties.** The First Selectperson shall:

- (1) Direct and supervise<sup>173</sup> the administration of all Departments and officers<sup>174</sup> and shall be responsible for the administration of all the affairs of the Town in respect to such Departments<sup>175</sup>;
- (2) Be responsible for the faithful execution of all laws, provisions of the Charter and Ordinances governing the Town<sup>176</sup>;

<sup>&</sup>lt;sup>172</sup> 2022 recodification of current Article VI, §6.1.C (2006). Derived from Chapter IV, §3(a) and (c) of the 1947 Act and 1956 Act. Note: Subsection 3(b) of both Acts confers the authority of the Board of Sewer Commissioners upon the Board of Selectmen and the Department of Public Works as set forth under Special Act No. 222 (1929); see also, Chapter XXVI of the 1956 Special Act; Chapter IV, §3 and Chapter XXIV of the 1975 Charter; and, Article VI, §6.1.C of the 1997 Charter.

<sup>&</sup>lt;sup>173</sup> NEW (2022)

<sup>&</sup>lt;sup>174</sup> 2022 recodification of current Article VI, §6.2.A(3).

<sup>175</sup> NEW (2022).

<sup>&</sup>lt;sup>176</sup> 2022 recodification of current Article VI, §6.2.A(4).

- (3) Make periodic reports to the RTM and may convene Special and Emergency RTM Meetings, as set forth in §3.4.D, above, attend and participate in RTM meetings, but shall have no vote<sup>177</sup>;
- **(4)** Submit annually to the Board of Finance a proposed budget, including a capital budget (including the update of the five-year capital plan set forth, below) for the next fiscal year in accordance with the provisions of Article IX of this Charter<sup>178</sup>.
- (5) Prepare and maintain a long-term financial and capital planning that takes into account the next five years and shall further consider the financial effects of the Town's plan of conservation and development ("POCD[S12]") as required by the General Statutes. The annual plan and update of the five-year capital plan shall be submitted to the Board of Finance, which shall take the Plan into account during budget deliberations as set forth in Article IX of this Charter<sup>179</sup>.
- **(6)** Keep the RTM and Board of Finance fully informed on the financial condition of the Town by issuance of quarterly reports of income and expense as to budget items, including grants, receipts, expenditures and changes to said budgeted amounts<sup>180</sup>.
- (7) Select, appoint and hire Appointed Town Officers, including department heads, except as otherwise provided for in this Charter or by the General Statutes [181][S13];
- **(8)** Investigate the availability of state and federal funds and grants on behalf of the Town and advise any of the Town's Departments and Boards and Commissions with respect to obtaining said funds and grants, and periodically report to the RTM and Board of Finance<sup>182</sup>;
- **(9)** Act, or designate another, as the bargaining agent for the Town, with the exception of the Board of Education, in all labor and employment matters, including authority to retain the services of labor consultants and attorneys to assist in such matters<sup>183</sup>;
  - (10) After the election of any Town official of whom an oath is required by

<sup>&</sup>lt;sup>177</sup> 2022 modification and recognition of current Article IV, §4.2.A (2006) entitled "Composition" which permits the First Selectperson to participate in RTM meetings.

<sup>&</sup>lt;sup>178</sup> NEW (2022).

<sup>&</sup>lt;sup>179</sup> NEW (2022)

<sup>&</sup>lt;sup>180</sup> NEW (2022)

<sup>&</sup>lt;sup>181</sup> 2022 recognition of current Article IX requirements of the First Selectperson.

<sup>&</sup>lt;sup>182</sup> NEW (2022).

<sup>&</sup>lt;sup>183</sup> NEW (2022). Pursuant to the Municipal Employee Relations Act under the General Statutes.

law, cause the Town official to be sworn to the faithful discharge of the duties of office <sup>184</sup>:

- **(11)** Upon the request of any Selectperson, inform the Board of Selectpersons of the First Selectperson's actions 185;
- (12) Have the ability to delegate such authority as may be necessary to the Selectpersons or to administrative assistants whose appointment may be authorized by the RTM<sup>186</sup>; and
- (13) Have the ability to convene the members of any or all Departments, Boards and Commissions to review and coordinate activities and to plan operations of the Town government<sup>187</sup>.

The First Selectperson shall have such additional powers and shall perform such other duties as may from time to time be required by Ordinance, provided that the same are not inconsistent with this Charter or the provisions of the Connecticut General Statutes.

# C. Appointment powers<sup>188</sup>.

- (1) The First Selectperson shall appoint the <u>Appointed</u> Town <u>officersOfficers</u>, Boards and Commission <u>members</u>, and employees set forth in Articles VII and VIII of the Charter, any others required by the General Statutes or by Ordinance<sup>189</sup> to be appointed by the First Selectperson and any other for which no other appointment provision is made in this Charter<sup>190</sup>.
- **(2)** All <u>Appointed</u> Town <u>officers</u>, <u>BoardsOfficers</u>, <u>Board</u> and <u>CommissionsCommission members</u> and employees of the Town appointed solely by the First Selectperson shall be responsible to the First Selectperson for the faithful performance of their respective duties and shall report to the First Selectperson<sup>191</sup>.
- (3) Mediation and Resolution of Differences<sup>192</sup>. The First Selectperson shall be responsible for the mediation and resolution of differences between Boards,

<sup>&</sup>lt;sup>184</sup> 2022 recodification of current Article VI, §6.2.A(5).

<sup>&</sup>lt;sup>185</sup> 2022 recodification of current Article VI, §6.2.A(6).

<sup>&</sup>lt;sup>186</sup> 2022 recodification and modification of current Article VI, §6.2.A(7).

<sup>&</sup>lt;sup>187</sup> 2022 recodification of current Article VI, §6.2.A(8).

<sup>&</sup>lt;sup>188</sup> 2022 recodification of current Article VI, §6.2.B (2006). Derived from Article VI, §6.2B of the 1997 Charter. 2022 recodification and modification of current Article VI, §6.1.D (2006) (first sentence). Derived from Chapter IV, §6 of the 1975 Charter; and, Article VI, §6.1.D of the 1997 Charter.

<sup>&</sup>lt;sup>189</sup> 2022 recodification and modification of current Article VI, §6.1.D (2006) (second sentence). Derived from Chapter IV, §6 of the 1975 Charter; and, Article VI, §6.1.D of the 1997 Charter.

<sup>&</sup>lt;sup>190</sup> 2022 recodification of current Article VI, 6.2.B(1)(2006).

<sup>&</sup>lt;sup>191</sup> Derived from Chapter IV, §3(d) of the 1947 and 1956 Acts.

<sup>&</sup>lt;sup>192</sup> NEW (2022).

Commissions, Departments and other public bodies within the Town government relating to an interpretation and/or coordination of Town policies and procedures.

- **D. Designation of Acting First Selectperson**<sup>193</sup>. Immediately upon taking office, the First Selectperson shall designate, in writing, to the Town Clerk the member of the Board of Selectpersons authorized to act as First Selectperson during the unavailability or temporary disability of the First Selectperson and during the period from the date a vacancy occurs until a successor First Selectperson takes office under the provisions of §§2.6.C and D of this Charter, above. Such designation may be changed in writing from time to time.
- **E.** Staff<sup>194</sup>. The First Selectperson may entitled to appoint a chief of staff and administrative assistant, or equivalent positions for which the RTM shall appropriate sufficient funds. In addition, the First Selectperson and staff may employ such other staff necessary for the administration of official duties as provided for in the budget. All such assistants and staff shall be appointed by the First Selectperson and shall serve at the pleasure of the First Selectperson.

#### F. Town Administrator 195.

- (1) Appointment and Duties. The First Selectperson shall appoint, and the budget shall provide for, a Town Administrator, who shall be the principal advisor to the First Selectperson for the management of Town Departments, and shall perform such duties as may be assigned by the First Selectperson [S15].
- (2) Term. The administrator shall be an unclassified employee of the town and serve at the pleasure of the First Selectperson.
- **(3) Experience.** Said administrator shall be appointed on the basis of substantial executive and administrative experience, education, competencies, and credentials, in management and administration, in accordance with the best practices recommended for local government management by reputable national organizations with subject matter expertise in the management of local government and public administration<sup>196</sup>.
- (4) Qualifications. Said qualifications shall be prepared by the Director of Human Resources, who shall also prequalify candidates for the position prior to

<sup>&</sup>lt;sup>193</sup> 2022 recodification of current Article VI, §6.2.C (2006). Derived from Article VI, §6.2.C of the 1997 Charter. <sup>194</sup> NEW (2022).

<sup>&</sup>lt;sup>195</sup> NEW (2022).

<sup>&</sup>lt;sup>196</sup> **Comment of the 2022 Charter Revision Commission.** The following organizations meet the criteria set forth in the Charter at the time of deliberations and approval: such as the Government Finance Officers Association, the International City/County Management Association, National League of Cities, National Academy of Public Administrators, the IBM Center for the Business of Government, International Public Management Association for Human resources and other equivalent or successor organizations

interview by the First Selectperson.

**G. Purchasing authority**<sup>197</sup>**.** The First Selectperson and the Purchasing Agent, acting in conjunction, shall be the general purchasing authority of the Town. All supplies, materials, equipment, othercommodities, Contracts for public works or services, other than professional services, required by any department, office, agency, board, authority, or commission of the Town, including the Board of Education, shall be purchased by the purchasing authority on a requisition, in such form as the Selectpersons may prescribe, signed by the head of the Department, office, agency, or chair of the Board or Commission or other like entities. No purchase order shall be issued without the signature of the Purchasing Agent or, in the Purchasing Agent's absence, of the First Selectperson.

#### §4.4. Compensation of executive branch members<sup>198</sup>.

The members of all Boards and Commissions except the Board of Selectperson shall serve without compensation unless the RTM shall otherwise direct. Except as provided in this Charter or otherwise by Law, the compensation of all Town officersOfficials shall be fixed by the Board of Selectpersons subject to the adoption of the Town Budget as provided in Article IX of this Charter, below 199.

# §4.5. Regulations<sup>200</sup>.

Any Town Officer, Board, or CommissionOfficial empowered to enact Regulations under the provisions of the General Statutes or of this Charter shall hold at least one (1) public hearing before the enactment of such Regulations. Except as otherwise provided by statute, the time and place of such hearing together with a copy of the proposed Regulations shall be published at least once not more than ten (10) nor less than five (5) Days before the date set for such hearing.

Except as otherwise provided by statute, any such Regulation shall be superseded by an Ordinance adopted by the RTM affecting the same subject matter.

# §4.6. Absence, Disability, Vacancy in the Office of First Selectperson.

Any vacancy in the Office of the First Selectperson shall be addressed as set forth in §2.6.D of this Charter, above.

<sup>&</sup>lt;sup>197</sup> Recodification of current Article XII, §12.8 (2006). Derived from Chapter XXI of the 1947 Act; and further amended by §20 of the 1951 Act and affirmed by Chapter XXI of the 1956 Act and Chapter XVIII of the 1975 Charter. Derived from Article XII, §12.8 of the 1997 Charter.

<sup>&</sup>lt;sup>198</sup> 2022 Recodification of current Article V, §5.2 (2006). Derived from Modification of Chapter II, §8 and Chapter IV, §5 of the 1947 Act and 1956 Act; Chapter II, §7 of the 1975 Charter and Article V, §5.2 of the 1997 Charter.

 <sup>199 2022</sup> repeal of current Article V, §5.3 (2006)("Official Bonds"). Derived from Modification of Chapter II, §9 of the 1947 Act and 1956 Act; Chapter II, §9 of the 1975 Charter and Article V, §5.3 of the 1997 Charter.
 200 2022 Recodification of current Article V, §5.4 (2006). Chapter XXVI, §8 of the 1975 Charter; and, Article V, §5.4 of the 1997 Charter.



# ARTICLE V – OTHER ELECTED OFFICERS, BOARDS AND COMMISSIONS

#### § 5.1. Town Clerk<sup>201</sup>.

- **A. Establishment and election**<sup>202</sup>**.** There shall be a Town Clerk elected at the times and for the term set forth in §2.3.A(3) and §2.3.C(1)(d) of this Charter.
- **B.** Powers and duties<sup>203</sup>. The Town Clerk shall have the powers and duties prescribed by this Charter, by Ordinance, and by the General Statutes. In particular, the Town Clerk shall:
  - (1) Devote full time to the duties of the office;
  - (2) Collect the fees or compensation provided by the General Statutes to be paid to the Town Clerk;
  - (3) Deposit all money required to be collected by the Town Clerk with the Chief Fiscal Officer, with whom the Town Clerk shall file a full statement of receipts at the time of each deposit; and
  - **(4)** Provide, as may be requested, a receipt for all money received to the person from whom it was received.
- **C. Staffing**<sup>204</sup>. All expenses of the Town Clerk's office, including necessary clerical assistance and Assistant Town Clerks, shall be paid by the Town within the limit of the appropriation therefor.
- **D.** Assistant Town Clerks<sup>205</sup>. The Town Clerk may appoint Assistant Town Clerks who shall be under the supervision of the Town Clerk and shall perform such duties as the Town Clerk specifies.
- **E.** Compensation<sup>206</sup>. In lieu of all fees and other compensation, the Town Clerk shall receive a salary fixed by the Board of Finance.
- **F.** Ordinances and votes<sup>207</sup>. The Town Clerk shall publish notice of action concerning Ordinances in accordance with this Charter. All Ordinances and RTM votes

<sup>&</sup>lt;sup>201</sup> Derived from Chapter VII of the 1947 and 1956 Acts and 1975 Charter.

<sup>&</sup>lt;sup>202</sup> 2022 recodification of current Article VII, §7.1.A (2006). Derived from Article VII, §7.1.A of the 1997 Charter.

<sup>&</sup>lt;sup>203</sup> 2022 recodification of current Article VII, §7.1.B (2006). Derived from Article VII, §7.1.B of the 1997 Charter.

<sup>&</sup>lt;sup>204</sup> 2022 recodification of current Article VII, §7.1.C (2006). Derived from Article VII, §7.1.C of the 1997 Charter.

<sup>&</sup>lt;sup>205</sup> 2022 recodification of current Article VII, §7.1.D (2006). Derived from Article VII, §7.1.D of the 1997 Charter.

<sup>&</sup>lt;sup>206</sup> 2022 recodification of current Article VII, §7.1.E (2006). Derived from Article VII, §7.1.E of the 1997 Charter.

<sup>&</sup>lt;sup>207</sup> 2022 recodification of current Article VII, §7.1.F (2006). Derived from Article VII, §7.1.F of the 1997 Charter.

shall be recorded by the Town Clerk in records kept for the purpose.

- **G.** Reports to Assessor and Tax Collector<sup>208</sup>. The Town Clerk shall promptly file with the Assessor and Tax Collector a complete abstract of all deeds and conveyances of land or of personal property, or of certificates of intention to transfer personal property, placed in the custody of the Town Clerk for record.
- H. Vacancy<sup>209</sup>. As A Vacancy in the Office of the Town Clerk shall be filled as set forth in §2.6.A of this Charter, above.

#### §5.2. Justices of the Peace.

- **A.** Establishment and election<sup>210</sup>. There shall be forty-five (45) Justices of the Peace<sup>211</sup> elected or nominated in the manner and for the terms prescribed in §2.3.A and BC of this Charter, above.
- **B.** Powers and duties<sup>212</sup>. Justices of the Peace shall have the powers and duties prescribed by the General Statutes for their respective offices.
- C. Vacancies<sup>213</sup>. Vacancies shall be filled as set forth in §2.<del>5</del>E 6.B. of this Charter, above.

#### §5.3. Elected Boards and Commissions: In General.

**A.** Establishment and election<sup>214</sup>. There shall be a Board of Education, Board of Finance, Board of Assessment Appeals, Town Plan and Zoning Commission, and Zoning Board of Appeals. The members of each Board and Commission shall be elected at the times and for the terms set forth in §2.3.A and BC of this Charter, above.

#### B. Meetings

<sup>208</sup> 2022 recodification of current Article VII, §7.1.G (2006). Derived from Article VII, §7.1.G of the 1997 Charter.

<sup>&</sup>lt;sup>209</sup> 2022 recodification and modification of current Article VII, §7.1.H (2006). Derived from Article VII, §7.1.H of the 1997 Charter.

<sup>&</sup>lt;sup>210</sup> 2022 modification and recodification of current Article VII, §7.2.A (2006). Derived from Article VII, §7.2.A of the 1997 Charter.

<sup>&</sup>lt;sup>211</sup> Chapter II, §1 of the 1947 Act, included 14 justices of the peace; the number was raised to 18 in Chapter II, §1 of the 1956 Act; and then to 30 in Chapter II, §1 of the 1975 Charter; and them, 45 in Article VII, §7.2.A of the 1997 Charter.

<sup>&</sup>lt;sup>212</sup> 2022 modification and recodification of current Article VII, §7.2.B (2006). Derived from Article VII, §7.2.B of the 1997 Charter.

<sup>&</sup>lt;sup>213</sup> 2022 recodification and modification of current Article VII, §7.2.C (2006). Derived from Article VII, §7.2.C of the 1997 Charter.

<sup>&</sup>lt;sup>214</sup> 2022 recodification of current Article VIII, §8.1.A (2006). Derived from Article VIII, §8.1.A of the 1997 Charter.

- (1) All elected Boards and Commissions, except the Board of Assessment Appeals, shall hold at least ten (10) Regular Meetings a year and shall give annual notice of such meetings as required by the General Statutes. Officers of each Board and Commission, except the RTM, shall be elected annually at an organization meeting so noticed with the Town Clerk held in the month of December<sup>215</sup>..
- (2) All elected Boards and Commissions shall comply with the provisions of §1.455 and §1.466 of this Charter<sup>216</sup>.
- **C. Vacancies.** A vacancy in the membership of any elected Board or Commission shall be filled in the manner prescribed in §2.6 of this Charter, above.

# §5.4. Board of Education<sup>217</sup>.

- **A.** Composition<sup>218</sup>. The Board of Education shall consist of nine (9) members, for staggered terms, as set forth in §2.3.A(5) and §2.3.C(1)(f) and C(2)(b) of this Charter, above, no more than six (6) of whom shall be registered with the same political party.
- **B.** Powers and duties<sup>219</sup>. The Board of Education shall have all the powers and duties conferred on boards of education generally by Chapter 170 of the General Statutes.

#### §5.5. Board of Finance.

- **A.** Composition<sup>220</sup>. The Board of Finance shall consist of nine (9) voting members, for staggered terms, as set forth in §2.3.A(4), §2.3.C(1)(e) and §2.3.C(2)(a) of this Charter, above, no more than six (6) of whom shall be registered with the same political party, and the Board of Selectpersons and the Fiscal Officer, ex officio, without vote.
- B. Powers and duties<sup>221</sup>. The Board of Finance shall appoint the outside auditors and shall have all of the powers and duties conferred by this Charter, by

<sup>217</sup> Derived from Chapter XIX of the 1947 and 1956 Acts; and Chapter XX of the 1975 Charter.

<sup>&</sup>lt;sup>215</sup> 2022 recodification of current Article VIII, §8.1.B (2006). Derived from Article VIII, §8.1.B of the 1997 Charter.

<sup>&</sup>lt;sup>216</sup> NEW (2022)

<sup>&</sup>lt;sup>218</sup> 2022 recodification and minor modification of current Article VIII, §8.2.A (2006). Derived from Article VIII, §8.2.A of the 1997 Charter.

<sup>&</sup>lt;sup>219</sup> 2022 recodification of current Article VIII, §8.2.B (2006). Derived from Article VIII, §8.2.B of the 1997 Charter

<sup>&</sup>lt;sup>220</sup> 2022 recodification and minor modification of current Article VIII, §8.3.A (2006). Derived from Chapter XVII, §1 of the 1947 and 1956 Acts and the 1975 Charter; and, Article VIII, §8.3.A of the 1997 Charter. See also, Chapter II, §4(c) of the 1947 and 1956 Acts.

<sup>&</sup>lt;sup>221</sup> 2022 recodification of current Article VIII, §8.3.B (2006). Derived from Article VIII, §8.3.B of the 1997 Charter.

Ordinance, and upon boards of finance generally by Chapter 106 of the General Statutes.

- C. Clerk of the Board of Finance<sup>222</sup>. The Board of Finance shall appoint a clerk and fix the clerk's salary. The clerk shall hold office at the pleasure of the Board of Finance. The clerk may be part time. The clerk shall:
  - (1) Keep minutes of Board meetings and be the custodian of its records, papers, and data relating to the conduct of its business;
  - (2) Be a certified or a licensed public accountant or otherwise have credentials, licenses and or certification and experience in the financial field; and
  - (3) Have the right to call upon all Town Departments, Boards and Commissions and officers for such information as the clerk may reasonably require in connection with the duties of the clerk and responsibilities of the Board of Finance<sup>223</sup>..
- **D. Approval of budgets**<sup>224</sup>. The Board of Finance shall approve the Town budget in the manner set forth in Article IX of this Charter.
- **E. Bidding, requisition, and payment procedures**<sup>225</sup>. The Board of Finance shall establish and may amend from time to time procedures and guidelines for bidding on purchases and Contracts by the Town as well as procedures for departmental requisition and for payments.

#### §5.6. Board of Assessment Appeals.

- **A.** Composition<sup>226</sup>. The Board of Assessment Appeals shall consist of five (5) members, for staggered terms as set forth in §2.3.A(10), §2.3.C(1)(k) and §2.3.C(2)(g) of this Charter, above, and to be elected in accordance with Chapter 146 of the General Statutes (C.G.S. § 9-164 et seq.).
  - B. Powers and duties<sup>227</sup>. The Board of Assessment Appeals shall have all the

<sup>&</sup>lt;sup>222</sup> 2022 recodification of current Article VIII, §8.3.C (2006). Modification of Chapter XVII, §3 of the 1947 and 1956 Acts and the 1975 Charter; and Article VIII, §8.3.C of the 1997 Charter.

<sup>&</sup>lt;sup>223</sup> 2022 recodification and modification of current Article VIII, §8.3.D (2006). Modification of Chapter XXII of the 1947 and 1956 Acts and Chapter XXIII of the 1975 Charter, which ratified Special Act No. 511 (1935); Special Act No. 270 (1939); and, Special Act No. 367 (1941); and, Article VIII, §8.3.D of the 1997 Charter.

<sup>&</sup>lt;sup>224</sup> 2022 recodification of current Article VIII, §8.3.E (2006). Modification of Chapter XVII, §4 of the 1947 and 1956 Acts and the 1975 Charter; and, Derived from Article VIII, §8.3.E of the 1997 Charter.

<sup>&</sup>lt;sup>225</sup> Recodification of current Article XII, §12.9 (2006). Derived from Chapter XVIII, §4 and §5 of the 1975 Charter and Article XII, §12.9 of the 1997 Charter.

<sup>&</sup>lt;sup>226</sup> 2022 recodification and minor modification of current Article VIII, §8.4.A (2006). Derived from Article VIII, §8.4.A of the 1997 Charter.

<sup>&</sup>lt;sup>227</sup> 2022 Recodification of current Article VIII, §8.4.B (2006). Derived from Article VIII, §8.4.B of the 1997 Charter.

powers and duties conferred on boards of assessment appeals generally by §§12-110 to 12-117 of Chapter 203 of the General Statutes.

#### §5.7. Town Plan and Zoning Commission.

- **A.** Composition<sup>228</sup>. The Town Plan and Zoning Commission shall consist of seven (7) voting members, for staggered terms as set forth in §2.3.A(6) and §2.3.C(1)(g) and (h); and, §2.3.C(2)(c) and (d) of this Charter, above, no more than five (5) of whom shall be registered with the same political party. There shall be three (3) alternate members of the Town Plan and Zoning Commission, as set forth in §2.3.A(7) of this Charter, above, no more than two (2) of whom shall be registered with the same political party.
- **B.** Powers and duties<sup>229</sup>. The Town Plan and Zoning Commission shall have all the powers and duties conferred by this Charter, by Ordinance, and on zoning commissions and planning commissions generally by Chapter 124 and Chapter 126 of the General Statutes (C.G.S. §8-1 et seq. and §8-18 et seq.). In particular, the Town Plan and Zoning Commission shall:
  - (1) Prepare, adopt, and amend a master plan for the development<sub>[S16]</sub> of the Town, or amend any master plan previously adopted<sup>230</sup>;
    - (2) Have control over the subdivision of land<sup>231</sup>;
  - (3) Make studies and recommendations on matters affecting health, recreation, traffic, and other needs of the Town dependent on, or related to, the master plan;
  - **(4)** Consider and report upon the design, location, and relation to the master plan of all new public ways, buildings, bridges, and other public places and structures<sup>232</sup>:

<sup>&</sup>lt;sup>228</sup> 2022 recodification and minor modification of current Article VIII, §8.5.A (2006). Derived from Chapter XI, §1 of the 1947 Act, which established five members. The Board was expanded to seven members under the 1956 Act. The October election date was retained in §12 of the 1951 Act and reaffirmed by Chapter XI, §1 of the 1956 Act and the 1975 Charter; and, Article VIII, §8.5.A of the 1997 Charter.

<sup>&</sup>lt;sup>229</sup> 2022 recodification of current Article VIII, §8.5.B (2006). Modification of Chapter XI, §2 of the 1947 Act. There was also an appeal to the court of common pleas in Chapter XI, §8 of the 1947 Act; further amended by §13 of the 1951 Act. The appeal to the court of common pleas was repeal by §14 of the 1951 Act. Reaffirmed by Chapter XI, §2 of the 1956 Act and the 1975 Charter; and, Article VIII, §8.5.B of the 1997 Charter.

<sup>&</sup>lt;sup>230</sup> Replacement of Chapter XI, §3 of the 1947 Act and 1956 Act.

<sup>&</sup>lt;sup>231</sup> Replacement of Chapter XI, §5 of the 1947 Act. At the time a definition of "subdivision" was included in Chapter XI, §7 of the 1947 Act, which was repealed by §14 of the 1951 Act, along with §5; as reaffirmed by Chapter XI of the 1956 Act.

<sup>&</sup>lt;sup>232</sup> Replacement of Chapter XI, §6 of the 1947 Act; Chapter XI, §5 of the 1956 Act; and Chapter XI, §4 of the 1975 Charter.

- (5) Consider and report upon the layout of new developments in the Town<sup>233</sup>:
- **(6)** Make detailed plans for the improvement, reconditioning, or development of areas which in its judgment contain special problems or show a trend toward lower land values; and
- (7) Make such other studies, recommendations, and inclusions in the master plan as will in its judgment be beneficial to the Town<sup>234</sup>.
- **C.** Appointment of Planning Director<sup>235</sup>. The Town Plan and Zoning Commission shall appoint a Planning Director who has been approved by the First Selectperson.
- **D.** Other employees and consultants<sup>236</sup>. The Town Plan and Zoning Commission may engage such employees and consultants as it requires to carry out its duties, including a zoning enforcement officer and assistants who, subject to the general supervision of the Planning Director, shall enforce all Laws, Ordinances, and Regulations relating to zoning and planning, and shall have such other duties as the Town Plan and Zoning Commission or the Planning Director may prescribe.

#### §5.8. Zoning Board of Appeals<sup>237</sup>.

- **A.** Composition<sup>238</sup>. The Zoning Board of Appeals shall consist of five (5) regular members, for staggered terms as set forth in §2.3.A(9), §2.3.C(1)(i) and (j); and, §2.3.C(2)(e) of this Charter, above, no more than four (4) of whom shall be registered with the same political party<sup>239</sup>, and three (3) alternates, as set forth in §2.3.C(2)(f) of this Charter, above, no more than two (2) of whom shall be registered with the same political party.
- **B.** Powers and duties<sup>240</sup>. The Zoning Board of Appeals shall have the powers and duties conferred on zoning boards of appeals generally by §§ 8-5 to 8-7e of Chapter

<sup>&</sup>lt;sup>233</sup> Replacement of Chapter XI, §4 of the 1947 and 1956 Acts and Chapter XI, §5 of the 1975 Charter.

<sup>&</sup>lt;sup>234</sup> Replacement of Chapter XI, §3 of the 1947 Act. §2 and 3 of the 1947 Act were repealed by §15 and 16 of the 1951 Act.

<sup>&</sup>lt;sup>235</sup> 2022 recodification of current Article VIII, §8.5.C (2006). Derived from Article VIII, §8.5.C of the 1997 Charter.

<sup>&</sup>lt;sup>236</sup> 2022 recodification of current Article VIII, §8.5.D (2006). Derived from Article VIII, §8.5.D of the 1997 Charter.

<sup>&</sup>lt;sup>237</sup> Modification of Chapter XII of the 1947 and 1956 Acts and 1975 Charter.

<sup>&</sup>lt;sup>238</sup> 2022 recodification and minor modification of current Article VIII, §8.6.A (2006). Derived from Article VIII, §8.6.A of the 1997 Charter. The

<sup>&</sup>lt;sup>239</sup> The current "minority party" standard was not included in Chapter XII §2 of the 1947 and 1956 Acts: "Not more than three members of said board shall be members of the same political party." The standard was increased to "four" in Article IX, §8.6.A of the 1997 Charter and has continued thereafter.

<sup>240 2022</sup> Recodification of current Article VIII, §8.6.B (2006). Derived from Article VIII, §8.6.B of the 1997 Charter.

# CHARTER OF THE TOWN OF FAIRFIELD 124 of the General Statutes.

# ARTICLE VI - APPOINTED OFFICERS AND BOARDS AND COMMISSIONS IN GENERAL

#### §6.1. Eligibility for appointed Town officeOffice<sup>241</sup>.

- A. Elector requirementrequirements for Certain Appointed Town Officers and Members of Boards and Commissions; Exception<sup>242</sup>. No person not at the time an Elector of the Town shall be eligible for appointment to any appointed Board or Commission or to the office of Town Treasurer, Constable or Town Attorney[S17]. Persons serving as Chief of Police or Fire Chief must be or become ana resident of the State of Connecticut, in accordance with the requirements of the respective appointing authorities.
- **B.** Effect of ceasing to be an Elector<sup>243</sup>. If any Appointed Town Officer required to be an Elector of the Town or any member of an appointed Board or Commission ceases to be an Elector of the Town, the office shall then become vacant.
- C. Single office requirement for appointed office<sup>244</sup>. Except as otherwise provided in §§8.5, 8.7, 8.8, and 8.11 of this Charter, no person shall be eligible to hold any appointed Town Office, including membership on any permanent appointed members of any Board or Commission, who is at the same time an elected RTM member, Town Officer, an elected state official, or a member of an elected Board or Commission or permanent appointed Board or Commission. For purposes of this paragraph, the term "Town officer" does not include Justices of the Peace or Constables.
- D. Eligibility for reappointment<sup>245</sup>. No person shall be appointed to more than two successive full terms on the same appointed Board or Commission, but such person shall be eligible for reappointment after an interval of not less than one (1) year except as otherwise provided in Sections §§8.14 and 8.15 of this Charter, below, with respect to the Ethics Commission and Golf Commission, respectively, or by Ordinance.

### §6.2. Authority and Qualification of Appointed Town Officers<sup>246</sup>.

A. Authority of Appointed Town Officers, including department heads<sup>247</sup>. Once appointed as described in this Charter, Appointed Town Officers, including

<sup>&</sup>lt;sup>241</sup> Derived from Chapter II, §3 of the 1947 and 1956 Acts and 1975 Charter.

<sup>&</sup>lt;sup>242</sup> 2022 modification and recodification of current Article III, §3.1.A (2006). Derived from Article III, §3.1.A of the 1997 Charter.

<sup>&</sup>lt;sup>243</sup> 2022 recodification of current Article III, §3.1.B (2006). Derived from Article III, §3.1.B of the 1997 Charter.

<sup>244 2022</sup> recodification of current Article III, §3.1.C (2006). Derived from Article III, §3.1.C of the 1997 Charter.

<sup>&</sup>lt;sup>245</sup> 2022 recodification of current Article III, §3.1.D (2006). Modification of Article III, §3.1.D of the 1997 Charter.

<sup>&</sup>lt;sup>246</sup> NEW (2022).

<sup>&</sup>lt;sup>247</sup> NEW (2022).

department heads shall, unless otherwise set forth in this Charter or Law, appoint, hire, discipline and remove all deputies, assistants and other employees of the Department and subject to (1) the authority of the First Selectperson or other appointing or supervisory authority; (2) any specific provisions of this Charter or Law pertaining to the authority of Appointed Town Officer, including department heads and the relationship with a governing Board or Commission; and, (3) any applicable collective bargaining agreements.

B. Qualifications of Appointed Town Officers, including department heads<sup>248</sup>. The job qualifications of all Appointed Town Officers, including department heads, in addition to those enumerated in this Charter, by the General Statutes or Special Act shall be established by Ordinance following the recommendation of<sub>[S18]</sub> the human resources or personnel director or as may otherwise be designated by Ordinance. Said job qualifications shall be prepared in accordance with nationally accepted professional standards and best practices in the applicable field and shall be reviewed and updated, if necessary, every four (4) years and whenever a Vacancy occurs in the position.

#### §6.3. Minority representation on Boards and Commissions<sup>249</sup>.

Unless other expressly set forth in this Charter, no more than a bare majority of the members of a Board or Commission shall be members of the same political party. This shall not apply to persons required to serve on a particular Board or Commission by virtue of holding another position or office in the Town.

#### §6.4. Terms of office<sup>250</sup>.

A. General Rule: Appointed Town Officers and Board and Commission Members. Unless otherwise provided by the General Statutes, this Charter or by Ordinance the terms of office of all Appointed Town Officers and appointed Town officers and members of permanent appointed Boards and Commissions shall commence on the fourth (4th) Monday in November, and shall continue for the termterms set forth in §2.3Articles VII and VIII of this Charter, above. Notwithstanding the foregoing, in the event that there is a vacancy and the Appointing Authority (Town Official, which may be an official Board, Commission withpossesses the authority to make an appointment) ("Appointing Authority") does not appoint a new Appointed Town officerOfficer or Board or Commission member by the fourth (4th) Monday in November after the municipal general election of the Appointing Authority, the former appointeeincumbent shall serve as a continuing Appointed Town officerOfficer or Board or Commission member until either the Appointing Authority fills the vacancy or until the sixty-fifth (65th) day after the fourth (4th) Monday in November, whichever occurs first.

B. Exception: Golf Commission and Ethics Commission. The members of

<sup>&</sup>lt;sup>248</sup> NEW (2022).

<sup>&</sup>lt;sup>249</sup> 2022 recodification of current Article III, §3.2 (2006). Derived from Article III, §3.2 of the 1997 Charter. <sup>250</sup> 2022 modification and recodification of current Article III, §3.3 (2006). Derived from Article III, §3.3 of the

the Ethics Commission and the Golf Commission shall commence their terms on April 1 in the year of their appointment for the duration of their term and shall only remain on said Commission if reappointed[S19].

#### §6.5. Resigning from appointed office<sup>251</sup>.

Any appointed Appointed Town officer or member of an appointed Board or Commission member may resign by submitting a written notice of resignation to the Town Clerk. The resignation shall become effective on the date specified in the notice of resignation or, if no date is specified, on the date the notice of resignation is submitted.

#### §6.6. Vacancies in appointed office<sup>252</sup>.

Unless otherwise set forth in the General Statutes or otherwise in this Charter, Vacancies in appointed Town offices and membership on all Appointed Town Offices and Boards and Commissions shall be filled by the Appointing Authority. Persons so appointed shall serve for the remainder of the term of the position vacated.

#### §6.7. Removal from appointed office for cause.

- **A. Board or Commission**<sup>253</sup>. Any Appointed Town Officer or member of any appointed Board or Commission may, except as otherwise provided in the General Statutes or this Charter, be removed for cause by the appointing Town officer or Board or Commission; Appointing Authority (including instances where Boards and Commissions have appointment authority), unless other set forth by Law provided, notice shall first be given in writing of the specific grounds for removal and the individual shall be given an opportunity to be heard in defense, alone or with counsel of the individual's choice, at a hearing before the appointing Town officer or Board or Commission Appointing Authority, held not more than ten (10) and not less than five (5) Days after transmittal of such notice in accordance with §1.4.C(15), above.). The hearing shall be held in accordance with the Freedom of Information Act (Chapter 14 of the General Statutes, C.G.S. §1-200 et seq.). Appeals may be taken where provided by statute.
- **B. Department Head or Director**<sup>254</sup>**.** Unless this Charter provides that a particular department head or director may be removed without cause, the Board of Selectpersons by unanimous vote of all of its members shall have the power to remove for cause any Department Head, or any director <u>or employee</u> appointed by a Board or Commission, provided notice and opportunity for a hearing before the Board of Selectpersons following the same protocols as set forth in this §6.7.A of this Charter [520]-

<sup>&</sup>lt;sup>251</sup> 2022 recodification of current Article III, §3.4 (2006). Derived from Article III, §3.4 of the 1997 Charter.

<sup>&</sup>lt;sup>252</sup> 2022 recodification of current Article III, §3.5 (2006). Derived from Article III, §3.5 of the 1997 Charter.

<sup>&</sup>lt;sup>253</sup> 2022 recodification of current Article III, §3.6.A (2006). Derived from Modification of Chapter II, §6 of the 1947 and 1956 Acts; Chapter II, §12 of the 1975 Charter; and, Article III, §3.6.A of the 1997 Charter.

<sup>&</sup>lt;sup>254</sup> 2022 recodification of current Article III, §3.6.B (2006). Derived from Article III, §3.6.B of the 1997 Charter.

above, and such hearing is held in accordance with the Freedom of Information Act.

#### §6.8. Required Cooperation<sup>255</sup>.

Each Town Officer, Town Official, RTM members and employee of any Department of the Town shall assist the Boards and Commissions and the pertinent Departments in carrying out the provisions of this section. Furthermore, in the event a Board or Commission is attached to a Department, the Department Head shall cooperate with said Board or Commission in formulating and carrying out the operational, management and personnel policies of the Department.

#### §6.9. Compliance<sup>256</sup>.

All appointed Boards and Commissions shall comply with the provisions of §§§1.15 and §5 through 1.167 of this Charter[521].

#### §6.10. Creation of Boards and Commissions<sup>257</sup>.

In addition to those appointed Boards and Commissions established by this Charter, the RTM may, on its own initiative; or at the recommendation of the First Selectperson, by Ordinance, establish any additional Boards and Commissions of the Town.

256 NEW (2022)

<sup>&</sup>lt;sup>255</sup> NEW (2022)

<sup>257</sup> NEW (2022)

#### **ARTICLE VII - APPOINTED TOWN OFFICERS**

§7.1. Appointed Town Officers<sup>258</sup>.

There shall be the following appointed Town Officers as necessary:

A. Officers Appointed by the First Selectperson<sup>259</sup>. The First Selectperson shall appoint the Appointed Town Officers listed in §§7.2 through 7.12 of this Charter, below and §§4.3.E and F of this Charter, above.

Town Attorney Assistant Town Attorney

Fiscal Officer Controller

Town Treasurer Director of Community and Director of Public Works Economic Development

Director of Human and Social Services Building Official Director of Parks and Recreation Purchasing Agent

Town Administrator Director of Human Resources

Chief of Staff

B. Officers appointed by the Board of Selectpersons<sup>260</sup>. The Board of Selectpersons shall appoint the <u>Appointed</u> Town Officers listed in §§7.13 through 7.17 of this Charter.

Internal Auditors Assessor
Tax Collector Tree Warden
Constables

C. Officers appointed by other bodies<sup>261</sup>. The <u>Appointed</u> Town Officers hereunder shall be appointed by the bodies specified in §§87.18 through 87.24 of this Charter.

Director of Health

Fire Chief

Animal Control Officer

Town Librarian

Chief of Police

Planning Director

Conservation Director

#### Officers Appointed by the First Selectperson

<sup>&</sup>lt;sup>258</sup> 2022 recodification of current Article IX, §9.1 (2006). Derived from Article IX, §9.1 of the 1997 Charter. <sup>259</sup> 2022 modification and recodification of current Article IX, §9.2 (2006). Derived from Article IX, §9.2 of the 1997 Charter.

<sup>&</sup>lt;sup>260</sup> 2022 recodification of current Article IX, §9.14 (2006). Derived from Article IX, §9.12 of the 1997 Charter. <sup>261</sup> 2022 recodification of current Article IX, §9.19 (2006). Derived from Article IX, §9.17 of the 1997 Charter.

#### §7.2. Town Attorney<sup>262</sup>.

- **A.** Appointment and qualifications. The Town Attorney shall be appointed by the First Selectperson and shall be an attorney admitted to practice in the State who has practiced in the State for at least five (5) years. The Town Attorney may be removed by the First Selectperson without cause.
- **B.** Compensation. The Town Attorney shall receive the compensation approved by the Board of Selectpersons within the appropriations made for the Town Attorney.

#### **C. Duties.** The Town Attorney shall:

- (1) Be the legal advisor of, and counsel and attorney for, the Town and all Town Departments and Town Officers Officials and the RTM, providing all necessary legal services in matters relating to the Town's interests or the official powers and duties of the Town officers and employees;
- (2) Prepare or review all Contracts and other instruments to which the Town is a party or in which it has an interest;
- (3) Upon the request of the Board of Selectpersons, or of any other Town Officer, Board or Commission, give a written opinion on any question of law relating to the powers and duties of the officer or body making the request;
- (4) Upon the direction of the Board of Selectpersons, appear for the Town or any Town Officers, Boards and Commissions, or Departments in any litigation or any other action brought by or against the Town or any Town OfficersOfficials; and
- **(5)** Prosecute or defend, appeal from or defend appeals from, and make settlements of, litigation and claims, as the Board of Selectpersons direct.

#### §7.3. Assistant Town Attorneys<sup>263</sup>.

**A. Number of positions and appointment.** The Board of Selectpersons, with the consent of the RTM, shall determine from time to time the number of Assistant Town Attorneys necessary to effectively carry out the legal business of the Town and shall increase or reduce the number of Assistant Town Attorney positions to the number determined to be necessary. The First Selectperson may then appoint the Assistant Town Attorneys and may remove Assistant Town Attorneys without cause.

**B.** Qualifications. Each Assistant Town Attorney shall be an attorney admitted to practice in the State.

 <sup>262 2022</sup> recodification of current Article IX, §9.3 (2006). Modification of Chapter VI of the 1947 Act; further amended by §9 of the 1951 Special Act and reaffirmed by Chapter VI of the 1956 Act and 1975 Charter.
 263 2022 recodification of current Article IX, §9.4 (2006). Derived from Article IX, §9.4 of the 1997 Charter.

**C. Duties.** The Assistant Town Attorneys shall have the duties assigned to them by, and shall be under the supervision of, the Town Attorney.

#### §7.4. Fiscal Officer and Controller<sup>264</sup>.

**A.** Appointment and qualifications. The Fiscal Officer shall be appointed by the First Selectperson and shall be a certified or a licensed public accountant or otherwise have experience in the financial field, in accordance with the provisions of §6.2.B of this Charter.

#### **B. Duties of the Fiscal Officer.** The Fiscal Officer shall:

- (1) Establish and supervise a central accounting and internal auditing system;
- **(2)** Be responsible for, and conduct a continuously current accounting of, the financial activities of the Town, including the Board of education;
- (3) Audit, before payment, all payrolls, bills, invoices and claims drawn against the Town, including the Board of Education; and
  - (4) Countersign and approve each requisition to be paid by the Town.
- **C. Duties of the Controller**. The Controller shall have the same qualifications as the Fiscal Officer and shall have the duties assigned by, and be under the supervision of, the Fiscal Officer. The Controller may countersign requisitions in the absence of the Fiscal Officer.

#### §7.5. Town Treasurer<sup>265</sup>.

**A. Appointment and eligibility**. The Town Treasurer shall be appointed by the First Selectperson and may be removed by the First Selectperson without cause. The Fiscal Officer may also serve as Town Treasurer. No person employed by or in the Purchasing Department may at the same time be Town Treasurer[S22].

**B. Duties.** The Town Treasurer shall have the duties conferred on town treasurers by Chapter 94 of the General Statutes (C.G.S. § 7-79 et seq.) and shall serve as an ex officio member of the Board of Library Trustees. The Town Treasurer may be part-time.

<sup>&</sup>lt;sup>264</sup> 2022 modification and recodification of current Article IX, §9.5 (2006). Derived from Article IX, §9.5 of the 1997 Charter.

<sup>&</sup>lt;sup>265</sup> 2022 recodification of current Article IX, §9.6 (2006). Derived from Article IX, §9.6 of the 1997 Charter.

#### §7.6. Director of Public Works<sup>266</sup>.

- **A. Appointment and qualifications.** The Director of Public Works shall be appointed by the First Selectperson in accordance with any requirements set forth in the General Statutes and in accordance with the provisions of §6.2.B of this Charter.
  - **B. Duties.** The Director of Public Works shall:
    - (1) Administer and supervise the Department of Public Works;
  - (2) Have charge and control of all buildings, materials, apparatus, equipment, and documents of the Department of Public Works;
  - (3) BeFollow Town Human Resources policies and procedures to appoint a Town Engineer who shall be chief technical advisor of the Town and all Departments, except the Board of Education, in all matters concerning the physical development of the Town and the design, construction, and maintenance of its physical plant[\$S23];
  - (4) Be an ex officio member without vote of the Town Plan and Zoning Commission with respect to planning and of, the Parks and Recreation Commission and the Flood Prevention, Climate Resilience and Erosion Control Board or, in the alternative may appoint a designee on said Board [S24];
  - (5) Have the ability to make and enforce reasonable rules and regulations not inconsistent with this Charter or the General Statutes necessary to efficiently exercise all powers and duties imposed on the Director of Public Works and the Department of Public Works;
  - **(6)** At the request of the Selectpersons, furnish technical advice reasonably required for the physical functioning of the Town or its government; and
  - (7) Perform other reasonable and related duties as directed by the Selectpersons.
- **C. Assistants.** The Director of Public Works may request that the First Selectperson engage superintendents and assistants to the Director of Public Works.

<sup>&</sup>lt;sup>266</sup> 2022 modification and recodification of current Article IX, §9.7 (2006). Modification of Chapter IX of the 1947 Act and 1956 Act. Section 1 of the Act established the powers and duties of a Town Engineer as the director of the Department. Section 3 established the position of the Superintendent of Highways and Bridges. Section 4 addressed engineering issues; see, also Chapter IX of the 1975 Charter; and, Article IX, §9.7 of the 1997 Charter.

#### §7.7. Building Official<sup>267</sup>.

- **A. Appointment and term.** The Building Official shall be appointed by the First Selectpersons and shall serve for a term of four (4) years.
- **B.** Number of Assistant Building Officials and Building Inspectors. The First Selectperson and the Building Official shall determine from time to time the number of Assistant Building Officials and Building Inspectors necessary to carry out the duties of the Building Department and, subject to appropriation, shall increase or decrease the number accordingly.

#### C. Duties.

- (1) The Building Official and the Assistant Building Officials shall have the duties conferred by this Charter and on building officials generally by Chapter 541 of the General Statutes (C.G.S. § 29-250 et seq.).
- (2) In particular, the Building Official and the Assistant Building Officials, under the supervision of the Building Official, shall:
  - (a) Inspect, supervise, regulate, and control the construction, reconstruction, altering, repairing, demolition, and removal of all structures within the Town:
  - **(b)** By diligent search and inspection, enforce all Laws, Ordinances, and Regulations governing the construction of buildings and other structures;
    - **(c)** Enforce the provisions of the Town building code;
  - **(d)** Cooperate and coordinate with the Fire Marshal in inspecting structures and enforcing provisions of the Town building code where fire safety may be a factor; and
  - **(e)** Keep complete public records of all applications made to them and of all permits and certificates of approval or occupancy issued by them and any other records required under Chapter 541.
- **D.** Building Inspectors<sup>268</sup>. Building Inspectors shall perform such duties as the

<sup>&</sup>lt;sup>267</sup> 2022 recodification of current Article IX, §9.8 (2006). Modification of Chapter X, §1 of the 1947 Act which established a Board of Building Commissioner (§§1-4); the position of Building Inspector (§5); and, Article IX, §9.8 ("Building Inspectors") of the 1997 Charter.

<sup>&</sup>lt;sup>268</sup> 2022 recodification of current Article IX, §9.8.D (2006). Modification of Chapter X, §6 of the 1947 and 1956 Acts; and, Article IX, §9.8.D of the 1997 Charter.

Building Official shall assign within the scope permitted by Law.

#### §7.8. Purchasing Agent<sup>269</sup>.

- **A. Appointment and qualifications.** The Purchasing Agent shall be appointed by the First Selectperson and shall have a background in business or purchasing and procurement.
- **B. Duties**. The Purchasing Agent shall have the duties set forth in §4.3.G of this Charter and such other duties as may be prescribed by the First Selectperson.

#### §7.9. Director of Human and Social Services<sup>270</sup>.

- **A. Appointment and qualifications.** The Director of Human and Social Services shall be appointed by the First Selectperson and shall be trained in social services.
  - **B. Duties.** The Director of Human and Social Services shall:
    - (1) Administer and supervise a Department of Human Services;
    - (2) Prescribe the duties of subordinates and employees;
  - (3) Coordinate the work of the Department of Human Services with that of other government agencies, private social service organizations, and other entities on social services matters as authorized by the General statutes, including matters which relate to the aging, people with disabilities, veterans, youth and underserved populations of the Town;
  - (4) Perform other duties prescribed by the Human Services Commission and the First Selectperson; and
  - (5) Report to the First Selectperson on matters of administration and operation and to the Human Services Commission on matters of policy.

### §7.10. Director of Parks and Recreation<sup>271</sup>.

- **A. Appointment and qualifications.** The Director of Parks and Recreation shall be appointed by the First Selectperson and shall have such qualifications as may be established by the Parks and Recreation Commission.
  - **B. Duties.** The Director of Parks and Recreation shall:

<sup>&</sup>lt;sup>269</sup> 2022 recodification of current Article IX, §9.9 (2006). Derived from Article IX, §9.9 of the 1997 Charter. <sup>270</sup> 2022 recodification of current Article IX, §9.10 (2006). Derived from Article IX, §9.10 of the 1997 Charter ("Director of Human Services").

<sup>&</sup>lt;sup>271</sup> 2022 recodification of current Article IX, §9.11 (2006). Derived from Article IX, §9.11 of the 1997 Charter.

- (1) Administer and supervise the Parks and Recreation Department;
- (2) Recommend policy to the Parks and Recreation Commission;
- (3) Submit to the Parks and Recreation Commission plans for the development and maintenance of public cemeteries, parks, playgrounds, beaches, beach facilities, marina facilities, public gardens, and other recreational areas of the Town, except for areas and facilities under the control of the Board of Education, the Golf Commission, or the Harbor Management Commission;
- **(4)** Submit to the Golf Commission, the Board of Education, and the Harbor Management Commission plans for the development and maintenance of recreational areas under the control of the Golf Commission, Board of Education, or the Harbor Management Commission.
- (5) Submit to the Parks and Recreation Commission plans for recreation programs in the Town, except programs run by the Board of Education or the Golf Commission:
- **(6)** Implement the plans approved by the Parks and Recreation Commission;
- (7) Coordinate the maintenance of parks and recreation facilities with the Department of Public Works:
- (8) Coordinate any recreational activities in Town open space areas with the Conservation Commission;
- **(9)** Perform such other duties as directed by the Parks and Recreation Commission or the First Selectperson; and
- (10) Report to the First Selectperson on matters of administration and operation and to the Parks and Recreation Commission on matters of policy.

If the Golf Commission or Harbor Management Commission ceases to exist, reference to that Commission in this section of this Charter shall be inoperative.

#### §7.11. Director of Community and Economic Development<sup>272</sup>.

**A. Appointment and supervision.** The Director of Community and Economic Development shall be appointed by the First Selectperson and shall report to the First Selectperson and may be removed by the First Selectperson without cause.

\_

<sup>&</sup>lt;sup>272</sup> 2022 recodification of current Article IX, §9.12 (2006).

- В. **Duties.** The Director of Community and Economic Development shall have such duties as determined by the First Selectperson so as to manage and supervise policies and programs relating to economic development and affordable housing and any federal or state programs associated with such issues.
- C. Liaison to Economic Development Commission and Affordable Housing Task Force[S25]. The Director of Community and Economic Development shall serve as a liaison to the Economic Development Commission and the First Selectperson's Affordable Housing Committee.

#### Director of Human Resources<sup>273</sup>. §7.12.

- Appointment and supervision. The Director of Human Resources shall be appointed by the First Selectperson and shall report to the First Selectperson.
- B. Duties. The Director of Human Resources shall have such duties as determined by the First Selectperson so as to manage and supervise policies and programs relating to all Town personnel practices, policies, and functions and all risk management functions associated with such issues.

#### Officers Appointed by the Board of Selectpersons

#### Internal Auditor or Auditors<sup>274</sup>. §7.13.

- Number. The Board of Selectpersons shall appoint at least one (1) Internal Α. Auditor and shall determine from time to time the number of Internal Auditors necessary to carry out the duties of the office and shall accordingly increase or decrease the number of Internal Auditors appointed.
  - B. **Duties.** The Internal Auditor or Auditors shall:
  - Monitor the Departments, officers, employees, Boards and (1) Commissions of the Town for fiscal policy compliance;
    - (2) Report on a regular basis to the Fiscal Officer; and
  - (3) Make reports to the Board of Selectpersons and the Board of Finance semi-annually and at any other times requested by either board.

<sup>&</sup>lt;sup>273</sup> 2022 recodification of current Article IX, §9.13 (2006).

<sup>&</sup>lt;sup>274</sup> 2022 recodification of current Article IX, §9.15 (2006). Derived from Article IX, §9.13 of the 1997 Charter.

#### §7.14. Assessor<sup>275</sup>.

The Assessor shall be appointed by the Board of Selectpersons and shall have the powers and duties conferred on assessors generally by Chapter 203 of the General Statutes (C.G.S. § 12-40 et seq.), and, on request of the Board of Finance, shall report in writing concerning any matters pertaining to those duties. The Assessor shall have and maintain the certification required by § 12-40a of Chapter 203 of the General Statutes.

#### §7.15. Tax Collector<sup>276</sup>.

- **A. Appointment and supervision.** The Tax Collector shall be appointed by the Board of Selectpersons and shall act under the direction and control of the Fiscal Officer.
- **B. Duties.** The Tax Collector shall have the powers and duties conferred in this Charter, by Ordinance, and on tax collectors generally by Chapter 204 of the General Statutes (C.G.S. § 12-122 et seq.).
- **C. Delinquency list.** The Tax Collector shall present annually, and at any other time the Board of Finance requires, a list of all amounts remaining unpaid on the rate bill for nine (9) months after becoming due. The list shall contain the name and address of each delinquent taxpayer, the amount of the tax, and, as a separate item, the interestand other charges due.

#### §7.16. Tree Warden<sup>277</sup>.

The Tree Warden shall be appointed by the Board of Selectpersons and shall have a term of two years or as otherwise provided by the General Statutes. The Tree Warden shall have the powers and duties conferred by Ordinance and on tree wardens generally as set forth in the General Statutes<sup>278</sup>.

#### §7.17. Constables<sup>279</sup>.

A. Appointment<sup>280</sup>. There shall be four (4) Constables appointed by the Board

<sup>&</sup>lt;sup>275</sup> 2022 recodification of current Article IX, §9.16 (2006). Modification of Chapter XVII, §6 and §8 of the 1947 and 1956 Acts and Chapter XVII, §6 and §7 of the 1975 Charter; and, Derived from Article IX, §9.14 of the 1997 Charter

<sup>&</sup>lt;sup>276</sup> 2022 recodification of current Article IX, §9.17 (2006). Modification of Chapter XVII, §7 and §8 of the 1947 and 1956 Acts and Chapter XVII, §8 of the 1975 Charter; and, Article IX, §9.15 of the 1997 Charter.

<sup>&</sup>lt;sup>277</sup> 2022 modification and recodification of current Article IX, §9.18 (2006). Derived from Article IX, §9.16 of the 1997 Charter.

<sup>&</sup>lt;sup>278</sup> **Comment of the 2022 Charter Revision Commission.** At the time of the adoption of this Charter, C.G.S. §23-58 permits a "term of two years." The duties are set forth in C.G.S. §23-59 et seq.).

<sup>&</sup>lt;sup>279</sup> 2022 modification and recodification of current Article IX, §9.13 (2006).

<sup>&</sup>lt;sup>280</sup> 2022 modification and recodification of current Article VII, §7.2.A and B (2006)(Establishment and Powers and Duties Clauses). Derived from of Chapter II, §1 of the 1947 and 1956 Acts and the 1975 Charter; and

of Selectpersons- for staggered terms of four (4) years, subject to the transition provision. No more than two (2) shall be from the same political party[s26].

Transitional Provision. On or about December 1, 2023, the Board of Selectpersons shall appoint two (2) members to a term of four (4) years and two (2) members to a term of two (2) years. On and after December 1, 2025, all Constables shall be appointed to a term of four (4) years.

- **B.** Powers and duties<sup>281</sup>. Constables shall have the powers and duties prescribed by the General Statutes for their respective offices.
- **C. Vacancies**<sup>282</sup>. Vacancies shall be addressed as set forth in §6.6 of this Charter.

#### Officers appointed by other bodies.

#### §7.18. Director of Health<sup>283</sup>.

- A. Appointment and term. The Director of Health shall be appointed by the Board of Health with the approval of the First Selectperson and shall serve for a term of four (4) years.
- **B.** Qualifications<sup>284</sup>. The Director of Health shall be appointed in accordance with the requirements of the –General Statutes and in accordance with the provisions of §6.2.B of this Charter.
- **C. Powers and duties.** The Director of Health shall have the powers and the duties conferred by this Charter, by Ordinance, and by the rules and Regulations of the Board of Health, and on town directors of health generally by Chapter 368e of the General Statutes (C.G.S. §19a-200 et seq.) and state codes and Regulations.
- **D. Supervision.** The Director of Health shall report to the First Selectperson on matters of administration and operation and to the Board of Health on matters of policy.

Article VII, §7.2.A of the 1997 Charter. Note: At all times since 1947 there have been 7 constables.

<sup>&</sup>lt;sup>281</sup> 2022 modification and recodification of current Article VII, §7.2.B (2006). Derived from Article VII, §7.2.B of the 1997 Charter.

<sup>&</sup>lt;sup>282</sup> 2022 recodification and modification of current Article VII, §7.2.C (2006). Derived from Article VII, §7.2.C of the 1997 Charter.

<sup>&</sup>lt;sup>283</sup> 2022 modification and recodification of current Article IX, §9.20 (2006). Derived from Chapter VIII, §3 of the 1947 and 1956 Acts and 1975 Charter; and, Derived from Article IX, §9.18 of the 1997 Charter.

<sup>&</sup>lt;sup>284</sup> **Comment of the 2022 Charter Revision Commission:** At the time of adoption of this Charter the qualifications are set forth in Chapter 368e of the General Statutes (C.G.S. §19a-200 et seq.).

#### §7.19. Chief of Police<sup>285</sup>.

**A. Appointment.** The Chief of Police shall be appointed by the Police Commission, with the approval of the First Selectperson, from among the three (3) highest scoring candidates who have passed a competitive examination for Chief of Police.

#### B. Powers and duties. The Chief of Police shall:

- (1) Be the executive officer of the Police Department;
- (2) Have authority to direct and control the conduct of all members and other employees of the Police Department; and
  - (3) Keep all records required by Law and by the Police Commission.
- **C. Departmental discipline.** Subject to a contrary provision of a collective bargaining agreement, disobedience to the lawful orders of the Chief of Police shall be grounds for disciplinary action by the Police Commission. The Chief of Police shall have the power without consulting the Police Commission to impose fines of not more than two (2) Days' pay or suspension of not more than one (1) week for disobedience to the Chief's lawful orders or for violations of the rules and Regulations of the Police Department<sup>286</sup>. The Chief of Police may recommend to the Police Commission that it take more severe disciplinary action.
- **D. Supervision.** The Chief of Police shall report to the First Selectperson on matters of administration and operation and to the Police Commission on matters of policy.

#### §7.20. Fire Chief<sup>287</sup>.

**A. Appointment.** The Fire Chief shall be appointed by the Fire Commission, with the approval of the First Selectperson, from the three (3) highest scoring candidates who have passed a competitive examination for Fire Chief.

#### **B.** Powers and duties. The Fire Chief shall:

- (1) Be the executive officer of the Fire Department;
- (2) Have authority to direct and control the conduct of all members of the Fire Department; and
  - (3) Keep the records required by Law and by the Fire Commission.

<sup>&</sup>lt;sup>285</sup> 2022 recodification of current Article IX, §9.21 (2006). Derived from Article IX, §9.19 of the 1997 Charter.

<sup>&</sup>lt;sup>286</sup> Derived from §3 of the 1945 Act; amended by §24 of the 1951 Act.

<sup>&</sup>lt;sup>287</sup> 2022 recodification of current Article IX, §9.22 (2006). Derived from Article IX, §9.20 of the 1997 Charter.

- **C. Departmental discipline.** Subject to a contrary provision of a collective bargaining agreement, disobedience to lawful orders of the Fire Chief shall be grounds to disciplinary action by the Fire Commission. The Fire Chief shall have power without consulting the Fire Commission to impose fines of not more than two (2) Days' pay or suspension for not more than one (1) week for disobedience to lawful orders or for violations of the rules and Regulations of the Fire Department. The Fire Chief may recommend to the Fire Commission that it take more severe disciplinary action.
- **D. Supervision.** The Fire Chief shall report to the First Selectperson on matters of administration and operation and to the Fire Commission on matters of policy.

#### §7.21. Planning Director<sup>288</sup>.

- **A. Appointment.** The Planning Director shall be appointed by the Town Plan and Zoning Commission with the approval of the First Selectperson.
- **B. Duties.** The Planning Director shall have the duties prescribed by the Town Plan and Zoning Commission and the First Selectperson.
- **C. Supervision.** The Planning Director shall report to the First Selectperson on matters of administration and operation and to the Town Plan and Zoning Commission on matters of policy.

#### §7.22. Animal Control Officer<sup>289</sup>.

- **A. Appointment and supervision.** The Animal Control Officer shall be appointed by the Police Commission and shall report to the Chief of Police.
- **B.** Powers and duties. The Animal Control Officer shall have the powers and duties prescribed by the Police Commission and conferred on animal control officers generally by Chapter 435 of the General Statutes.

#### §7.23. Conservation Director<sup>290</sup>.

- **A. Appointment.** The Conservation Director shall be appointed by the Conservation Commission with the approval of the First Selectperson.
- **B. Duties.** The Conservation Director shall have the duties prescribed by the Conservation Commission and the First Selectperson.

<sup>&</sup>lt;sup>288</sup> 2022 recodification of current Article IX, §9.23 (2006). Derived from Article IX, §9.21 of the 1997 Charter.

<sup>&</sup>lt;sup>289</sup> 2022 recodification of current Article IX, §9.24 (2006). Derived from Article IX, §9.22 of the 1997 Charter.

**C. Supervision.** The Conservation Director shall report to the First Selectperson on matters of administration and operation and to the Conservation Commission on matters of policy.

#### §7.24. Town Librarian<sup>291</sup>.

**A. Appointment and qualifications.** The Town Librarian shall be appointed by the Board of Library Trustees, with the approval of the First Selectperson, and shall have such qualifications as may be required by the Board of Library Trustees.

#### **B. Duties.** The Town Librarian shall:

- (1) Report to and have such duties as are defined by the Board of Library Trustees on matters of policy, and by the First Selectperson on matters of administration.
- **(2)** Manage and supervise policy, programs and personnel relating to all public town libraries.
- (3) Have responsibilities as set forth by the Board of Library Trustees pursuant to Sections 8.17.C.

<sup>&</sup>lt;sup>291</sup> 2022 recodification of current Article IX, §9.26 (2006).

# ARTICLE VIII - SPECIFIC APPOINTED BOARDS AND COMMISSIONS, AND DEPARTMENTS

#### §8.1. Boards and Commissions<sup>292</sup>.

- A. Permanent Bodies Appointed by the First Selectperson<sup>293</sup>. The First Selectperson shall appoint the members of the bodies listed in this sub-section and in Section 8.2 of this Charter in the numbers and for the terms set forth therein: Conservation Commission.
- B. Permanent Bodies Appointed by the Board of Selectpersons<sup>294</sup>. The Board of Selectpersons shall appoint the members of the bodies listed in this sub-section and in Sections 8.3 through 8.14 of this Charter in the numbers and for the terms set forth therein:

Police Commission and Dept.
Police and Fire Retirement Board
Board of Health/Public Health
Dept.
Board of Building Appeals
Water Pollution Control Authority
Human Services Commission and
Dept.

Fire Commission and Department Parks and Recreation Commission Flood Prevention, Climate Resilience and Erosion Control Board

Historic District Commission Golf Commission

- C. Boards and Commissions Appointed by the Board of Selectpersons and Approved by RTM<sup>295</sup>. The Board of Selectpersons shall appoint the members of the body listed in this sub-section and in Sections 8.15 of this Charter in the numbers and for the terms set forth therein: Ethics Commission.
- **D.** Self-Perpetuating Boards and Commissions approved by the Board of Selectpersons<sup>296</sup>. The Board of Selectpersons shall approve the members of the body listed in this sub-section and in Sections 8.16 of this Charter in the numbers and for the terms set forth therein: Board of Library Trustees.

#### E. Meetings<sup>297</sup>.

(1) All appointed Boards and Commissions except the Ethics Commission

<sup>&</sup>lt;sup>292</sup> 2022 recodification of current Article X, §10.1.A (2006). Derived from Article X, §10.1.A of the 1997 Charter

<sup>&</sup>lt;sup>293</sup> 2022 recodification of current Article X, §10.2 (2006). Derived from Article X, §10.2 of the 1997 Charter.

<sup>&</sup>lt;sup>294</sup> 2022 recodification of current Article X, §10.4 (2006). Derived from Article X, §10.4 of the 1997 Charter.

<sup>&</sup>lt;sup>295</sup> NEW (2022).

<sup>&</sup>lt;sup>296</sup> NEW (2022).

<sup>&</sup>lt;sup>297</sup> 2022 recodification of current Article X, §10.1.B (2006). Derived from Article X, §10.1.B of the 1997 Charter.

and the Board of Building Appeals shall hold at least ten (10) regular stated meetings a year and shall give annual notice of such meetings as required by the General Statutes. The Ethics Commission and the Board of Building Appeals shall meet when they have business to transact. Officers of each Board and Commission shall be elected annually at an organization meeting so noticed by the Town Clerk held in the month of December, with the exception of the Ethics and Golf Commissions, shall be elected annually in the month of April. No person sitting on a Board or Commission at the designation of another Board or Commission may be elected an officer of the Board or Commission to which he or she has been designated.

F. **Vacancies**<sup>298</sup>. Vacancies shall be filled as set forth in §6.6 of this Charter.

Bodies appointed by the First Selectperson.

#### §8.2. Conservation Commission<sup>299</sup>.

Α. Members and terms. The Conservation Commission shall consist of seven (7) members, not more than four (4) of whom shall be registered with the same political party, and three alternate members, not more than two (2) of whom shall be registered with the same political party. Both members and alternate members shall be appointed by the First Selectperson for a term of five (5) years. Members' terms shall be staggered so that no more than two (2) members' terms expire in one (1) year. Alternate members' terms need not be staggered.

#### B. Powers and duties (Conservation Functions).

- The Conservation Commission shall have all of the powers and duties (1) conferred by this Charter, by Ordinance, and on conservation commissions generally by §7-131a of Chapter 97 of the General Statutes.
- In order to carry out its powers, the Conservation Commission (2) shall:
  - Conserve. develop. supervise, regulate natural (a) and resources, including water resources and open space land in the Town;
  - Conduct investigations into the use and possible use of land in (b) the Town:
  - Keep an index of all open areas, publicly or privately owned, for the purpose of obtaining information on the proper use of such areas;

<sup>&</sup>lt;sup>298</sup> 2022 revision and recodification of current Article X, §10.1.C (2006). Derived from Article X, §10.1.C of the 1997 Charter.

<sup>&</sup>lt;sup>299</sup> 2022 recodification of current Article X, §10.3 (2006). Derived from Article X, §10.3 of the 1997 Charter.

- **(d)** Have the ability to recommend to appropriate agencies plans and programs for the development and use of open areas;
- **(e)** Have the ability, as approved by the RTM, to acquire land and easements in the name of the Town and promulgate rules and regulations, including but not limited to the establishment of reasonable charges for the use of land andeasements, for any of its purposes; and
- **(f)** Have the ability to coordinate the activities of unofficial bodies organized for similar purposes.
- C. Powers and duties (Inland Wetland Functions). The Conservation Commission shall have the powers and duties conferred by this Charter, by ordinance, and on inland wetlands and watercourses agencies generally by §§ 22a-42 to 22-44 of Chapter 440 of the General Statutes. In particular, the Commission shall:
  - (1) Provide for the protection, preservation, maintenance and use of inland wetlands and watercourses, for their conservation, economic, aesthetic, recreational, and other public and private uses and values in order to provide to the citizens of the Town an orderly process to balance the need for the economic growth of the Town and the use of its land with the need to protect the environment and its natural resources;
  - (2) Adopt, amend and promulgate such Regulations as are necessary to protect and define the inland wetlands and watercourses;
    - (3) Develop a comprehensive program in furtherance of its purposes;
  - **(4)** Advise, consult and cooperate with other agencies of the Town, State and Federal governments;
  - **(5)** Encourage and conduct studies and investigations and disseminate relevant information; and
  - **(6)** Inventory and evaluate the inland wetlands and watercourses in such form as it deems best suited to effect its purposes.
- **D. Director and other employees or consultants.** The Commission shall appoint a Director with the approval of the First Selectperson. The Commission shall have the power to engage such employees or consultants as it requires to carry out its duties, including a wetlands administrator and assistants who, subject to the general supervision of the Director, shall enforce all Laws, ordinances and Regulations relating to matters over which it has jurisdiction and who shall have such other duties as the Commission or the Director may prescribe.

Permanent bodies appointed by the Board of Selectpersons.

#### §8.3. Police Commission and Department<sup>300</sup>.

- **A. Members and terms.** The Police Commission shall consist of seven (7) members appointed by the Board of Selectpersons, no more than four (4) of whom shall be registered with the same political party. Each member shall have a term of five (5) years with the terms staggered so that not more than two (2) terms expire in one (1) year.
- **B.** Powers and duties. The Police Commission shall have the powers and duties conferred on police commissions generally by §7-276 of Chapter 104 of the General Statutes, except as those may be limited by this Charter. In particular, the Police Commission shall:
  - (1) Have general management supervision of the Police Department of the Town and of all property and equipment used by or in connection with the operation of the Department;
  - (2) Make rules and Regulations consistent with the General Statutes and this Charter for the governance of the Police Department and its personnel, and may prescribe penalties for violations of its rules and Regulations;
  - (3) Subject to the provisions in this Charter regarding appointment of the Chief of Police, have sole power to appoint and promote to all positions in the Police Department, which appointments and promotions shall be made on the basis of merit;
  - **(4)** Determine the qualifications for each rank and grade in the Police Department; and
  - (5) Within the appropriations made for that purpose determine the number of officers and other employees of the Police Department and of the several ranks and grades and their compensation.
- **C. Appointment of Acting Chief.** During the absence or disability of the Chief of Police, the Police Commission may designate a member of the Department as Acting Chief of Police to perform the duties of the Chief of Police.
- **D.** Appointment of special officers. Special officers for the protection of specified private property and special traffic duty or for the preservation of peace may be appointed by the Commission under such rules and Regulations as the Commission shall

 $<sup>^{300}</sup>$  2022 recodification of current Article X, §10.5 (2006). Derived from Chapter XIV of the 1947 Act, based upon §1 of Special Act No. 186 (1945); further amended by §23 of the 1951 Act and affirmed by Chapter XIV of the 1956 Act and the 1975 Charter; and, Article X, §10.5 of the 1997 Charter.

from time to time establish. Such officers shall serve at the pleasure of the Commission and in any event for terms not to exceed one (1) year unless re-appointed.

#### §8.4. Fire Commission and Department<sup>301</sup>.

- **A. Members and terms.** The Fire Commission shall consist of seven (7) members appointed by the Board of Selectpersons, no more than four (4) of whom shall be members of the same political party. Each member shall have a term of five (5) years with the terms staggered so that no more than two terms expire in one (1) year.
- **B.** Powers and duties. The Fire Commission shall have all the powers and duties conferred on fire commissions generally by §7-301 of Chapter 104 of the General Statutes, except as those may be limited by this Charter. In particular, the Fire Commission shall:
  - (1) Have general management and supervision of the Fire Department of the Town and of all property and equipment used by or in connection with the operation of the Department, including the hydrants used for fire purposes;
  - (2) Make rules and Regulations consistent with the General Statutes and this Charter for the governance of the Fire Department and its personnel, and may prescribe penalties for violations of its rules and Regulations;
  - (3) Subject to the provisions in this Charter regarding appointment of the Fire Chief, have sole power to appoint and promote to all positions in the Department, and all appointments and promotions in the Fire Department shall be made on the basis ofmerit;
  - (4) Determine the qualifications for each rank and grade in the Fire Department; and
  - (5) Within the limits of the appropriations made for that purpose, determine the number of persons employed by the Department and of the several ranks and grades and their compensation.
- **C.** Appointment of Acting Fire Chief. During the absence or disability of the Fire Chief, the Commission may designate a member of the Department as Acting Fire Chief to perform the duties of the Fire Chief.

<sup>&</sup>lt;sup>301</sup> 2022 recodification of current Article X, §10.6 (2006). Modification of Chapter XV of the 1947 and 1956 Acts and 1975 Charter. At the time the Commission membership was based upon the fire districts in town. Derived from Article X, §10.6 of the 1997 Charter.

#### §8.5. Police and Fire Retirement Board<sup>302</sup>.

- A. Members and terms. The Police and Fire Retirement Board shall consist of seven (7) members: The First Selectperson (who shall be Chair), the Fiscal Officer, three (3) members appointed by the Board of Selectpersons for a term of three (3) years, and one (1) member to serve for a period of one (1) year, to be elected by the members of the Fire Department and members of the Police Department, respectively. No more than two (2) of tethree (3) members appointed by the Board of Selectpersons shall be registered with one (1) political party and their terms shall be staggered so that not more than two (2) terms expire in one (1) year.
- **B.** Powers and duties. The Police and Fire Retirement Board shall be the trustees of the retirement fund created under the existing police and fire retirement system for regular firefighters and police officers employed by the Town and shall have full control and management of the fund, with the power to invest and reinvest the same in accordance with the General Statutes respecting the investment of trust funds. The Police and Fire Retirement Board shall be responsible for assuring that the administration of the Police and Retirement plan is in strict accordance with the plan documents.
- **C.** Annual report. On or before January 1 of each year, the Police and Fire Retirement Board shall file an annual report with the Board of Selectpersons showing the financial condition of the police and fire retirement system as of the end of the last-completed fiscal year, including an actuarial evaluation of assets and liabilities, and setting forth such other facts, recommendations and data as may be of value to the members of the police and fire retirement system of the Town.
- **D.** Changes in retirement system. Before any change in the police and fire retirement system negotiated under the General Statutes is submitted for consideration to the Board of Finance, and for action to the RTM, the First Selectperson shall secure a written actuarial evaluation and report of such change.

#### §8.6. Department of Public Works<sup>303</sup>.

The Department of Public Works shall have all of the administrative powers and duties vested in the Town by this Charter or by the General Statutes with respect to the following functions of the Town:

**A.** The construction, reconstruction, care, maintenance, operation, altering, paving, repairing, draining, cleaning, snow clearance, lighting, and inspection of all Town streets, highways, bridges, sidewalks, curbs, street signs, guide posts, dams, incinerators, dumps, water supply, sewerage systems, and other public improvements, and buildings and equipment owned or used by the Town, except school buildings and

<sup>&</sup>lt;sup>302</sup> 2022 recodification of current Article X, §10.7 (2006). Derived from Chapter XXI of the 1975 Charter and Article X, §10.7 of the 1997 Charter.

<sup>303 2022</sup> recodification of current Article X, §10.8 (2006). Derived from Article X, §8 of the 1997 Charter.

equipment, police and fire equipment, and buildings and equipment under the control of the Board of Library Trustees;

- **B.** The removal of encroachments and, together with the Tree Warden, the planting, preservation, care and removal of trees, shrubs and other vegetation within highways, or public places, or on Town property;
- **C.** The maintenance, care and improvement of, and construction work required in connection with, public cemeteries, parks, playgrounds, beaches, marina facilities, and recreational areas of the Town, as requested by the Selectpersons, the Parks and Recreation Commission, the Conservation Commission, the Board of Education, or other bodies as may be designated by Ordinance.
- **D.** The custody of all maps of the Town not otherwise entrusted to any other Department, Board, Commission, or Town OfficerOfficial; and
- **E.** The maintenance of maps or other records showing highways, building or veranda lines, street profiles, and plans and profiles of storm and sanitary sewers.

#### §8.7. Board of Health and Public Health Department<sup>304</sup>.

- A. Members and terms of the Board of Health<sup>305</sup>. The Board of Health shall consist of seven (7) members, five (5) of them appointed by the Board of Selectpersons for four (4) year terms, one (1) designated by the Board of Education from its members, and one (1) designated by the Human Services Commission from its members. At least one (1) of the members so appointed shall be a physician. At least two (2) additional members shall be licensed health care professionals. Not more than three (3) of the members appointed by the Board of Selectpersons shall be registered with the same political party and their terms shall be staggered so that no more than two (2) terms expire in one (1) year. The members designated by the Board of Education and by the Human Services Commission shall not serve beyond their tenure on the appointing bodies.
- **B.** Organization. The Board of Health shall be the general policy-making body for the Public Health Department and shall make all necessary rules and Regulations for its administration.
- **C. Personnel**<sup>306</sup>. The Public Health Department shall consist of a Director of Health and such sanitarians, nurses, dental hygienists, secretaries, clerks and other personnel as may be necessary to operate the Department, including such physicians as may be temporarily engaged from time to time.

-

<sup>&</sup>lt;sup>304</sup> 2022 recodification of current Article X, §10.9 (2006). Derived from Article X, §10.9 of the 1997 Charter.

<sup>&</sup>lt;sup>305</sup> Derived from Chapter VIII, §1 of the 1947 Act; as further amended by §10 of the 1951 Act and reaffirmed by Chapter VIII, §1 of the 1956 Act and the 1975 Charter.

<sup>306</sup> Derived from Chapter VIII, §2 of the 1947 Act and the 1956 Act.

#### D. Powers and duties.

- (1) The Board of Health may make and amend such reasonable rules for the promotion and preservation of the public health, health services in public schools, and sanitation as required, provided the same shall not be inconsistent with the General Statutes, the state public health code, other state departmental regulations, or the ordinances and public health code of the Town.
- (2) The Board of Health shall advise the Director of Health in all matters relating to public health, and health services in the schools of the Town, and shall appoint the school and Well Child Clinic medical advisors.

#### §8.8. Parks and Recreation Commission<sup>307</sup>.

#### A. Members and terms.

- (1) The Parks and Recreation Commission shall consist of nine (9) voting members, eight (8) of whom shall be appointed by the Board of Selectpersons and one (1) of whom shall be a member of the Board of Education designated by the Board of Education to serve on the Parks and Recreation Commission. Notwithstanding §6.3 of this Charter, no more than five (5) of the members appointed by the Board of Selectpersons shall belong to the same political party. The Director of Parks and Recreation and the Director of Public Works shall serve on the Parks and Recreation Commission ex officio without vote.
- (2) The term of office of the members appointed by the Board of Selectpersons shall be five (5) years with terms staggered so that no more than two (2) members are appointed in any one year.
- **B.** Powers and duties. The Parks and Recreation Commission shall be the policy-making body for the Parks and Recreation Department, shall together with the First Selectperson supervise the Director of Parks and Recreation, and shall assist in establishing requirements and qualifications for personnel needed for the Department.
- **C. Department.** The Parks and Recreation Department shall consist of a Director of Parks and Recreation and such other personnel as may be necessary to operate the Department.

<sup>&</sup>lt;sup>307</sup> 2022 recodification of current Article X, §10.10 (2006). Derived from Chapter XIII of the 1947 and 1956 Acts and the Charter of 1975. At the time the Commission consisted of five members; as further amended by §17 of the 1951 Act. A Board of Recreation existed under Chapter XVI of the 1947 Act and 1975 Charter; §1 was amended and §4 was repealed by §§18 and 19 of the 1951 Act. Chapter XVI, §2 remained in effect under the 1956 Act. Derived from Article X, §10.10 of the 1997 Charter.

#### §8.9. Board of Building Appeals<sup>308</sup>.

- **A. Members and terms.** The Board of Building Appeals shall consist of five (5) members appointed by the Board of Selectpersons, no more than three (3) of whom shall be registered with the same political party. The members shall have terms of five (5) years, which shall be staggered so that no more than one (1) term expires in one (1) year.
- **B.** Qualifications. The members shall have the qualifications set forth in the State Building Code.
- **C. Powers and duties**. The Board of Building Appeals shall have the powers and duties conferred on boards of building appeals generally by § 29-266 of Chapter 541 of the General Statutes, including hearing appeals from the decisions of the Building Inspector.

#### §8.10. Flood Prevention, Climate Resilience and Erosion Control Board<sup>309</sup>.

- A. Members and terms. The Flood Prevention, Climate Resilience and Erosion Control Board shall consist of five (5) members appointed by the Board of Selectpersons, no more than three (3) of whom shall be registered with the same political party. The members shall have terms of five (5) years, which shall be staggered so that not more than one (1) term expires in one (1) year. The Director of Public Works, or a designee appointed by the Director same by a non-voting ex officio member of the CommissionBoard.
- **B.** Powers and duties. Said Board shall have the powers and duties conferred on flood and erosion control boards generally by §25-84 through 25-94[528] of Chapter 477 of the General Statutes, as amended.
- **C.** Temporary members. If any member of the Board is disqualified from participating in any appraisal of damages or assessment of benefits, the remaining members of the Board shall appoint an elector as a temporary board member. The temporary board member shall have all of the powers and duties of the disqualified member, but only with respect to the matter as to which the member is disqualified.
- C. Alternate Members [S29]. There shall be three (3) alternate members of the Board, no more than two (2) of whom shall be registered with the same political party. Alternate members shall have terms of three (3) years, which shall be staggered so that not more than one (1) term expires in one year. Such alternate members shall, when seated as herein provided, have all the powers and duties set forth for such board and its members.

<sup>&</sup>lt;sup>308</sup> 2022 recodification of current Article X, §10.11 (2006). See, §2 of Chapter X of the 1947 Act ("Board of Building Commissioners"); further amended by §11 of the 1951 Act and reaffirmed by Chapter X of the 1956 Act and the 1975 Charter; and, Article X, §10.11 of the 1997 Charter.

 $<sup>^{309}</sup>$  2022 modification and recodification of current Article X, §10.12 (2006). Derived from Article X, §10.12 of the 1997 Charter. Comment of the 2022 Charter Revision Commission: The change in title corresponds to the legislative changes made in P.A. 21-115.

If a regular member is absent or is disqualified, the chairman of the board shall designate an alternate to so act, choosing alternates in rotation so that they shall act as nearly equal a number of times as possible. If any alternate is not available in accordance with such rotation, such fact shall be recorded in the minutes of the meeting"

**Transitional Provision.** On or about the fourth (4<sup>th</sup>) Monday of November 2022, the Board of Selectpersons shall appoint an alternate to a term of three (3) years; an alternate to a term of two (2) years; and, an alternate to a term of one (1) year. On and after the fourth (4<sup>th</sup>) Monday of November 2023, as their terms expire, alternates shall be appointed to a term of three (3) years.

#### §8.11. Water Pollution Control Authority<sup>310</sup>.

- **A. Members and terms**. The Water Pollution Control Authority shall consist of seven (7) members, one (1) of whom shall be a member of the Board of Selectpersons, and six (6) of whom shall be appointed by the Board of Selectpersons. No more than four (4) of the members appointed by the Board of Selectpersons shall be registered with the same political party. The members appointed by the Board of Selectpersons shall have terms of four (4) years, which shall be staggered so that not more than two (2) terms expire in one (1) year. The Fiscal Officer shall be a member ex officio, without vote.
- B. Organization and personnel. For purposes of this Charter, the Water Pollution Control Authority shall be deemed a commission except where any provision of this Charter of general application to commissions conflicts with a provision of the General Statutes concerning water pollution control authorities. The Water Pollution Control Authority may establish rules and adopt bylaws for the transaction of its business. The clerk of the Water Pollution Control Authority shall keep a record of its proceedings and shall be custodian of all books, papers and other documents of the Water Pollution Control Authority. The Water Pollution Control Authority may employ such personnel as may be required for the performance of its duties and may fix their compensation.
- **C. Powers and duties.** The Water Pollution Control Authority shall have the power to:
  - (1) Operate, maintain and manage, and through the Department of Public Works, plan, lay out, acquire, construct, reconstruct, equip, repair, maintain, supervise and manage and, through the Department of Public Works, operate a sewerage system;
  - **(2)** Acquire, by purchase, condemnation or otherwise, any real property or interest in real property which it shall determine to be necessary for use in connection with such sewerage system;

<sup>&</sup>lt;sup>310</sup> 2022 modification and recodification of current Article X, §10.13 (2006). Derived from Article X, §10.13 of the 1997 Charter.

- (3) Apportion and assess the whole or any part of the cost of acquiring, constructing or reconstructing any sewerage system or portion thereof upon the lands and buildings in the Town which, in its judgment, shall be especially benefited by the system (whether they abut on such system or not), and upon the owners of such lands and buildings and fix the time when such assessments shall be due and payable and provide that they may be paid in such number of substantially equal annual installments, not exceeding thirty (30), as it shall determine;
- (4) Establish and from time to time revise just and equitable charges or rates for connection with and use of the sewerage system;
- (5) Order any owner or occupant of any real estate to which the sewerage system is available to connect the drainage and sewerage thereof with the system and to disconnect, fill up and destroy any cesspool, privy vault, drain or other arrangement on such real estate for the reception of such drainage or sewerage; and
- **(6)** Generally, to have and possess all of the powers and duties conferred upon water pollution control authorities by the General Statutes.

#### §8.12. Historic District Commission<sup>311</sup>.

- A. Members and terms. The Historic District Commission shall consist of five (5) members, no more than three (3) of whom shall be registered with the same political party, and three (3) alternate members, no more than two (2) of whom shall be registered with the same political party. Both members and alternate members shall be appointed by the Board of Selectpersons, with the advice and consent of the RTM. Terms of membership shall be for five (5) years and shall be staggered so that no more than one (1) member's term and one (1) alternate member's term expires in each year.
- **B.** Powers and duties. The Historic District Commission shall have the powers and duties conferred upon historic district commissions and historic properties commissions generally by Chapter 97a of the General Statutes (C.G.S. §7-147a et seq.).

#### §8.13. Human Services Commission and Department<sup>312</sup>.

A. Members and terms. The Human Services Commission shall consist of nine (9) members appointed by the Board of Selectpersons, not more than five (5) members of whom shall be registered with the same political party. Members shall have terms of four (4) years which shall be staggered so that no more than four (4) terms expire in one (1) year. The Human Services Commission shall represent those social services matters in accordance with the authority set forth under the General Statutes or otherwise by Law,

<sup>&</sup>lt;sup>311</sup> 2022 recodification of current Article X, §10.14 (2006). Derived from Article X, §10.14 of the 1997 Charter. <sup>312</sup> 2022 recodification of current Article X, §10.16 (2006). Derived from Article X, §10.16 of the 1997 Charter.

including matters which relate to the aging, people with disabilities, veterans, youth and underserved populations; unless other committee, task force or Board or Commission has been authorized to specifically address any particular social service matter. The Commission may form its own committees with additional persons to assist and advise the Human Services Commission in matters which concern the conditions and needs of aging and of <a href="https://handle.com/handl

**B.** Powers and duties. The Human Services Commission shall be the policy-making body for the Department of Human Services.

#### C. Department of Human Services.

- (1) Powers and duties. The Department of Human Services, together with the Board of Selectpersons, shall have all the powers and duties relating to social services granted to and imposed upon towns by the General Statutes. The Department of Human Services shall act on behalf of the Town in all social service matters in conjunction with State and Federal agencies. Social service matters involving community health issues shall be coordinated with the Public Health Department.
- (2) Acceptance and use of private donations. The Department of Human Services shall have the power to accept on behalf of the Town donations of any kind to be used generally or specifically for its purposes and to carry out any specific wishes of a donor. The power to accept donations shall not be construed to eliminate the authority any other Town officerOfficial or bodythe RTM may have to review specific donations. All donated monies shall be delivered to the Town Treasurer to be maintained in a special account subject to the order of the Department of Human Services, in accordance with the terms of the gift in each instance.

#### §8.14. Golf Commission<sup>313</sup>.

- **A. Members and terms.** The Golf Commission shall consist of seven (7) members appointed by the Board of Selectpersons, no more than four (4) of whom shall be registered with same political party. Each member shall have a term of five (5) years, commencing April 1, with the terms staggered so that no more than two (2) terms expire in the same year. No member shall be eligible for reappointment to the Golf Commission for a period of five (5) years after the end of his or her term.
- **B.** Powers and duties. The Golf Commission shall be the policy-making body for the Par 3 Golf Course and the H. Smith Richardson Golf Course. It shall have the following powers, subject to appropriation:

<sup>&</sup>lt;sup>313</sup> 2022 recodification of current Article X, §10.18 (2006). Derived from Article X, §10.18 of the 1997 Charter.

- (1) To make, amend and repeal bylaws, rules and regulations relative to play, hours of operation, fees, charges, and all other decisions necessary for the successful operation of those courses;
- (2) To fix and revise from time to time and to charge and collect fees, rents and other charges for the use of any golf course facilities on behalf of the Town in an amount sufficient to maintain operating and maintenance expenses.

#### Appointed by Board of Selectpersons and Approved by RTM

#### §8.15. Ethics Commission<sup>314</sup>.

#### A. Members and terms.

- (1) The Ethics Commission shall consist of five (5) members, appointed by unanimous vote of the Board of Selectpersons and confirmed by a Majority Vote of the RTM. No more than three (3) members shall be registered with the same political party[S30].
- (2) Notice of appointment shall be served by the Board of Selectpersons upon the Moderator of the RTM and the Town Clerk. A vote for approval or rejection of each person appointed shall be taken at an RTM meeting held more than ten (10) Days after service of the notice on the Town Clerk. Failure to vote within sixty (60) Days of the service on the Town Clerk shall be deemed to be approval and confirmation by the RTM. If any appointment is rejected by the RTM, the Board of Selectpersons shall within twenty-one (21) Days after the rejection notify the RTM Moderator and Town Clerk of further appointments to replace the rejected appointments. The RTM shall then vote on the new appointments. These too shall be deemed approved and confirmed if not voted upon within sixty (60) Days of the notice.
- (3) The terms shall commence on April 1. Terms of members shall be two (2) years and shall be staggered so that no more than three (3) terms expire in one (1) year. No member may serve more than the shorter of two (2) terms or one (1) term plus a partial term created by filling a vacancy for an unexpired term.

Transitional Provision [831]. On or about July 1, 2024, the Board of Selectpersons shall appoint two (2) members to a term ending on March 31, 2025; said vacancy shall be filled on April 1, 2025 for a term of two (2) years. On or about July 1, 2023, the Board of Selectpersons shall appoint three (3) members to a term ending on March 31, 2024; said vacancy shall be filled on April 1, 2024 for a term of two (2) years.

<sup>&</sup>lt;sup>314</sup> 2022 recodification of current Article X, §10.15 (2006). Derived from Article X, §10.15 of the 1997 Charter.

#### **B.** Powers and duties. The Ethics Commission shall:

- (1) Receive complaints alleging violations of the Standards of Conduct or any ordinance establishing a Code of Ethics for Town OfficersOfficials and employees;
- (2) Upon sworn complaint or upon the vote of three (3) members, investigate the actions and conduct of elected and Appointed Town Officers Officials, members of the RTM, and employees of the Town to determine whether there is probable cause that a violation has occurred of the Standards of Conduct or Code of Ethics;
- (3) On its own motion issue general opinions and interpretations of the Standards of Conduct or the Code of Ethics;
- (4) Upon the request of a principal officer of a Department, or any member of a Town Board or Commission, or any member of the RTM render an advisory opinion with respect to any specific relevant situation under the Standards of Conduct or Code of Ethics;
- (5) Consider written requests for advisory opinions referred by a department head with respect to any problem submitted to the department head in writing by an employee in the Department (whose name need not be disclosed to the Ethics Commission) concerning that employee's duties in relationship to the Standards of Conduct or Code of Ethics where the department head elects not to decide the issue within the Department;
- **(6)** Adopt such regulations as it deems advisable to assure procedures for the orderly and prompt performance of the Commission's duties;
- (7) Upon a finding of probable cause initiate hearings to determine whether there has been a violation of the Standards of Conduct or Code of Ethics;
- (8) Have the power to retain its own counsel, administer oaths, issue subpoenas and subpoenas *duces tecum* (enforceable upon application to the Superior Court) to compel the attendance of persons at hearings and the production of books, documents, records, and papers; and
- (9) Upon finding of a violation of the Standards of Conduct or Code of Ethics, at its discretion, recommend appropriate disciplinary action to the Board of Selectpersons or appropriate Appointed Town Officers, including department heads.

#### C. Procedure.

(1) On complaints.

- (a) In any investigation to determine probable cause the Ethics Commission shall honor all requests for confidentiality, consistent with the requirements of State Law. Unless a finding of probable cause is made or the individual against whom a complaint is filed requests it, complaints alleging a violation of the Standards of Conduct or Code of Ethics shall not be disclosed by the Ethics Commission.
- **(b)** Any person accused of a violation shall have the right to appear and be heard by the Ethics Commission and to offer any information which may tend to show there is no probable cause to believe the person has violated any provision of the Standards of Conduct or the Code of Ethics.
- **(c)** The Ethics Commission shall, not later than ten (10) Days after the termination of its probable cause investigation, notify the complaining person and the person against whom the complaint was made that the investigation has been terminated and the results.
- (d) At hearings after a finding of probable cause, the Ethics Commission shall afford the person accused the protection of due process consistent with that established for state agencies under the "Connecticut Uniform Administrative Procedures Act," including but not limited to the right to be represented by counsel, the right to call and examine witnesses, the right to the production of evidence by subpoena, the right to introduce exhibits, and the right to cross-examine opposing witnesses.
- **(e)** In the absence of extraordinary circumstances, the hearing shall be held within ninety (90) Days of the initiation of the investigation. The Ethics Commission shall, not later than thirty (30) Days after the close of the hearing, publish its findings together with a memorandum of its reasons. Any recommendation for disciplinary action shall be contained in the findings.
- (f) An individual directly involved or directly affected by the action taken as a result of the Ethics Commission's findings or recommendation may seek judicial review of such action and of the Ethics Commission's findings or recommendation unless the action taken was a referral of the matter to proper authorities for criminal prosecution.
- (2) On requests for advisory opinions. Within forty-five (45) Days from the receipt of a request for an advisory opinion, the Ethics Commission shall either render the opinion or advise as to when the opinion shall be rendered.
- **D.** Quorum. A quorum for the Ethics Commission shall be not less than four (4) members in attendance. All members who attended all hearings on the matter, and all members who certify that they have read or heard the entire transcript of the hearing they

did not attend, shall be eligible to vote on the proposed Ethics Commission action. The Ethics Commission shall find no person in violation of any provision of the Standards of Conduct or Code of Ethics except upon the concurring vote of three-fourths (3/4<sup>ths</sup>) of those members voting.

#### **Self-Perpetuating Board Approved by Board of Selectpersons**

#### §8.16. Board of Library Trustees<sup>315</sup>.

**A. Members and terms.** The Board of Library Trustees shall consist of the Chief Fiscal Officer, ex officio without vote, and six (6) trustees appointed in the following manner: Annually, the Board of Library Trustees, with the approval of the Board of Selectpersons, shall appoint one (1) trustee to serve for a term of three (3) **years**[S32].

Transitional Provision. On or about December 1, 2023, the Board of Selectpersons shall appoint two (2) members to a term of four (4) years and two (2) members to a term of two (2) years. On and after December 1, 2025 all Constables shall be appointed to a term of four (4) years.

**B.** Vacancy and reappointment limitation. Any vacancy in the Board of Library Trustees, from any cause other than the expiration of a term, shall be filled for the remainder of the term by appointment by the remaining trustees, with the approval of the Board of Selectpersons.

#### **C. Powers and duties**. The Board of Library Trustees shall:

- (1) Manage, control, maintain, and operate all property of the Town devoted to library purposes, except such property as may be under the jurisdiction of the Board of Education:
- (2) Identify and adopt written policies to govern the operation and programs of the Library;
- (3) Seek adequate funds to carry out Library operations and monitor the use of those funds to provide Library services;
- **(4)** Turn over money which may be collected from revenue generating services to the Town Treasurer;
- (5) Subject to appropriation appoint a Town Librarian with the approval of the First Selectperson and maintain a regular performance appraisal process; and,

<sup>&</sup>lt;sup>315</sup> 2022 recodification of current Article X, §10.17 (2006). Derived from §1 of the 1949 Act and, further amended by §22 of the 1951 Act; and; Chapter XXV of the 1956 Special Act; Chapter XXII of the 1975 Charter; and, Article X, §10.17 of the 1997 Charter.

**(6)** Ensure that the Library has a strategic plan with implementation and evaluation components.

### D. Acceptance and use of private donations.

- (1) Acceptance and use. Subject to the provisions contained in this Charter and in the General Statutes, the Board of Library Trustees may accept any gift of property of any character upon any terms and conditions which the donor may prescribe and which may be acceptable to the Board of Library Trustees, provided no gift which imposes upon the Town an obligation to incur any expense in order to keep, use or maintain the gift may be accepted by the Board of Library Trustees unless it is approved by the RTM. The Board of Library Trustees may establish one (1) or more library funds withany of such property and shall have the exclusive control and management of, may hold title to, and may manage and invest and reinvest, the property in accordance with the Laws of the State governing the investment of trust funds.
- which any of such property or funds shall be held, the Board of Library Trustees shall transferis permitted to use the gross income (Dividends) at least quarterly to the Chief Fiscal Officer to be expended by the Town for capital improvements to the library or for such special purposes as may be required to comply with the terms and conditions of any gift or by state standards. The principal within the funds is not to be used for operational budgeting purposes, it and is governed and managed solely by the Board of Trustees. The use of any principal within the funds is to be used only for capital investment improvements (outside of annual operating budgets) and thus voted on by the Board of Trustees[s33]. Subject to appropriation, the Board of Library Trustees may employ such agents, experts, and other personnel as it may deem advisable in connection with the administration and management of any of such property or funds.

CHARTER OF THE TOWN OF FAIRFIELD

# **ARTICLE IX - BUDGET PROCEDURE AND RELATED MATTERS**

# §9.1. The Budget Process<sup>316</sup>.

- A. Cooperation of Town OfficersOfficials and Employees<sup>317</sup>. The First Selectperson, Board of Selectpersons, Board of Finance, the Board of Education and RTM are required to work together, in good faith, throughout the year in order to develop and approve a Town budget which. The budget shall include a delineation of all expected revenues and expenditures and, for the purposes of short- and long-term financial planning, include detailed estimates of revenues, capital expenses and operating expenses all as required by this Charter and any Ordinances pertaining thereto. Each of these officials and employees of the Town are required to utilize best practices in the field of municipal and public finance, in order to comply with Law and with the generally accepted accounting principles (or such successor policies thereto) and shall embrace principles of accountability, transparency and outreach in order to expand public participation, engagement and trust in the budgetary process of the Town.
- **B.** Fiscal year<sup>318</sup>. The fiscal year of the Town shall commence on July 1 and conclude on June 30 unless otherwise set forth in the General Statutes.
- **C. Budget a public record: Public Inspection**<sup>319</sup>**.** The entire budget, comprised of the general fund budget and capital budget, shall be a public record in the office of the Town Clerk and shall be open to public inspection at other designated public facilities including libraries and <u>schoo9lsschools</u>, as may be determined by the First Selectperson, Board of Selectpersons, Board of Education or the RTM. Moreover, the First Selectperson shall provide access on-line through social media, the Town web-site and dashboards and email chains to community organizations and members of the public who request such information.
- **D.** Public Engagement<sup>320</sup>. The First Selectperson, Board of Selectpersons, Board of Finance, Board of Education and RTM shall develop procedures designed to encourage public participation in the budget process.
- **E. Budget Calendar**<sup>321</sup>. Not later than the second (2<sup>nd</sup>) Meeting of the Board of Selectpersons, in January of each year, the Chief Fiscal Officer or equivalent official responsible for the management of the budget process<sub>[S34]</sub> shall, in accordance with §9.1.A of this Charter, following consultation with the First Selectperson, Board of Selectpersons, the Chair of the Board of Finance and Moderator of the RTM<sub>[S35]</sub>, cause to be published a budget calendar in order to inform the public of the significant

317 NEW (2022).

<sup>316</sup> NEW (2022).

<sup>318</sup> NEW (2022).

<sup>319</sup> NEW (2022).

<sup>320</sup> NEW (2022).

<sup>321</sup> NEW (2022).

requirements of the budget process, including but not limited to:

- (1) Submission of Budget Estimates by Submitting Parties, as set forth in §9.3.A of this Charter;
- (2) Date(s) of the Joint Review of the First Selectperson's Initial Recommendations by the Board of Selectperson, Board of Finance and RTM, as set forth in §9.3.B of this Charter;
- (3) Recommendation of the Proposed Executive Budget to the Board of Finance, as set forth in §9.4.A of this Charter;
- (4) Date of the Public Hearing of the Board of Finance as set forth in §9.5.B of this Charter;
- **(5)** Date of Adoption of the Proposed Town Budget as set forth in §9.5.C of this Charter:
- **(6)** Date of the Annual Budget Meeting and Final RTM Action on the Approved Annual Town Budget, as set forth in §§9.2 and 9.6.B of this Chapter;
- (7) Proposed date of Board of Finance, as shall be set by the Board of Finance, determination of property tax rate as set forth in §§9.5.E and 9.7 of this Charter.

# §9.2. Date of annual budget meeting<sup>322</sup>.

The RTM shall hold the annual budget meeting on the second (2<sup>nd</sup>) Monday in May of each year ("RTM Annual Budget Meeting").

### §9.3. Annual Budget Estimates.

A. Submission of General Fund and Capital Budget Estimates to the First Selectperson<sup>323</sup>. All Town Officers, Boards and Commissions Officials and Departments of the Town, including the Board of Education ("Submitting Parties"), shall submit to the First Selectperson<sup>324</sup>:

(1) such items and details of their respective general fund and capital

<sup>&</sup>lt;sup>322</sup> 2022 modification and recodification of current Article XII, §12.1 (2006). Derived from Article XII, §12.1 of the 1997 Charter; and, Chapter III, §6 (second sentence) and Chapter XVII, §4 of the 1947 and 1956 Acts and 1975 Charter.

<sup>&</sup>lt;sup>323</sup> 2022 modification and recodification of current Article XII, §12.2,A (2006). Derived from Article XII, §12.2 of the 1997 Charter.

<sup>&</sup>lt;sup>324</sup> 2022 modification and recodification of current Article XII, §12.2,A (2006)(First sentence). Derived from Article XII, §12.2 of the 1997 Charter.

budgets for the next fiscal year<sup>325</sup>; and,

(2) any additional information which they possess (including, but not limited to, records, books, accounts, Contracts, reports and other papers and documents as specified by the First Selectperson ("Budget Estimates") all of which, in the judgment of the First Selectperson, are necessary to discharge the duties imposed upon the First Selectperson by this Charter.

Said Budget Estimates shall be submitted on or prior to a date designated by the First Selectperson, which date shall be early enough for the First Selectperson to review, revise, compile and transmit recommendations to the Board of Selectpersons, Board of Finance and RTM for purposes of a Multi-Board Budget Workshop with said Submitting Parties ("First Selectperson's Budget Recommendation") 326.

- B. Presentation before Joint Meetings of the Board of Selectpersons, Board of Finance and RTM<sup>327</sup>[S36].
  - (1) First Selectperson's Budget Recommendation. Not later than second (2<sup>nd</sup>) Monday in March, the First Selectpersons' Budget Recommendation shall be submitted to the Board of Selectpersons for consideration and to the Board of Financer Finance and RTM for initial review. At such time the First Selectperson shall address said Budget Recommendation before a joint meeting of the Board of Selectpersons, Board of Finance and RTM.
  - Multi-Board Budget Meeting Meetings. Following submission and budget address, the First Selectperson shall convene aan initial joint budget Meeting of the Board of Selectpersons and the Board of Finance for the purpose of receiving testimony and information from all Submitting Parties, including the Board of Education on the Budget Recommendation. Said testimony shall assist the bodies in their respective reviews of the said recommendations and the impact on the Departments and taxpayers. The Meeting
    - (a) All subsequent Meetings shall be called to order by the First Selectperson and be facilitated, in turn, by the Chair of the Board of Finance and/or the Moderator of the RTM, should the RTM be involved in the Meeting<sup>328</sup> and shall be conducted in accordance with rules of order to

-

<sup>&</sup>lt;sup>325</sup> 2022 modification and recodification of current Article XII, §12.2,A (2006)(First sentence) setting forth the definition of Budget Estimate". Derived from Article XII, §12.2 of the 1997 Charter.

<sup>&</sup>lt;sup>326</sup> 2022 modification and recodification of current Article XII, §12.2.A (2006)(Second sentence). Derived from Article XII, §12.2 of the 1997 Charter.

<sup>&</sup>lt;sup>327</sup> NEW (2022)

<sup>&</sup>lt;sup>328</sup>-Comment of the 2022 Charter Revision Commission: Both the Board of Finance and RTM will have additional opportunities to discuss the budget with the First Selectperson and Submitting Parties throughout the process. The objective of this first session is to eliminate redundancy and duplication of effort where possible. We recommend the RTM create a robust committee system and allow the members of committee

facilitate a comprehensive review to the benefit of the participating bodies and the public.—329 [837].

(2)(b) The rules may permit expansion of the Meeting to include the RTM in the event RTM committees (not to exceed ten (10) members) would participate as a rotating group of interlocutors to participate in the questioning of the Submitting Parties.

# §9.4. Review and recommendation by Board of Selectpersons to the Board of Finance.

- A. Proposed Executive Budget; Submission to Board of Finance<sup>330</sup>. Following the joint meetings, set forth in §9.3.B, and any other meetings it deems necessary to review the First Selectperson's Budget Recommendation say, the Board of Selectpersons shall make recommendation of a Proposed Executive Budget to the Board of Finance, not later than the fourth (4<sup>th</sup>) Monday of March. The Proposed Executive Budget shall be in the form, and shall contain the details, required by the Board of Finance from time to time<sup>331</sup>.
- **B.** Variation of procedure<sup>332</sup>. The Board of Selectpersons, with the approval of the Board of Finance, may modify and vary the budget submission process in the interest of efficiency or in the event of special circumstances.

### §9.5. Review and recommendation by Board of Finance.

- **A.** Further examination<sup>333</sup>. The Board of Finance may hold meetings to review the Proposed Executive Budget as it determines necessary prior to the Public Hearing set forth in §9.5.B of this Charter.
- **B.** Public hearing by Board of Finance<sup>334</sup>. The Board of Finance shall hold a public hearing on the Proposed Executive Budget prior to the public meeting set forth in §9.5.C of this Charter.

to handle the questions for the Submitting Parties over which they have subject-matter jurisdiction.

<sup>329</sup> **Comment of the 2022 Charter Revision Commission:** Both the Board of Finance and RTM will have additional opportunities to discuss the budget with the First Selectperson and Submitting Parties throughout the process. The objective of this first session is to eliminate redundancy and duplication of effort where possible. We recommend the RTM create a robust committee system and allow the members of committee to handle the questions for the Submitting Parties over which they have subject-matter jurisdiction.

<sup>&</sup>lt;sup>330</sup> 2022 modification and recodification of current Article XII, §12.2.B (2006). Derived from Article XII, §12.2 of the 1997 Charter.

<sup>&</sup>lt;sup>331</sup> 2022 Recodification of current Article XII, §12.3.A (2006). Derived from Article XII, §12.3 of the 1997 Charter.

<sup>&</sup>lt;sup>332</sup> 2022 Recodification of current Article XII, §12.2.C (2006). Derived from Article XII, §12.2 of the 1997 Charter.

<sup>333</sup> NEW (2022).

<sup>&</sup>lt;sup>334</sup> 2022 modification and recodification of current Article XII, §12.3.B (2006). Derived from Article XII, §12.3 of the 1997 Charter.

- **C. Proposed Town Budget: Submission to RTM**<sup>335</sup>. After the public hearing as set forth in §9.5.B of this Charter, above, the Board of Finance shall hold a public meeting not later than the fourth (4<sup>th</sup>) Monday in April at which time it shall act upon all matters relating to the Proposed Executive Budget. Thereafter, the Board of Finance's Proposed Town Budget shall be submitted to the RTM.
- **D.** Board of Finance Presentation of the Proposed Town Budget to RTM<sup>336</sup>. The Board of Finance shall present the Proposed Town Budget to the RTM for consideration at the RTM Annual Budget Meeting.
- **E. Determination of property tax rate**<sup>337</sup>**.** After the RTM Annual Budget Meeting and receipt of the report on the grand list from the Board of Assessment Appeals, the Board of Finance shall determine the rate of property tax for the next fiscal year, taking into account the provisions of §9.7, below.

### §9.6. Review and determination by the RTM: Approved Annual Town Budget.

- **A.** RTM Deliberations and Further Examination<sup>338</sup>. The RTM may hold meetings, as it determines necessary before the Public Hearing set forth in §9.6.C of this Charter and the Annual Budget Meeting.
- **B.** The Approved Annual Town Budget<sup>339</sup>. At the Annual Meeting, the RTM shall act upon the Annual Town Budget for the next fiscal year:

# §9.7. Effect of referendum on the budget<sup>340</sup>.

Any item in the Approved Annual Town Budget referred to a referendum vote as provided in §3.6 of this Charter, above and disapproved shall be amended to accord with such vote. In the event of a referendum affecting any item contained in the annual Town budget, the time within which the Board of Finance shall determine the Town tax for the year following such appropriation shall be extended to five (5) Days after the referendum vote.

<sup>&</sup>lt;sup>335</sup> 2022 modification and recodification of current Article XII, §12.3.C (2006). Derived from Article XII, §12.3 of the 1997 Charter.

<sup>&</sup>lt;sup>336</sup> 2022 modification and recodification of current Article XII, §12.3.D (2006). Derived from Article XII, §12.3 of the 1997 Charter.

<sup>&</sup>lt;sup>337</sup> 2022 modification and recodification of current Article XII, §12.3.E (2006). Derived from Article XII, §12.3 of the 1997 Charter.

<sup>&</sup>lt;sup>338</sup> 2022 modification and recodification of current Article XII, §12.4 (2006)(First sentence). Derived from Article XII, §12.4 of the 1997 Charter.

<sup>&</sup>lt;sup>339</sup> 2022 modification and recodification of current Article XII, §12.4 (2006)(Second sentence). Derived from Article XII, §12.4 of the 1997 Charter.

<sup>&</sup>lt;sup>340</sup> 2022 modification and recodification of current Article XII, §12.5 (2006). Derived from Article XII, §12.5 of the 1997 Charter.

### §9.8. Appeals from the Board of Finance.

- A. Appeals to RTM<sup>341</sup>. Any Town Officer, Official (where a Board, or Commission, by majority vote of its members) or Department of the Town may appeal to the RTM from a vote of the Board of Finance to recommend a reduction in the amount of any request by said Town Officer, Board, Commission, Official or Department for an appropriation of Town funds as part of the annual budget process or at another time in the fiscal year, or for a budget transfer. The Town Officer, Official (where a Board, or Commission, by a majority vote of its members) or Department may appeal to restore the entire amount originally requested or any part of such amount specified in the appeal.
- **B. Method of appeal**<sup>342</sup>. The appeal shall be made in writing and shall be filed with the Town Clerk within ten (10) Days after written notice of the vote of the Board of Finance shall have been received by the Town Officer, Board, Commission, Official or Department making the appeal.
- **C.** RTM hearing<sup>343</sup>. Not later than the date of the RTM Annual Budget Meeting if the appeal is from a budget request, or the next regular meeting of the RTM after receiving an appeal from a vote of the Board of Finance in any other case, the RTM shall:
  - (1) Hold a hearing on such appeal, at which both the Board of Finance and the appellant shall be entitled to be heard;
  - **(2)** At the conclusion of the hearing, put the question of sustaining the appeal to a vote.
- **D. Vote necessary to sustain appeal**<sup>344</sup>. If two-thirds or more of the total number of RTM members present and voting at such meeting shall vote to sustain the appeal, the requested appropriation or transfer shall be made without the recommendation of the Board of Finance, subject, with respect to the appropriation, to referendum as provided in this Charter.

# §9.9. Expenditure in excess of appropriation forbidden<sup>345</sup>.

No Town Officer, Board, Commission, Official or Department shall expend any sum for any purpose in excess of the amount appropriated by the Town for such purpose unless such expenditure shall first be approved, and appropriate transfers in the budget made, by the Board of Finance.

<sup>&</sup>lt;sup>341</sup> 2022 modification and recodification of current Article XII, §12.6.A (2006). Derived from Article XII, §12.6 of the 1997 Charter.

<sup>&</sup>lt;sup>342</sup> Recodification of current Article XII, §12.6.B (2006). Derived from Article XII, §12.6 of the 1997 Charter.

<sup>&</sup>lt;sup>343</sup> 2022 modification and recodification of current Article XII, §12.6.C (2006). Derived from Article XII, §12.6 of the 1997 Charter.

<sup>&</sup>lt;sup>344</sup> 2022 modification and recodification of current Article XII, §12.6.C (2006). Derived from Article XII, §12.6 of the 1997 Charter.

<sup>&</sup>lt;sup>345</sup> Recodification of current Article XII, §12.7 (2006). Derived from Article XII, §12.7 of the 1997 Charter.



# ARTICLE X - MISCELLANEOUS[839]

### §10.1. Official Seal<sup>346</sup>.

The Town shall adopt, by Ordinance, a Town Seal with such suitable inscription or design as it determines. Said seal shall be filed with the Office of the Secretary of the State by the Town Clerk. The Town Clerk shall have custody of the seal.

# §10.2. Existing ordinances<sup>347</sup>.

All ordinances of the Town shall continue in full force and effect, except as they are inconsistent with the provisions of this Charter.

# §10.3. Separate provisions<sup>348</sup>.

If any provision of this Charter is declared by a court of competent jurisdiction to be void or unconstitutional, such action shall not affect the validity of any other provision.

# §10.4. Periodic Review of the Charter<sup>349</sup>.

No later than April of 2032, and no more than every ten (10) years thereafter, the Board of Selectpersons shall appoint a Charter Revision Commission, pursuant to the General Statutes. Nothing in this section limits the right of the Board of Selectpersons to appoint one or more Charter Revision Commissions other than the Commission required by this section, but no such Commission shall be appointed if its appointment would preclude the Board of Selectpersons from appointing a Commission as required by this section.

### §10.5. Submission and effective date<sup>350</sup>.

This Charter shall be submitted to the <u>electors Electors</u> of the Town at the general election to be held Tuesday, November 7, 2006.8, 2022. Voting shall be in accordance with the Laws of the State of Connecticut and the proposed Charter may be submitted in the form of one (1) or several questions as determined by the Board of Selectpersons. The Charter or such portions thereof as may be approved by the <u>electors Electors</u> of the Town shall take effect on November 27, 2006.

<sup>&</sup>lt;sup>346</sup> 2022 modification and recodification of current Article XIV, §14.1. Derived from Chapter XXIV, §4 of the 1947 and 1956 Acts; and Chapter XXVI, §4 of the 1975 Charter. See, C.G.S. §7-101.

<sup>&</sup>lt;sup>347</sup> 2022 recodification of current Article XIV, §14.2. Derived from Chapter XXIV, §5 of the 1947 and 1956 Acts; and Chapter XXVI, §5 of the 1975 Charter.

<sup>&</sup>lt;sup>348</sup> 2022 recodification of current Article XIV, §14.3. Derived from Chapter XXIV, §6 of the 1947 and 1956 Acts; and Chapter XXVI, §6 of the 1975 Charter.

<sup>349</sup> NEW (2022).

<sup>&</sup>lt;sup>350</sup> 2022 recodification of current Article XIV, §14.4. Derived from Chapter XXIV, §7 of the 1947 and 1956 Acts; and, and Chapter XXVI, §9 of the 1975 Charter.

Addendum<sub>[S40]</sub>

Organization 2022, with the exception of Town Government 351,

The government provisions pertaining to the term of the Town shall consist office, composition of the following Town officers and bodies as well as other Town officers and bodies not described in this Charter but designated by ordinanceentity or the General Statutes:

### A. Elected Positions 352.

Number	Body of Position	<del>Term</del>
<del>56 (max.)</del>	Representative Town Meeting	2 years <sup>353</sup>
3	Board of Selectpersons	4 years <sup>354</sup>
1	Town Clerk	4 years <sup>355</sup>
2	Registrars of Voters	4 years <sup>356</sup>
7	Constables	2 years <sup>357</sup>
9	Board of Finance	6 years <sup>358</sup>
5	Board of Assessment Appeals	4 years <sup>359</sup>
9	Board of Education	4 years 360
5	Zoning Board of Appeals	4 or 2
		<del>years 361</del>
3	Zoning Board of Appeals Alternates	4 years
7	Town Plan and Zoning Commission	4 or 2
		<del>years 362</del>

<sup>&</sup>lt;sup>351</sup> Recodification of current Article I, §1.4. Derived from Article I, §1.4 of the 1997 Charter

<sup>&</sup>lt;sup>352</sup> Recodification of current Article I, §1.4.A. Derived from Article I, §1.4.A of the 1997 Charter.

<sup>&</sup>lt;sup>353</sup> Derived from Chapter I, §4 and Chapter II, §1 and §4(b) of the 1947 and 1956 Acts; and, Chapter I, §4 and Chapter II, §1 and §4(c) of the 1975 Charter.

Derived from Chapter II, §1 and §4(b) of the 1947 and 1956 Acts; as, reenacted in Chapter II, §1 and §4(c) of the 1975 Charter and Article I, §1.5.A of the 1999 Charter [3 members—two year term]. The four year term was adopted in the 2006 Charter.

<sup>355</sup> Derived from Chapter II, §1 and §4(b) and Chapter VII of the 1947 and 1956 Acts; as reenacted by Chapter II, §1 and §4(c) and Chapter VII of the 1975 Act; Article I, §1.4.A, Article II, §2.3.B and Article VII, §7.1 of the 1997 Charter.

<sup>&</sup>lt;sup>356</sup> Derived from Chapter II, §1 and §4(b) of the 1947 and 1956 Acts; as, reenacted in Chapter II, §1 and §4(c) of the 1975 Charter and Article I, §1.4.A of the 1999 Charter. Note: There were two Registrars of Voters. The four year term was adopted in the 2006 Charter.

<sup>&</sup>lt;sup>357</sup> Derived from Chapter II, §1 and §4(b) of the 1947 and 1956 Acts; as, reenacted in Chapter II, §1 and §4(c) of the 1975 Charter and Article I, §1.4.A of the 1999 Charter.

<sup>&</sup>lt;sup>358</sup> Derived from Chapter II, §1 and §4(c) of the 1947 and 1956 Acts; as, reenacted in Chapter II, §1 and §4(d) of the 1975 Charter and Article I, §1.4. A of the 1997 Charter [9 members].

<sup>&</sup>lt;sup>359</sup> Derived from Article I, §1.4.A of the 1997 Charter.

<sup>&</sup>lt;sup>360</sup> Derived from Chapter II, §1 and 4(e) of the 1947 and 1956 Acts [six year term]. Amended by Chapter II, §1 and §4(e) of the 1975 Charter (four year term) and Article I, §1.4.A of the 1997 Charter [six year term]. The current four year term was adopted in the 2006 Charter.

<sup>&</sup>lt;sup>361</sup> Note: The ZBA was an appointed Board; see, Chapter II, §2 of the 1947 and 1956 Acts. The elected ZBA was enacted in Chapter II, §4(g) of the 1975 Charter and Article I, §1.4.A of the 1997 and 2006 Charter.

<sup>&</sup>lt;sup>362</sup> Derived from Chapter II, §1 and 4(e) of the 1947 and 1956 Acts [six year term]. A two member Board of Tax Review and an appointed Chair were included in said 1947 Charter; see also, Chapter XVIII of the 1947 and 1956 Acts and,

3		Town Plan and Zoning Commission Alternates	4 years		
B. Appoin					
(1) Appointed by the First Selectman:					
Numl	<del>ber</del>	Body of Position	<del>Term</del>		
1		Town Attorney <sup>364</sup>			
Indefi	inite	Assistant Town Attorneys <sup>365</sup>			
1		Fiscal Officer <sup>366</sup>			
1		Controller <sup>367</sup>			
1		Town Treasurer 368			
1		Director of Community and Economic <sup>369</sup> -			
		Development			
1		Director of Human and Social Services 370			
1		Director of Public Works <sup>371</sup>			
1		Building Official 372	4 years		
1		Director of Parks and Recreation 373			
1		Purchasing Agent 374			
1		Director of Human Resources 375			

Chapter II, §1 of the 1956 Act and Chapter XIX of the 1975 Charter. Further reenacted in Chapter II, §1 and §4(h) and (i) of the 1975 Charter and Article I, §1.4.A of the 1997 and 2006 Charters.

<sup>&</sup>lt;sup>363</sup> Current Article I, §1.4.B (2006).

<sup>&</sup>lt;sup>364</sup> Derived form Chapter II, §2 of the 1947 and 1956 Acts and Chapter II, 2 of the 1975 Charter and Article I, §1.B.i of the 1997 and 2006 Charters, respectively. The was also a Director of Welfare (see also, Chapter V of the 1947 Act and 1956 Act), Town Engineer, Dog Warden and a three member Retirement Board, all appointed by the Board of Selectmen; a Building Inspector appointed by the Building Commission; a Highway Superintendent appointed by the Director of Public Works; and, a Town Health Officer appointed by the Board of Health. The Retirement Board was included in Chapter XX of the 1947 Act and 1956 Act.

<sup>365</sup> See, prior fn.

<sup>&</sup>lt;sup>366</sup> Derived from Article II, 2 of the 1975 Charter and Article I, §1.B.i of the 1997 and 2006 Charters respectively. Note: the position of Assistant Fiscal Officer was set forth in the 1997 Charter.

<sup>&</sup>lt;sup>367</sup> Current Article I, §1.4.B.i (2006).

<sup>&</sup>lt;sup>368</sup> Derived from Chapter II, §2 of the 1947 and 1956 Act. Treasurer was an elected official. The Treasurer became an appoint officer in Chapter II, §2 of the 1997 Charter and Article I, §1.B.i of the 2006 Charter. Note: There was an Assistant Treasurer list in the 1975 and 1997 Charters.

<sup>&</sup>lt;sup>369</sup> Current Article I, §1.B.i (2006).

<sup>&</sup>lt;sup>370</sup> Current Article I, §1.B.i (2006).

<sup>&</sup>lt;sup>371</sup> Derived from Chapter II, §2 of the 1947 and 1956 Acts [delineated as the "town engineer and director of public works"]; reenacted as the Director of Public Works in Chapter II, §2 of the 1975 Charter and Article I, §4.2.B.i of the 2006 Charter.

Derived from Chapter II, §2 of the 1947 and 1956 Acts. Note: The Building Inspector was eliminated in the 1975 Charter yet included in Article I, §1.4.B.i of the 1997 Charter. Note: The "Building Official" was, again, set forth in Article I, §1.4.B.i of the 1997 and, as "Building Inspectors" in the 2006 Charter. Note: Is this the equivalent to the Building Official?

<sup>373</sup> Derived from Article I. §1.4.B.i of the 1997 and 2006 Charters.

<sup>&</sup>lt;sup>374</sup> Derived from Article I, §1.4.B.i of the 1997 and 2006 Charters.

<sup>&</sup>lt;sup>375</sup> Derived from Article I, §1.4.B.i of the 1997 and 2006 Charters

7	Conservation Commission <sup>376</sup>	<del>5 years</del>
3	Conservation Commission Alternates 377	<del>5 years</del>
<del>(2) App</del>		
Number	Body of Position	Term
<del>1 or more</del>	Internal Auditors <sup>379</sup>	
1	Assessor <sup>380</sup>	
1	Tax Collector <sup>381</sup>	
1	Tree Warden 382	<del>1 year</del>
7	Police Commission	5 years <sup>383</sup> -
7	Fire Commission	5 years 384
7	Police and Fire Retirement Board	3 or 1
		<del>years <sup>385</sup></del>
7	Board of Health	4 years <sup>386</sup>
8	Parks and Recreation Commission	5 years 387
<del>5</del>	Board of Building Appeals	5 years <sup>388</sup>
5	Flood and Erosion Control Board	5 years 389
5	Ethics Commission	2 vears <sup>390</sup>

<sup>&</sup>lt;sup>376</sup> Derived from Article I, §1.4.B.i of the 1997 and 2006 Charters.

<sup>377</sup> See, fn, above.

<sup>&</sup>lt;sup>278</sup> Chapter II, §2 of the 1947 and 1956 Actsincluded an appointed Zoning Board of Appeals and Building commission.

<sup>&</sup>lt;sup>379</sup> Derived from Article I, §1.4.B.ii of the 1997 and 2006 Charters

<sup>&</sup>lt;sup>380</sup> Derived from Chapter II, §2 of the 1947 and 1956 Acts, appointed by the Board of Finance. Thereafter appointed by the Board of Selectmen in Chapter II, §2 of the 1975 Charter and Article I, §1.4.B.ii of the 1997 and 2006 Charters.

<sup>&</sup>lt;sup>381</sup> Derived from Chapter II, §2 of the 1947 Act and 1956 Acts, appointed by the Board of Finance. Appointed by the Board of Selectmen in Article I, §1.4.B.ii of the 1997 and 2006 Charters,

<sup>&</sup>lt;sup>382</sup> Derived from Article I, §1.4.B.ii of the 1997 and 2006 Charters.

<sup>&</sup>lt;sup>383</sup> See Appendix for transition to increased membership. Modification of Chapter II, §2 of the 1947 and 1957 Acts and reaffirmed in Chapter II, §2 of the 1975 Charter and Article I, §1.4.B.ii of the 1997 and 2006 Charters. Note: There were five Police Commissioners.

<sup>&</sup>lt;sup>384</sup> Derived from Chapter II, §2 of the 1947 and 1957 Acts and reaffirmed in Chapter II, §2 of the 1975 Charter and Article I, §1.4.B.ii of the 1997 and 2006 Charters. See Appendix for transition to increased membership.

<sup>&</sup>lt;sup>385</sup> Derived from Article I, §1.4.B.ii of the 1997 and 2006 Charters. Selectmen appoint 3 members for 3 years; police and fire department members each appoint 1 member for 1 year. The final 2 members are the First Selectman and the Fiscal Officer. Note: A Retirement Board if enumerated in earlier versions of the Charter.

<sup>&</sup>lt;sup>386</sup> Derived from Chapter II, §2 of the 1947 and 1956 Acts and 1975 Charter and Article I, §1.4.B.ii of the 1997 and 2006 Charters. Selectmen appoint 5 members for 4 years; Board of Education and Human Services Commission each appoints 1 of its members, whose terms may not last beyond their terms on the Board of Education and Human Services Commission, respectively.

<sup>&</sup>lt;sup>387</sup> Derived from Chapter II, §2 of the 1947 and 1956 Acts and 1975 Charter (Separate Parks and recreation Commissions) and Article I, §1.4.B.ii of the 1997 and 2006 Charters. Selectmen appoint 8 members for 5 years; Board of Education appoints 1 of its members. Derived from Chapter II, §2 of the 1947 Act where there were five Park Commissioners and three members of the Board of Recreation. Reenacted by Chapter II, §2 of the 1956 Act.

<sup>388</sup> Derived from Chapter II, §2 of the 1975 Charter and Article I, §1.4.B.ii of the 1997 and 2006 Charters.

<sup>&</sup>lt;sup>389</sup> Derived from Article I, §1.4.B.ii of the 1997 and 2006 Charters.

<sup>&</sup>lt;sup>390</sup> Derived from Chapter II, §2 of the 1947 and 1956 Acts and 1975 Charter and Article I, §1.4.B.ii of the 1997 and 2006 Charters. Requires RTM approval.

7	<b>Water Pollution Contr</b>	4 years <sup>391</sup>	
<del>5</del>	Historic District Com	Historic District Commission	
3	Historic District Comr		5 years 5 years
9	Human Services Com		4 years 393
7	Golf Commission		5 years 394
	pointed by Other Bodie  Position		Town
Number		Appointing Authority Board of Health	Term
1	Director of Health		4 years <sup>395</sup>
1	Chief of Police	Police Commission 396	
1	Fire Chief	Fire Commission <sup>397</sup>	
4	Planning Director	Town Plan and Zoning	
		Commission <sup>398</sup>	
1	Animal Control	Police Commission <sup>399</sup>	
	Officer		
1	<b>Conservation Director</b>	Conservation-	
		Commission <sup>400</sup>	
			C 401
7	Board of Library	Board of Library	<del>o years</del> ···
7			6 years 401
7 1	Board of Library Trustees Town Librarian	Board of Library Trustees Board of Library	<del>o years</del> ···

<sup>&</sup>lt;sup>391</sup> Derived from Article I, §1.4.B.ii of the 1997 and 2006 Charters. 1 position is held by a member of the Board of Selectmen.

<sup>&</sup>lt;sup>392</sup> Derived from Article I, §1.4.B.ii of the 1997 and 2006 Charters.

<sup>&</sup>lt;sup>393</sup> Derived from Article I, §1.4.B.ii of the 1997 and 2006 Charters.

<sup>&</sup>lt;sup>394</sup> Current Article I, §1.4.B.ii (2006).

With the approval of the First Selectman. Derived from Chapter II, §2 of the 1947 and 1956 Acts and reaffirmed in Chapter II, §2 of the 1975 Charter and Article I, §1.4.B.ii of the 1997 and 2006 Charters

<sup>&</sup>lt;sup>396</sup> With the approval of the First Selectman. Derived from Article I, §1.4.B.iii of the 1997 and 2006 Charters.

<sup>&</sup>lt;sup>397</sup> With the approval of the First Selectman. Derived from Article I, §1.4.B.iii of the 1997 and 2006 Charters.

<sup>&</sup>lt;sup>398</sup> With the approval of the First Selectman. Derived from Article I, §1.4.B.iii of the 1997 and 2006 Charters.

<sup>&</sup>lt;sup>399</sup> Derived from Article I, §1.4.B.iii (2006 Charter).

With the approval of the First Selectman. Derived from Article I, §1.4.B.iii of the 2006 Charter.

<sup>&</sup>lt;sup>401</sup> With the approval of the Board of Selectmen; 1 of the positions is held bythe Town Treasurer. Derived from Article I, §1.4.B.iii of the 2006 Charter.

<sup>&</sup>lt;sup>402</sup> With the approval of the First Selectman.

### APPENDIX - TRANSITION PROCEDURES

**Board of Education.** Prior to the enactment of this Charter, the Charter provided for six (6) year terms for members of the Board of Education. Upon enactment of this Charter, the nine (9) membersof the Board of Education will have four (4) year terms. Members of the Board will be functions of an elected at alternating Town elections so that five (5) members are elected at one (1) Town election and four (4) at the next. This will be phased in over the period from 2007 to 2013 then continue as follows:

Year of Election. Future:	Original Election	<del>2007</del>	<del>2009</del>	<del>2011</del>	<del>2013</del>	<del>2015</del>	<del>2017</del>
BOE members							
4-year term @ 11/07	<del>2001</del>	2011		<del>2015</del>		2019	
4 year term @ 11/07	2001	<del>2011</del>		<del>2015</del>		<del>2019</del>	
6-year term @ 11/07; 4	2001	2013			<del>2017</del>		<del>2021</del>
years from 11/13							
4-year term @-11/09	<del>2003</del>		2013		<del>2017</del>		<del>2021</del>
4 year term @ 11/09	<del>2003</del>		<del>2013</del>		<del>2017</del>		<del>2021</del>
4-year term ————————————————————————————————————	<del>2003</del>		<del>2013</del>		<del>2017</del>		<del>2021</del>
A voor torm	<del>2005</del>			<del>2015</del>		<del>2019</del>	
4 year term @ 11/11	<del>2003</del>			<del>2013</del>		<del>2019</del>	
4-year term	<del>2005</del>			<del>2015</del>		<del>2019</del>	
<ul><li># 11/11</li><li>4 year term</li><li># 11/11</li></ul>	<del>2005</del>			2015		<del>2019</del>	
		2	2		4	5	4
<del>Total per</del> <del>Year:</del>		3	3	5	4	<del>5</del>	4

Police and Fire Commissions. In order to maintain these two (2) boards with an odd number of members at all times and at the same time not have more than two (2) terms expire in the same year, two (2) new members will be appointed to each of these boards in November, 2007. One new member will have a five (5) year term. The other new member will have a three (3) year term, at the expiration of which the term will become and remain five (5) years official.